PRICING GUIDE

FOR

DETARIFFED AND/OR UNREGULATED LOCAL EXCHANGE TELECOMMUNICATIONS SERVICES

PROVIDED BY

FIRST COMMUNICATIONS, LLC

GRANDFATHERED SERVICES FOR PREVIOUS CUSTOMERS OF CORECOMM NEWCO, INC.

SERVICES ARE NOT AVAILABLE TO NEW CUSTOMERS

This Pricing Guide contains the descriptions, regulation, and rates applicable to the furnishing of telecommunications services provided by First Communications, LLC with principal offices at 3340 W. Market Street, Akron, OH 44333.

The Pricing Guide includes the Local Exchange services offered to Customers within the State of Ohio that are detariffed and/or unregulated by the Public Utilities Commission of Ohio.

The general terms and conditions applicable to the services listed in this Pricing Guide are contained in the Company's P.U.C.O. Tariff No. 6 on file with the Public Utilities Commission of Ohio.

Copies of the Tariff on file with the Public Utilities Commission of Ohio may be inspected during normal business hours at the Company's principal place of business 3340 W. Market Street, Akron, OH 44333 or at the Public Utilities Commission of Ohio.

| I. | PRICE LIST APPLICABLE TO ALL LOCAL EXCHANGE AND/OR |
|----|--|
| | LONG DISTANCE CUSTOMERS |

| 2.6.2.8 | Phone Pa Presubsc 2.12.5 2.12.6 Adminis | I Check Charges | Page 5Page 5Page 5Page 5Page 5Page 5 |
|---------|---|--|--------------------------------------|
| II. | | LIST FOR LOCAL EXCHANGE SERVICES IN THE EX | XCHANGE |
| | | SERVED BY AMERITECH OHIO | _ |
| 4.1 | | Ordering Charges | _ |
| | 4.1.2 | Service Establishment Charge Rate Schedule | - |
| | 4.1.3 | Service Change Charge Rate Schedule | |
| | 4.1.4 | Service Change, Other Rate Schedule | |
| | 4.1.5 | Restoration of Service Rate Schedule | Page 14 |
| 4.3 | Exchang | e Access Service | Page 15 |
| | 4.3.3 | Basic Exchange Access Services | Page 15 |
| | 4.3.4 | Additional Charges Applied to Basic Exchange | |
| | | Access Services | Page 18 |
| 4.4 | Local Us | sage Services | Page 19 |
| | 4.4.3.1 | Measured Rate Service | - |
| | 4.4.3.2 | Local Calling Plus | Page 19 |
| | 4.4.3.3 | Message Rate Services | • |
| | 4.4.3.4 | Flat Rate Charges | _ |
| | 4.4.3.5 | Local Customer Service | - |
| | 4.4.3.6 | CoreComm Local Plan | _ |
| | 4.4.3.7 | E-Chat Service | C |
| | 4.4.3.8 | Residential Bundled Service | C |
| | 4.4.4.2 | CoreComm One-Way Optional Service | |
| | 4.4.5 | CoreComm Local Plus | |
| | 4.4.6 | Miscellaneous Local Service Options | • |
| | 4.4.7 | CoreComm Local Plus 5 | |
| | 4.4.8 | Core 100 | |
| 4.5 | | Exchange Access Service Enhancement Features | - |
| | 4.5.1 | General | - |
| | 4.5.2 | Identification of Out Dialed Calls (IOD) Service | _ |
| | 4.5.3 | Optional Exchange Access Services Enhancement | |
| | 1.5.5 | Features. | Page 46 |
| | 4.5.4 | Other Optional Exchange Access Service Enhancement | 1 ugc +0 |
| | 1.5.Т | Features | Page 40 |
| | 4.5.5 | Advantage Custom Business Service | _ |
| | 4.5.6 | | _ |
| | 4.5.0 | CoreComm Advantage Line Plus Service | 1 age 32 |

| | 4.5.7 | CoreComm Unlimited Small Business | Page 53 |
|--------|-----------|--|----------|
| 4.6.2 | Intralata | Operator Assistance Surcharges | Page 54 |
| 4.7.2 | | y Services | |
| 4.8.2 | Integrate | ed Services Digital Network (ISDN) Services | Page 59 |
| | 4.8.3 | National ISDN Direct Rate Schedule | Page 63 |
| | 4.8.4 | ISDN Prime | Page 65 |
| | 4.8.5 | Grandfathered Integrated Services Digital Network | |
| | | (ISDN) Service | _ |
| 4.9 | | g Services | |
| | | Transport Service | |
| | | Exchange Service | |
| 4.13 | | neous Service | |
| | | entrex Services | |
| 4.15.1 | | e Telecommunications Service | |
| | | Message Telecommunications Service Rate Schedu | |
| 4.16 | | Trials and Special Promotions | _ |
| 4.17 | | al Case Basis Arrangements | _ |
| 4.18 | Resident | tial Customer Referral Credit | Page 97 |
| III. | | FOR LONG DISTANCE SERVICE CUSTOMERS BSCRIBED TO CORECOMM SERVICE | |
| | 14.2.1.1 | Standard Service | Page 98 |
| | 14.2.1.2 | Option No. 1 Per Minute Usage Rates - Customer | Page 99 |
| | 14.2.1.3 | Option No. 2 Flat Rate Plans - Residential and | |
| | | Business Customers | |
| | | 800, 877 or 888 Service | |
| | | 700 IntraLATA Service | _ |
| | | Presubscribed Long Distance Customer Plans | |
| | | Local Customer Plans | |
| | | Business Customer Term and Volume Plans | _ |
| | | Residential Customer Rate Plans | _ |
| | | 0 Business Customer Rate Plans | |
| | | 1 Optional IntraLATA Message Toll Services | |
| | | 2 CoreComm Standalone Long Distance | _ |
| | | 3 CoreComm Business Switched Long Distance | |
| | 14.2.1.14 | 4 CoreComm Business Dedicated Long Distance | Page 139 |
| | 14.2.2 | Private Line Inter-POP Service | Page 141 |
| | 14.2.3 | Dedicated Access Service | • |
| | 14.2.4 | Miscellaneous Services | _ |
| | 14.2.4.1 | Directory Assistance | Page 144 |
| | | Operator Services | |
| | 14.2.4.3 | Inmate Calling Service | Page 144 |

| | 14.2.4.4 | CoreComm Calling Card Service - Customers | Page 145 |
|-------|----------|---|---------------|
| | 14.2.4.5 | Restoration of Service Charge | Page 145 |
| | 14.2.4.6 | Project Account ("PAC") Codes | Page 145 |
| | | • | C |
| IV. | PRICE I | LIST FOR SMART PACK SERVICES | |
| | 15.2 | Smart Pack Miscellaneous Charges | Page 147 |
| | 15.3 | Smart Pack Local Service | |
| | 15.4 | Smart Pack Long Distance Service | Page 151 |
| | 15.5 | Smart Pack Member Discount | |
| | 15.6 | Smart Pack Centrex. | |
| | 15.7 | Smart Pack Trunks | |
| | 13.7 | Smart I tex Tranks | 1 age 13 i |
| V. | PRICE I | LIST FOR CORE PACK SERVICES | |
| | 16.2 | Core Pack Business Services | Page 156 |
| VI. | SWITCH | HED ACCESS RATES | |
| | | | |
| | 17.5 | Rate Regulations | Page 162 |
| | 17.6 | Rates and Charges | Page 162 |
| | | | |
| VII. | CARRIE | ER COMMON LINE CHARGE | |
| | 18.4 | Rates and Charges | Page 163 |
| VIII. | PUBLIC | TELEPHONE SERVICE | |
| | 19.2 | Independent Payphone Provider Service | Page 164 |
| | | | C |
| IX. | COREC | ONNECT | |
| | 20.3.1 | CoreConnect Preferred | Page 167 |
| | 20.3.2 | CoreConnect Preferred Plus | |
| | 20.3.3 | CoreConnect Trunks | _ |
| | 20.3.4 | CoreConnect PRI | _ |
| | | | \mathcal{L} |

Effective June 1, 2021 4

I. PRICE LIST APPLICABLE TO ALL LOCAL EXCHANGE AND/OR LONG DISTANCE CUSTOMERS

2.6.2.4 <u>RETURNED CHECK CHARGE</u>

When payment for service is made by check, a charge will be made by the Company for each check returned by a bank to the Company for reason of non-sufficient funds.

per check \$25.00

2.6.2.8 PHONE PAYMENT CHARGE

When a Customer makes a debit card, credit card or check payment by speaking directly with the Company's representative, they will be assessed a phone payment charge.

Per payment \$3.00

2.12 PRESUBSCRIPTION

| | <u>Co</u> | <u>mplex</u> | <u>Si</u> | <u>mple</u> |
|----------------------------------|-----------|---------------|-----------|---------------|
| | Residence | Non-Residence | Residence | Non-Residence |
| 2.12.5 IntraLATA Presubscription | | | | |
| Initial line, trunk or port | \$5.00 | \$5.00 | \$5.00 | \$5.00 |
| Additional line, trunk or port | \$1.50 | \$1.50 | \$1.50 | \$1.50 |
| 2.12.6 InterLATA Presubscription | | | | |
| Initial line, trunk or port | \$5.00 | \$5.00 | \$5.00 | \$5.00 |
| Additional line, trunk or port | \$5.00 | \$5.00 | \$5.00 | \$5.00 |
| No Selection | | | | |
| Initial line, trunk or port | \$0.00 | \$0.53 | \$0.00 | \$0.53 |
| Additional line, trunk or port | \$0.00 | \$1.50 | \$0.00 | \$1.50 |

2.13 <u>ADMINISTRATIVE FEE</u>

An Administrative Fee is applied to First Communications accounts for the administration and facilitation of maintaining the physical assets required to provide service to our customers, which include but are not limited to invoicing, service support systems, facilities management and network operations.

The Administrative Fee is a monthly recurring charge per account of \$5.00 for residential customers and \$10.00 for commercial customers.

2.14 REGULATORY COMPLIANCE FEE

A Regulatory Compliance Fee will apply to all usage and MRC billed at a rate of at least 3.8573%.

II. PRICE LIST FOR LOCAL EXCHANGE SERVICES IN THE EXCHANGE AREAS SERVED BY AMERITECH OHIO

4.1 SERVICE ORDERING CHARGES

4.1.2 Service Establishment Charge--Rate Schedule

This charge applies when the Company initially establishes the Customer's account for any service provided by the Company, unless otherwise noted. This charge is applied in addition to any other monthly, installation, or non-recurring charge which is associated with the service the Customer orders.

| | <u>Complex</u> | | Sim | <u>ple</u> | |
|-----------------------------|----------------|-----------|------------------|------------|--|
| | | Non- | | Non | |
| | Residence | Residence | Residence | Residence | |
| General | | | | | |
| -Line Establishment Charges | | | | | |
| -Conversion of existing | | | | | |
| local line from incumbent | | | | | |
| local exchange carrier | | | | | |
| per location, per occasion | N/A | N/A | \$10.00 | N/A | |
| -Non-conversion | | | | | |
| one time per order | \$17.00 | \$15.85 | \$17.00 | \$25.50 | |
| -Central Office Connection | | | | | |
| per termination | \$9.00 | \$17.00 | \$ 9.00 | \$13.00 | |
| -Line Connection Charge | | | | | |
| per termination | \$10.00 | \$16.50 | \$10.00 | \$24.35 | |

| | Com | | Sim | <u>ple</u> Non |
|---|-----------|--------------------------|-----------|-------------------|
| a 'Pira' | Residence | Non- <u>Residence</u> | Residence | Residence |
| Semi-Public Service, one-way or two-way | | | | |
| -Service Ordering Charge per location, per occasion | | | | \$48.60 |
| -Central Office Connection per termination | | | | \$13.00 |
| -Line Connection Charge per termination | | | | \$63.00 |
| <u>Centrex Stations</u> <u>and Terminals</u> -Service Ordering Charge | | | | |
| per location, per occasion -Central Office Connection | | \$46.80 | | |
| per termination -Line Connection Charge | | \$2.75 | | |
| per termination | | \$3.35 | | |
| <u>PBX Trunks</u> -Service Ordering Charge | | | | |
| per location, per occasion -Central Office Connection | \$24.50 | \$24.50 | | |
| per termination | \$14.50 | \$14.50 | | |
| -Line Connection Charge per termination | \$10.35 | \$10.35 | | |
| FX Service | \$431.45 | \$455.70 | | |

Effective June 1, 2021 7

| | Comp | | Sim | |
|--|----------------|------------------|-----------|------------------|
| | Dagidanaa | Non- | Dagidanaa | Non Dagidanaa |
| OFF-PROPERTY EXTENSION AND STATIONS, per extension! station per location | Residence | <u>Residence</u> | Residence | Residence |
| Exchange Service Extension | | | | |
| -In same CO | \$129.20 | \$129.20 | | |
| -In different CO | \$234.20 | \$234.20 | | |
| Intercom System Station | \$87.70 | \$87.70 | | |
| PBX Station Line or PBX Extension Station Line -Terminating in same CO | | | | |
| area as PBX system | \$378.40 | \$378.40 | | |
| -Terminating in different | Φ445 2O | ¢445.20 | | |
| CO area from PBX system | \$445.30 | \$445.30 | | |
| Miscellaneous Service/Feature (per individual service, per occasion-Custom Calling Service and advanced Custom Calling Service (to establish one or more custom or advanced custom calling feature(s)) | on) \$20.00 | \$20.00 | \$20.00 | \$20.00 |
| | | | | |
| <u>Directory Service</u> -Transfer of Toll Charge | | | | |
| • | \$41.55 | \$41.55 | \$9.50 | \$9.80 |
| Service | \$41.55 | \$41.55 | \$9.50 | \$9.80 |

| | <u>Co</u> : | <u>mplex</u> | <u>S</u> : | <u>imple</u> |
|--|-------------|---------------|------------|---------------|
| D (CHE 1) | Residence | Non-Residence | Residence | Non-Residence |
| Remote Call Forwarding (per feature) | N/A | \$70.25 | N/A | N/A |
| Intercom Calling Service (per individual exchange service to establish Intercom Calling Service) | | | \$16.00 | \$11.25 |
| Multi-Ring Service | \$16.00 | | \$16.00 | |
| Caller ID Block | \$16.00 | | \$16.00 | |
| Call Control | \$16.00 | N/A | \$16.00 | N/A |

4.1.3 Service Change Charge--Rate Schedule

This charge applies anytime a Customer requests that his/her service or class of service be changed. This charge may be applied to a service as many times as the Customer requests that his/her service be changed and is charged in addition to any other monthly or installation charge which is associated with the service the Customer orders.

| | <u>Complex</u> | | <u>Simple</u> | |
|--|------------------|------------------|------------------|------------------|
| | | Non- | | Non- |
| | <u>Residence</u> | <u>Residence</u> | <u>Residence</u> | <u>Residence</u> |
| | | | | |
| Exchange Service | *** | *** | | |
| -one way to two-way, vice versa | \$28.45 | \$41.55 | | |
| -one or two way to combination, | \$28.45 | \$41.55 | | |
| vice versa | | | | |
| -PBX Trunk to Airport Trunk, | \$28.45 | \$41.55 | | |
| vice versa | | | | |
| -Individual to Airport Individual, | \$28.45 | \$41.55 | | |
| vice versa | | | | |
| -Message rate to flat rate, | | | Not | |
| vice versa | \$28.45 | | Available | |
| -Measured rate to flat rate, | | | Not | |
| vice versa | \$28.45 | \$41.55 | Available | |
| -Message rate to measured rate, | | | Not | |
| vice versa | \$28.45 | \$41.55 | Available | \$9.30 |
| -Semi-Public from One-Way to | 7-3115 | , | | 42.50 |
| Two-Way, vice versa | | | | \$9.30 |
| -Minute Line rate to flat rate, | | | Not | Ψ>.50 |
| vice versa | \$28.45 | | Available | |
| -Minute Line rate to message | Ψ20.43 | | Not | |
| _ | \$28.45 | | Available | |
| rate, vice versa -Minute Line rate to measured | φ ∠0.4 3 | | Not | |
| | ¢20.45 | | | |
| rate, vice versa | \$28.45 | | Available | |

| | | <u>Com</u> | olex Non- | Sim | <u>ple</u> Non- |
|-------|-----------------------------------|-----------------|---------------|-----------------|--------------------|
| | | Residence | Residence | Residence | Residence |
| 4.1.4 | Service Change. Other Rate Sch | nedule | | | |
| | The following charges are applica | able to each ch | ange from one | to another with | n no |
| | lapse in service. | | | | |
| | -Local Exchange Service | \$28.85 | \$41.55 | \$9.25 | \$9.30 |
| | -Optional Local Calling | | | | |
| | Area Service | N/A | N/A | N/A | N/A |
| | -Community Calling | | | | |
| | Service | N/A | N/A | N/A | N/A |
| | -Optional Off-Peak Toll | | | | |
| | Service | N/A | N/A | N/A | N/A |
| | -Econo-Call Service | N/A | N/A | N/A | N/A |
| | <u>Miscellaneous</u> | | | | |
| | Service/Feature | | | | |
| | (per individual service, per | | | | |
| | occasion) | | | | |
| | -Custom Calling Service | | | | |
| | and advanced Custom | | | | |
| | Calling Service (to | | | | |
| | change one or more | | | | |
| | custom or advanced | | | | |
| | custom calling feature(s)) | \$20.00 | \$20.00 | \$20.00 | \$20.00 |
| | Directory Services | | | | |
| | -From listed to private or | | | | |
| | semi-private, or from | | | | |
| | semi-private to private | | | | |
| | listing service | \$20.00 | \$20.00 | \$20.00 | \$20.00 |
| | -Discontinue emergency | | | | |
| | message referral for | | | | |
| | private listing service | \$17.90 | \$17.90 | \$14.00 | \$9.80 |
| | -Convert, for listing | | | | |
| | purposes only, an | | | | |
| | existing off-premises | | | | |
| | extension location to the | | | | |
| | main location, vice versa, | | | | |
| | when both locations are | . | . | * | |
| | with the same CO area | \$17.90 | \$41.55 | \$14.00 | \$9.80 |
| | | | | | |

| | <u>Co</u> <u>Residence</u> | mplex Non-Residence | <u>Si</u> Residence | imple Non-Residence |
|--|--|--|--|--|
| | Residence | Non-Residence | Residence | <u>ivon-residence</u> |
| - Change transfer of toll charge service | \$41.55 | \$41.55 | \$16.00 | \$9.80 |
| Telephone Number Change | | | | |
| Individual Exchange Service Initial Number Changed, Per Customer Subsequent Changes, Per Customer Present number in one number dialer Initial Number Changed, Per Customer Subsequent Changes, Per Customer Remote Call Forwarding | \$30.00 \$50.00 \$30.00 \$50.00 | \$30.00 \$50.00 \$30.00 \$50.00 | \$30.00 \$50.00 \$30.00 \$50.00 | \$30.00 \$50.00 \$30.00 \$50.00 |
| - Initial Number Changed, Per Customer | \$34.95 | \$34.95 | \$34.95 | \$34.95 |
| - Subsequent Changes, Per Customer | \$50.00 | \$50.00 | \$50.00 | \$50.00 |
| Class of Service Change | | | | |
| - Residence to Non-residence | \$20.00 | | \$29.95 | |
| - Non-residence to Residence | Ψ20.00 | N/A | Ψ=21.70 | N/A |
| Assignment or Transfer of Service | \$20.00 | \$41.55 | \$20.00 | \$20.00 |
| Intercom Calling Service (To change to or from Intercom Calling Service subsequent to the initial installation, per exchange service) | | NOT AVAI | LABLE | |
| Central Office Local Area Network | | | | |
| - Simple and/or complex post rearrangement; per location, per occasion | \$15.00 | \$15.00 | \$15.00 | \$15.00 |
| - CO LAN service is added subsequent to establishment of access service; per location, per occasion | \$15.00 | \$15.00 | \$15.00 | \$15.00 |

| | <u>Complex</u> | | <u>Simple</u> | |
|---|----------------|-------------------|---------------|-------------------|
| | Residence | Non- Residence | Residence | Non- Residence |
| Caller I.D. Blocking | \$16.00 | \$9.00 | | |
| Multi-Ring Service | | | | |
| per exchange service, per occasion | N/A | \$9.00 | N/A | |
| Call Control per exchange service, | | | | |
| per occasion | \$16.00 | N/A | \$16.00 | N/A |
| IntraLATA Presubscription Initial line, trunk or port | N/A | N/A | N/A | N/A |
| Additional line, trunk or port | N/A | N/A | N/A | N/A |

4.1.5 <u>Restoration of Service--Rate Schedule</u>

| | <u>Complex</u> | | <u>Simple</u> | |
|--|----------------|-----------|---------------|------------|
| | | Non- | | <u>Non</u> |
| | Residence | Residence | Residence | Residence |
| Service restored after temporary denial but prior to completion of order to discontinue service, per | | | | |
| service or system | \$50.00 | \$50.00 | \$50.00 | \$50.00 |
| Service restored after temporary suspension | \$62.30 | \$75.85 | \$50.00 | \$50.00 |
| Service restored after temporary interception | \$62.50 | \$87.70 | \$50.00 | \$50.00 |

4.3 Exchange Access Service

4.3.3 Basic Exchange Access Services

Rates for Basic Exchange Access Services are based upon class of service (as set forth in CoreComm Tariff No. 3 "Definitions" section) and access area as defined in CoreComm's Tariff No. 3, Section 4.2.2). All rates in this Section are applied monthly unless specifically identified otherwise.

4.3.3.1 Network Access Lines

A. Standard Network Access Line Rates

| | | Access Area | |
|---|-------------|-------------|-------------|
| | Area B | Area C | Area D |
| Residential Single Line, with Touch-Tone | \$9.10 | \$9.10 | \$9.10 |
| Residential Single Line, without Touch-Tone | \$6.70 | \$6.70 | \$6.70 |
| Residential Multi-Line, with Touch-Tone | \$24.70 | \$26.70 | \$26.70 |
| Residential Multi-Line, without Touch-Tone | \$23.70 | \$23.70 | \$23.70 |
| Business Single Line, with Touch-Tone* | Zone | Zone | Zone |
| | 1 - \$35.00 | 1 - \$37.00 | 1 - \$37.00 |
| | 2 - \$37.00 | 2 - \$39.00 | 2 - \$39.00 |
| | 3 - \$37.00 | 3 - \$39.00 | 3 - \$39.00 |
| Business Single Line, without Touch-Tone | \$23.00 | \$23.00 | \$23.00 |
| PBX Trunk, with Touch-Tone | \$24.70 | \$26.70 | \$26.70 |
| PBX Trunk, without Touch-Tone | \$23.70 | \$23.70 | \$23.70 |

These charges apply to Basic Exchange Access Services in addition to the charges found in Section 4.3.4 of CoreComm's Tariff No. 3.

Effective December 13, 1999, the Company will no longer accept new subscriptions to Business Single Line Non-Rotary Service (with or without Touch-Tone) or Business Rotary/PBX Trunk Service (with or without Touch-Tone) as provided above. Customers who were subscribers to such services prior to that date may maintain service at the above provided rates. Term commitments shall be required for all new subscriptions to Business Line and Business Rotary/PBX Trunk Services on or after December 13, 1999, as provided in Section 4.3.3.1B following.

Effective May 1, 2001, the Company will no longer accept new subscriptions to Residential Single Line or Multi-Line Services (with or without Touch-Tone) as provided above. Customers who were subscribers to such services prior to that date may maintain service as provided herein. As of May 1, 2001, new subscriptions to Residential services are available only as provided in Section 4.4 of this tariff.

^{*} Customers purchasing service under this plan or package may qualify for the Multi Product Credit. The Multi Product Credit provides Customers purchasing qualifying commercial telephone line package(s) at the same service address as an On Net 1.5mb or greater Circuit, including, but not limited to, the Essential products (VoIP), with an \$8 credit on the Customer's commercial telephone line package(s). In the event of On Net circuit disconnection, the Multi Product Credit shall no longer be available.

B. Business Customer Term Agreement Rates

The following services are available to Business Customers who subscribe to the Company's services for one of the following term commitments:

| Business Message Line Service Option 1, per line Non-Rotary with or without Touch Tone | All Access Areas Access Line Rates are included in usage rates in Sec. 4.4.3.3.B.1 |
|---|--|
| Business Message Line Service, per line * | |
| Non-Rotary with or without Touch Tone ¹ | |
| - Month-to-Month Term | Zone |
| | 1 - \$41.80 |
| | 2 - \$43.80 |
| | 3 - \$43.80 |
| - 1 Year Term | Zone |
| | 1 - \$35.00 |
| | 2 - \$37.00 |
| | 3 - \$37.00 |
| - 2 Year Term | Zone |
| | 1 - \$33.00 |
| | 2 - \$35.00 |
| | 3 - \$35.00 |
| - 3 Year Term | Zone |
| | 1 - \$31.50 |
| | 2 - \$33.50 |
| | 3 - \$33.50 |
| Business Measured Line Service, per line | |
| Non-Rotary with or without Touch Tone | |
| - Month-to-Month | Zone |
| Worth to Worth | 1 - \$41.80 |
| | 2 - \$43.80 |
| | 3 - \$43.80 |
| - 18 Month Term | \$23.00 |
| - 3 Year Term | \$21.00 |
| DDV T 1 W/d W/d T 1 T | |
| PBX Trunk, With or Without Touch Tone | Φ22.00 |
| - Month-to-Month Term | \$33.00 |
| - 1 Year Term | \$29.00 |
| - 2 Year Term | \$27.00 |
| - 3 Year Term | \$25.00 |

Available only to customers subscribing to Message Line Service Option 2 as provided in Section 4.4.3.3 of this tariff.

¹ Customers purchasing service under this plan or package may qualify for the Multi Product Credit. The Multi Product Credit provides Customers purchasing qualifying commercial telephone line package(s) at the same service address as an On Net 1.5mb or greater Circuit, including, but not limited to, the Essential products (VoIP), with an \$8 credit on the Customer's commercial telephone line package(s). In the event of On Net circuit disconnection, the Multi Product Credit shall no longer be available.

4.3.3.2 Business Network Access Lines Quantity Discounts

| | Monthly Ra | Access Area Monthly Rate per Network Access Line | | |
|------------------------|------------|---|----------|--|
| Access Line Quantities | D | C | D | |
| | <u>B</u> | <u>C</u> | <u>D</u> | |
| 48-95 | \$17.90 | \$19.70 | \$19.50 | |
| 96-199 | \$17.10 | \$18.70 | \$18.30 | |
| 200-293 | \$16.30 | \$17.70 | \$17.10 | |
| 294-387 | \$15.50 | \$16.70 | \$15.90 | |
| 388-579 | \$14.70 | \$15.70 | \$14.70 | |
| 580-above | \$13.90 | \$14.70 | \$13.50 | |

Effective December 13, 1999, the Company will no longer offer Business Network Access Line Quantity Discounts to new subscribers to the Company's service. Customers who were subscribers to the Company's Business Line Services and who qualified for Business Network Access Line Quantity Discounts prior to that date shall continue to receive monthly discounts as provided herein.

4.3.4. Additional Charges Applied to Basic Exchange Access Services

- (A) The End User Common Line charge as set forth in Section 8.7.1 of the Company's FCC Tariff No. 1.
- (B) The 911 Telecommunications Surcharge is \$1.50 per month.
- (C) The Local Number Portability ("LNP") surcharge as set forth in Section 8.7.2 of the Company's FCC Tariff No. 1.

4.4 <u>Local Usage Services</u>

4.4.3.1 Measured Rate Services

B. Rate Schedule

| Band | | <u>PEAK</u> | OF | F-PEAK |
|-------------|-----------------------|---------------------------|-----------------------|-------------------|
| | Initial Minute | Each Add'l Minute | Initial Minute | Each Add'l Minute |
| | | | | |
| A | \$0.0300 | \$0.0075 | \$0.0160 | \$0.0038 |
| В | \$0.0350 | \$0.0110 | \$0.0180 | \$0.0060 |
| C | \$0.0390 | \$0.0156 | \$0.0195 | \$0.0079 |
| | | | | |
| | Band | <u>Mileage</u> | | |
| | A | 0 – 11 Miles | | |
| | В | 11.1 - 22 Miles | | |
| | C | Over 22 Miles | | |
| | | | | |
| | Peak: | 8 a.m. to but not includ | ling 9 p.m. | |
| | Off-Peak: | 9 p.m. to but not include | ling 8 a.m. | |

Customers subscribing to the Company's Measured Rate Service are eligible to subscribe to the Company's long distance service packages established in Section 14.2.1.8 of this tariff.

4.4.3.2 Local Calling Plus

When ordered by the Public Utilities Commission of Ohio between specific exchanges, all rules and regulations for local message charges for Measured Rate Service are applicable to Local Calling Plus. However, Measured Rate Service is not required to take advantage of Local Calling Plus. It also is available to message and flat-rate service customers.

Local Calling Plus Charge Schedule

| | Residence | Business |
|-----------------------|-----------|----------|
| Flat rate, per minute | \$0.0225 | \$0.0225 |

4.4.3.3 Message Rate Services

B. Rate Schedule

1. <u>Business Option 1</u>:*

| | Monthly Usage Package | |
|---|-----------------------|-------------|
| | 18 Months | 36 Months |
| No Call Allotment | | |
| - Monthly | Zone | Zone |
| • | 1 - \$37.00 | 1 - \$41.00 |
| | 2 - \$39.00 | 2 - \$44.00 |
| | 3 - \$39.00 | 3 - \$44.00 |
| - Rate per message | \$ 0.08 | \$ 0.08 |
| 75 Call Allotment | | |
| - Monthly Usage Package | Zone | \$30.00 |
| | 1 - \$43.00 | |
| | 2 - \$45.00 | |
| | 3 - \$50.00 | |
| - Rate per message in excess of allotment | \$ 0.08 | \$ 0.08 |
| 400 Call Allotment | | |
| - Monthly Usage Package | \$55.00 | \$53.00 |
| - Rate per message in excess of allotment | \$ 0.08 | \$ 0.08 |

^{*} Customers purchasing service under this plan or package may qualify for the Multi Product Credit. The Multi Product Credit provides Customers purchasing qualifying commercial telephone line package(s) at the same service address as an On Net 1.5mb or greater Circuit, including, but not limited to, the Essential products (VoIP), with an \$8 credit on the Customer's commercial telephone line package(s). In the event of On Net circuit disconnection, the Multi Product Credit shall no longer be available.

2. <u>Business Option 2</u>:

| Rate Per Message |
|------------------|
| \$0.0800 |
| \$0.0750 |
| \$0.0725 |
| \$0.0700 |
| |

Effective June 1, 2021 20

4.4.3.4.1 Flat Rate Service -- Rates and Charges

Flat Rate Service consists of a usage package that includes an unlimited number of local messages.

Residence Individual Residence PBX Trunk Usage Package Monthly Rate \$10.15 NOT AVAILABLE

Effective May 1, 2001, the Company will no longer accept new subscriptions to Flat Rate Service. Customers who were subscribers to such service prior to that date may maintain service as provided above.

4.4.3.5.1 <u>Local Customers Service (or "Home Options") -- Rates and Charges</u>

| Option I | Zone |
|-------------|--------------|
| | 1 - \$114.95 |
| | 2 - \$114.95 |
| | 3 - \$114.95 |
| Option II | Zone |
| | 1 - \$114.95 |
| | 2 - \$114.95 |
| | 3 - \$114.95 |
| Option III | Zone |
| | 1 - \$114.95 |
| | 2 - \$114.95 |
| | 3 - \$114.95 |
| Option IV | Zone |
| | 1 - \$114.95 |
| | 2 - \$114.95 |
| | 3 - \$114.95 |
| Option V | Zone |
| | 1 - \$114.95 |
| | 2 - \$114.95 |
| | 3 - \$114.95 |
| Second Line | Zone |
| | 1 - \$114.95 |
| | 2 - \$114.95 |
| | 3 - \$114.95 |

| <u>Feature</u> | Point Value |
|------------------------------------|-------------|
| 30 Minutes Long Distance Package* | 1 |
| 60 Minutes Long Distance Package* | 2 |
| 100 Minutes Long Distance Package* | 3 |
| 180 Minutes Long Distance Package* | 5 |
| Automatic Callback, per Line | 1 |
| Call Forwarding | 1 |
| Call Screening | 1 |
| Call Waiting | 1 |
| Caller ID With Name and Number | 2 |
| Multi-Ring Service, first number | 1 |
| Repeat Dialing, per Line | 1 |
| Selective Call Screening | 1 |
| Speed Call | 1 |
| Three-Way Calling, per Line | 1 |
| | |

Option F to the Local Customer Plan, Section 14.2.1.7 following, may be purchased separately at the monthly charge of \$95.00.

Effective January 2, 2001, no new subscriptions will be accepted. Customers who subscribed to Options I through V as of January 1, 2001 (for the purposes of Section 4.4.3.5 "Eligible Customers") may maintain their subscription at the above rates, subject to the following:

- 1. Option upgrades or downgrades are not permitted (e.g., from option I to III or vice versa).
- 2. Rates provided in Section 4.5.3 apply to all <u>new features</u> added on or after January 2, 2001
- 3. Eligible Customers who delete features on or after January 2, 2001 will be converted to the Company's Unlimited Local Usage Area Calling Service established in Section 4.4.3.6 and rates for all features will apply as provided in Section 4.5.3.

4.4.3.6 CoreComm Local Plan

A. <u>Monthly Rate</u>

| | Monthly Rate |
|--|--------------|
| CoreComm Local Plan | |
| - All Access Areas, first access line | Zone |
| , | 1 - \$114.95 |
| | 2 - \$114.95 |
| | 3 - \$114.95 |
| - All Access Areas, each additional line | Zone |
| | 1 - \$114.95 |
| | 2 - \$114.95 |
| | 3 – \$114.95 |

Local Calling Plus charges apply as provided in Section 4.4.3.2 of this Tariff.

B. <u>Long Distance Service</u>

| | Initial 60 Seconds | Each Add'l 6 Seconds | |
|-----------------|--------------------|----------------------|--|
| - IntraLATA MTS | \$0.100 | \$0.010 | |
| - InterLATA MTS | \$0.100 | \$0.010 | |

4.4.3.7 eChat Service

Effective August 15, 2001, no new subscriptions to eChat Service will be accepted. However, Customers subscribing to eChat Service as of August 15, 2001 may maintain their subscriptions, subject to the following limitations.

- 1. eChat Service may not be moved to a new location.
- 2. eChat Service, where disconnected for non-payment, may not be re-established.

B. <u>eChat Basic Service</u>

1. Monthly Rate

| Rute | Monthly Data |
|--|----------------------|
| - All Access Areas, first line | Monthly Rate Zone |
| - All Access Aleas, first file | 1 - \$42.95 |
| | 2 - \$44.95 |
| | - + |
| A 11 A A 1 1 11 41 1 11 | 3 – \$49.95 |
| - All Access Areas, each additional line | Zone |
| | 1 - \$26.95 |
| | 2 - \$28.95 |
| | 3 - \$33.95 |

These monthly charges apply in addition to the charges established in Section 4.3.4 of this Tariff.

Local Calling Plus charges apply as provided in Section 4.4.3.2 of this Tariff.

2. Switched Inbound 800/877/888 Long Distance Usage

| | Initial 60 Seconds | Each Add'l 6 Seconds |
|--|--------------------|----------------------|
| Switched Inbound IntraLATA Usage | \$0.150 | \$0.015 |
| - Switched Inbound InterLATA Usage | \$0.150 | \$0.015 |

Additionally, the following per call surcharge shall apply to all calls to the Customer's 800/877/888 number that originate from a payphone.

- Per Call Surcharge

\$1.25

3. <u>Switched Outbound Message Toll Service (MTS) Long Distance</u>

| | <u>Initial 60 Seconds</u> | Each Add'l 6 Seconds |
|-----------------|---------------------------|----------------------|
| - IntraLATA MTS | \$0.100 | \$0.010 |
| - InterLATA MTS | \$0.100 | \$0.010 |

4. <u>Miscellaneous Charges</u>

Project Account Codes (PAC)*

- Non-verified codes
- Set-Up Charge \$5.00
- Verified codes
- Set-Up Charge \$5.00
- Custom codes
- Set-Up Charge \$50.00

^{*} PACs are available where facilities permit.

Monthly Rate

GRANDFATHERED SERVICES FOR PREVIOUS CUSTOMERS OF CORECOMM NEWCO, INC.

C. <u>eChat Plus Service</u>

1. Monthly Rate

| - All Access Areas, first line | \$29.95 |
|--|-------------|
| - All Access Areas, each additional line | Zone |
| , | 1 - \$23.95 |
| | 2 - \$25.95 |
| | 3 - \$30.95 |

These monthly charges apply in addition to the charges established in Section 4.3.4 of this Tariff.

Local Calling Plus charges apply as provided in Section 4.4.3.2 of this Tariff.

2. <u>Switched Inbound 800/877/888 Long Distance Usage</u>

| | Initial 60 Seconds | Each Add'l 6 Seconds |
|---|--------------------|----------------------|
| Switched Inbound IntraLATA UsageSwitched Inbound InterLATA Usage | \$0.150 \$0.150 | \$0.015 \$0.015 |
| | | |

Additionally, the following per call surcharge shall apply to all calls to the Customer's 800/877/888 number that originate from a payphone.

- Per Call Surcharge

3. Switched Outbound Message Toll Service (MTS) Long Distance

| | Initial 60 Seconds | Each Add'l 6 Seconds | |
|-----------------|--------------------|----------------------|--|
| - IntraLATA MTS | \$0.100 | \$0.010 | |
| - InterLATA MTS | \$0.100 | \$0.010 | |

\$1.25

4. <u>Calling Card Service</u>

The following rates shall apply for all intrastate calling card calls:

| | Initial 60 Seconds | | Each Ad | d'l 6 Seconds |
|--|--------------------|--------------------|--------------------|--------------------|
| | Peak Period | Off-Peak Period | Peak Period | Off-Peak Period |
| - IntraLATA Calls - InterLATA Calls | \$0.260 \$0.260 | \$0.220 \$0.220 | \$0.026 \$0.026 | \$0.022 \$0.022 |

\$0.25

Additionally, the following per call surcharge shall apply to all calling card calls:

- Per Call Surcharge

Additionally, the following per call surcharge shall apply to all calling card calls that originate from a payphone.

- Per Call Surcharge \$1.25

5. <u>Miscellaneous Charges</u>

Project Account Codes (PAC)*

| Non-verified codesSet-Up Charge | \$5.00 |
|--|---------|
| Verified codesSet-Up Charge | \$5.00 |
| Custom codesSet-Up Charge | \$50.00 |

^{*} PACs are available where facilities permit.

4.4.3.8 Residential Bundled Services

4.4.3.8.1 <u>Local Plus 2</u>

A. <u>Monthly Rate</u>

| M | [ont | hl' | y] | <u>Rate</u> |
|---|------|-----|------------|-------------|
| | | | | |

| - All Access Areas, first line | Zone |
|--|--------------|
| | 1 - \$114.95 |
| | 2 - \$114.95 |
| | 3 – \$114.95 |
| - All Access Areas, each additional line | \$114.95 |

These monthly charges apply in addition to the charges established in Section 4.3.4 of this Tariff. Local Calling Plus charges apply as provided in Section 4.4.3.2 of this Tariff.

B. Features Available (choice of two included) *

Automatic Callback

Basic Voice Mail

Busy Line Transfer

Alternate Answer

Call Screening

Multi Ring

Repeat Dialing

Caller ID w/ Name

Call Forwarding

Caller Way Calling

C. <u>Switched Outbound Message Toll Service (MTS) Long Distance Rates</u>

| | Initial Minute | Each Add'l 6 Seconds |
|-----------------|----------------|----------------------|
| - IntraLATA MTS | \$0.05 | \$0.005 |
| - InterLATA MTS | \$0.05 | \$0.005 |

^{*} Local features are not included in all areas; some restrictions apply.

4.4.3.8.2 <u>Local Plus 5</u>

A. Monthly Rate

| | Monthly Rate |
|----------------------|--------------|
| | Zone |
| | 1 - \$114.95 |
| All Access Areas, | 2 - \$114.95 |
| first line | 3 – \$114.95 |
| | Zone |
| | 1 - \$114.95 |
| All Access Areas, | 2 - \$114.95 |
| each additional line | 3 – \$114.95 |

These monthly charges apply in addition to the charges established in Section 4.3.4 of this Tariff. Local Calling Plus charges apply as provided in Section 4.4.3.2 of this Tariff.

4.4.3.8.2 <u>Local Plus 5 (continued)</u>

B. Features Available (choice of five included) *

Automatic Callback

Basic Voice Mail

Busy Line Transfer

Alternate Answer

Call Screening

Multi Ring

Repeat Dialing

Caller ID w/ Name

Speed Dialing

Call Forwarding

Three Way Calling

^{*} Local features are not included in all areas; some restrictions apply.

C. Switched Outbound Message Toll Service (MTS) Long Distance Rates

| | Customers Subscribing Prior to 5/1/2003 | | Customers Subscribing from 5/1/2003 Onward | | Strategic Alliance – Discounted Employer Group & Member Association Pricing | |
|------------------|--|---------------------------|--|---------------------------------|--|---------------------------------|
| | Initial Minute | Each Additional 6 Seconds | Initial Minute | Each Additional 6 Seconds | Initial Minute | Each Additional 6 Seconds |
| IntraLATA MTS | \$0.0800 | \$0.0080 | \$0.0500 | \$0.0050 | \$0.0450 | \$0.0045 |
| InterLATA MTS | \$0.0500 | \$0.0050 | \$0.0500 | \$0.0050 | \$0.0450 | \$0.0045 |

Effective May 1, 2003, new customers to Local Plus 5 service will pay the new rates specified above for switched long distance service. Current Local Plus 5 customers may continue their service at the long distance rates in effect prior to May 1, 2003 or choose service under the new rates.

<u>Strategic Alliance – Discounted Employer Group & Member Association Pricing</u> Customers who are employed by, or are members of, the following Partner Groups are eligible to receive discounted Local Plus 5 long distance service as outlined above. Customers who choose Strategic Alliance Option 1 under Section 4.4.3.8.2.A receive a discount off of the standard Local Plus 5 long distance rate. The Partner Groups are listed in Section 4.4.3.8.2.A.

4.4.3.8.3 Unlimited¹

A. Monthly Rate

| | Monthly Rate | | | | | |
|-------------------|--|--|---|---------------|---|---------------|
| | Customers Subscribing Prior to 5/1/2003 | Customers Subscribing from 5/1/2003 to 6/18/2003 | Customers Subscribing from 6/18/2003 Onward | | Strategic Alliance – Discounted Employer Group & Member Association Pricing | |
| | | | Option 1 – | Option 2 – | Option 1 – | Option 2 – |
| | | No Long | With | Without | With | Without |
| | | Distance | CoreComm | CoreComm | CoreComm | CoreComm |
| | | Required | Long Distance | Long Distance | Long Distance | Long Distance |
| | Zone | Zone | Zone | Zone | Zone | Zone |
| | 1 - \$119.95 | 1 - \$119.95 | 1 - \$119.95 | 1 - \$119.95 | 1 - \$119.95 | 1 - \$119.95 |
| All Access | 2 - \$119.95 | 2 - \$119.95 | 2 - \$119.95 | 2 - \$119.95 | 2 - \$119.95 | 2 - \$119.95 |
| Areas, first line | 3 - \$119.95 | 3 - \$119.95 | 3 - \$119.95 | 3 - \$119.95 | 3 - \$119.95 | 3 – \$119.95 |
| | Zone | Zone | Zone | Zone | Zone | Zone |
| All Access | 1 - \$119.95 | 1 - \$119.95 | 1 - \$119.95 | 1 - \$119.95 | 1 - \$119.95 | 1 - \$119.95 |
| Areas, each | 2 - \$119.95 | 2 - \$119.95 | 2 - \$119.95 | 2 - \$119.95 | 2 - \$119.95 | 2 - \$119.95 |
| additional line | 3 – \$119.95 | 3 – \$119.95 | 3 – \$119.95 | 3 – \$119.95 | 3 - \$119.95 | 3 – \$119.95 |

These monthly charges apply in addition to the charges established in Section 4.3.4 of this Tariff. Local Calling Plus charges apply as provided in Section 4.4.3.2 of this Tariff. Effective May 1, 2003, new customers to the Unlimited plan will pay a higher rate for subscribing to an additional line.

Effective June 18, 2003, new customers subscribing to Unlimited will be able to choose service from Option 1 or Option 2 pricing alternatives. Option 1 gives customers a \$5 discount for subscribing to the Company's long distance service under the rates listed in Section 4.4.3.8.3.C in this tariff. Customers subscribing to Unlimited prior to June 18, 2003 may maintain their service under their existing rates, terms and conditions even if they move and re-establish service after June 18, 2003. Current customers can also switch their service to Option 1 or 2.

-

¹ This plan included unregulated services, which are not under the PUCO's jurisdiction.

<u>Unlimited</u> (continued) 4.4.3.8.3

A. Monthly Rate (continued)

Strategic Alliance – Discounted Employer Group & Member Association Pricing

Customers who are employed by, or are members of, the following Partner Groups are eligible to receive discounted Unlimited service as outlined above. Customers who choose Strategic Alliance Option 1 receive a discount of \$5.05 off of the standard Unlimited price if they subscribe to the Company's long distance service per Section 4.4.3.8.3.C in this tariff. Customers who choose Strategic Alliance Option 2 receive a \$7.05 discount off of the standard Option 2 price.

Strategic Alliance Partner Groups

В.

Worldwide Communications Group

AAA Jump Home Cellnet Wendy's Nationwide Monstermoving.com

Features Available (choice of five included) *

Automatic Callback Call Waiting Basic Voice Mail Call Screening Busy Line Transfer Multi Ring Alternate Answer Repeat Dialing Caller ID w/ Name Speed Dialing Three Way Calling Call Forwarding

^{*} Local features are not included in all areas; some restrictions apply.

4.4.3.8.3 <u>Unlimited (continued)</u>

C. Switched Outbound Message Toll Service (MTS) Long Distance Rates

| | Customers Subscribing Prior to 6/18/2003 | | Customers Subscribing from 6/18/2003 Onward | | Strategic Alliance – Discounted Employer Group & Member Association Pricing | |
|---------------|--|--------------------|---|--------------------|--|--------------------|
| | Initial | Each Additional | Initial | Each Additional | Initial | Each Additional |
| | Minute | 6 Seconds | Minute | 6 Seconds | Minute | 6 Seconds |
| IntraLATA MTS | \$0.0800 | \$0.0080 | \$0.0500 | \$0.0050 | \$0.0450 | \$0.0045 |
| InterLATA MTS | \$0.0500 | \$0.0050 | \$0.0500 | \$0.0050 | \$0.0450 | \$0.0045 |

Effective June 18, 2003, new customers to Unlimited service will pay the new rates specified above for switched long distance service. Current Unlimited customers may continue their service at the long distance rates in effect prior to June 18, 2003 or choose service under the new rates.

Strategic Alliance – Discounted Employer Group & Member Association Pricing

Customers who are employed by, or are members of, the following Partner Groups are eligible to receive discounted Unlimited plan long distance service as outlined above. Customers who choose Strategic Alliance Option 1 under Section 4.4.3.8.3.A receive a discount off of the standard Unlimited long distance rate. The Partner Groups participating in this program are listed in Section 4.4.3.8.3.A.

4.4.3.8.4 2-Line Premium Connect¹

A. Monthly Rate

| | Monthly Rate | | | |
|-------------------------|--|-----------------------|-----------------------|--|
| | Customers Subscribing Prior to 5/1/2003 Customers Subscribing from 5/1/20 Onward | | | |
| | | Option 1 – With | Option 2 – Without | |
| | | CoreComm Long | CoreComm Long | |
| | | Distance | Distance | |
| | Zone | Zone | | |
| | 1 - \$135.95 | 1 - \$135.95 | | |
| All Access Areas, first | 2 - \$135.95 | 2 - \$135.95 | | |
| line | 3 - \$135.95 | 3 - \$135.95 | \$135.95 | |
| All Access Areas, | | | | |
| additional line | Included ² | Included ² | Included ² | |

Effective May 1, 2003, new customers subscribing to 2-Line Premium Connect will be able to choose service from Option 1 or Option 2 pricing alternatives. Option 1 gives customers a \$10 discount for subscribing to the Company's long distance service under the rates listed in Section 4.4.3.8.4.C in this tariff. Customers subscribing to 2-Line Premium Connect prior to May 1, 2003 may maintain their service under their existing rates, terms and conditions even if they move and re-establish service after May 1, 2003. Current customers can also switch their service to Option 1 or 2.

These monthly charges apply in addition to the charges established in Section 4.3.4 of this Tariff. Local Calling Plus charges apply as provided in Section 4.4.3.2 of this Tariff.

¹ This plan included unregulated services which are not under the PUCO's jurisdiction.

² An additional line is included as noted in the terms and conditions of Section 4.4.3.8.4 of the tariff.

2-Line Premium Connect¹ (continued) 4.4.3.8.4

B. Features Available (choice of five included) *

Automatic Callback Call Waiting Basic Voice Mail Call Screening Busy Line Transfer Multi Ring Alternate Answer Repeat Dialing Speed Dialing Caller ID w/ Name Three Way Calling Call Forwarding

C. Switched Outbound Message Toll Service (MTS) Long Distance Rates

| | Customers Subscribing Prior to 5/1/2003 | | Customers Subscribing from 5/1/2003 Onward | |
|------------------|--|----------|--|------------------------------|
| | Initial Each Additional Minute 6 Seconds | | Initial Minute | Each Additional 6 Seconds |
| IntraLATA MTS | \$0.0800 | \$0.0080 | \$0.0500 | \$0.0050 |
| InterLATA MTS | \$0.0500 | \$0.0050 | \$0.0500 | \$0.0050 |

Effective May 1, 2003, new customers to 2-Line Premium Connect service will pay the new rates specified above for switched long distance service. Current 2-Line Premium Connect customers may continue their service at the long distance rates in effect prior to May 1, 2003 or choose service under the new rates.

^{*} Local features are not included in all areas; some restrictions apply.

¹ This plan included unregulated services which are not under the PUCO's jurisdiction.

4.4.3.8.5 <u>CoreComm Freedom</u>

A. Monthly Rate

| | Monthly Rate |
|---|--------------|
| - All Access Areas, first line ¹ | Zone |
| , | 1 - \$139.95 |
| | 2 - \$139.95 |
| | 3 – \$139.95 |
| - All Access Areas, each additional line | Zone |
| 1111111000001110000, 0001111000001111100 | 1 - \$139.95 |
| | 2 - \$139.95 |
| | 3 – \$139.95 |

These monthly charges apply in addition to the charges established in Section 4.3.4 of this Tariff. Local Calling Plus charges apply as provided in Section 4.4.3.2 of this Tariff.

CoreComm Freedom plan includes unlimited calling during all time-of-day rate periods for direct dialed 1+ interLATA interstate, interLATA intrastate, intraLATA interstate, and intraLATA intrastate domestic calls.

Minutes used for Operator Assisted Calling, Calling Card, Toll Free Service, Directory Assistance, and International calling are excluded. Unlimited long distance service is for residential domestic direct-dial calls within the 50 U.S. states. Long distance calling for business or Internet access is not included. If the CoreComm Freedom plan is used for non-residential use as described previously, customer will be moved to a measured long distance service plan.

B. Features Available (any of the following) *

Automatic Callback

Busy Line Transfer

Alternate Answer

Call Screening

Multi Ring

Caller ID w/ Name

Call Forwarding

Speed Dialing

Three Way Calling

_

^{*} Local features are not included in all areas; some restrictions apply.

¹ CoreComm Freedom service is not currently available in Lima and Mansfield.

4.4.4 Other Exchange Access Services

4.4.4.2 CoreComm One-Way Optional Service

B. Rates and Charges

1. Option 1

The monthly rates for the network access line, central office termination and usage package for Optional Local Area Service are the rates specified in Sections 4.3 and 4.4 preceding, as appropriate, for the grade and class of service involved plus the following increment, for one-way calling to the exchange areas listed in Section 4.4.4.2.A preceding. The following rates are applicable to all CoreComm One-Way Optional Service Customers except those who qualify under Section 4.4.4.2.B.2.

| | Monthly Message Rate ² |
|---|-----------------------------------|
| PBX Trunk | |
| - Akron | \$7.25 |
| - Dayton Met. Area (Beavercreek, Bellbrook, | |
| Centerville, Fairborn, Miamisburg-West | |
| Carrollton, Vandalia) | Not Avail. |
| - Kirtland | Not Avail. |
| - Mentor | \$24.25 |
| - Painesville | \$24.45 |
| - Xenia | Not Avail. |
| - Yellow Springs-Clifton Option B | Not Avail. |
| Non-Residence Services, Non-rotary and Rotary | |
| - Akron | \$7.25 |
| - Dayton Met. Area (Beavercreek, Bellbrook, | |
| Centerville, Fairborn, Miamisburg-West | |
| Carrollton, Vandalia) | Not Avail. |
| - Kirtland | Not Avail. |
| - Mentor | \$24.45 |
| - Painesville | \$24.45 |
| - Xenia | Not Avail. |
| - Yellow Springs-Clifton Option B | Not Avail. |

² The local message allowance and additional local message charges applicable to outgoing calls within both the basic and optional local service areas are covered in Section 4.4.3 preceding

4.4.4.2 CoreComm One-Way Optional Service (Cont'd)

The following Optional Local Area Service incremental rate is applicable per Centrex station with one-way calling to the exchange areas listed in Section 4.4.4.2.A preceding.

The incremental rate is in addition to the monthly rate for the Centrex station.

| Centrex Service | |
|--|-----------------|
| Optional Local Area Service increment, per station | Monthly Price 3 |
| | |
| - Akron | \$1.50 |
| - Dayton Met. Area (Beavercreek, Bellbrook, | |
| Centerville, Fairborn, Miamisburg-West | |
| Carrollton, Vandalia) | Not Avail. |
| - Kirtland | Not Avail. |
| - Mentor | \$3.00 |
| - Painesville | \$3.15 |
| - Xenia | Not Avail. |
| - Yellow Springs-Clifton Option B | Not Avail. |
| | |

-

³ The local message allowance and additional local message charges applicable to outgoing calls within both the basic and optional local service areas are covered in Section 4.4.3 preceding

4.4.4.2 <u>CoreComm One-Way Optional Service</u> (Cont'd)

2. <u>Option 2</u>

The monthly rates for the network access line, central office termination and usage package for Optional Local Area Service are the rates specified in Sections 4.3 and 4.4 preceding, as appropriate, for the grade and class of service involved plus the following increment, for one-way calling to the exchange areas listed in Section 4.4.4.2.A preceding. The following rates are applicable only to Customers who were previously subscribed to Zone Centrex service (also referred to as City-wide or Telemanagement Centrex) and who convert their service to CoreComm One-Way Optional Service.

| Non-Residence Services, Non-rotary and Rotary | Monthly Message Rate 4 |
|---|------------------------|
| - Akron | Not Avail. |
| - Dayton Met. Area (Beavercreek, | |
| Bellbrook, Centerville, Fairborn, | |
| Miamisburg-West Carrollton, Vandalia) | Not Avail. |
| - Kirtland | Not Avail. |
| - Mentor | \$15.50 |
| - Painesville | Not Avail. |
| - Xenia | Not Avail. |
| - Yellow Springs-Clifton Option B | Not Avail. |

-

⁴ The local message allowance and additional local message charges applicable to outgoing calls within both the basic and optional local service areas are covered in Section 4.4.3 preceding

4.4.5. CoreComm Local Plus

A. General

CoreComm Local Plus offers Customers a basic exchange access line, unlimited local calling to parties located in the Customer's local usage service area (as established in Section 4.4.2 of this Tariff), an 800/877/888 number, Call Waiting and Caller ID with Name (as defined in Section 4.5.3 of this Tariff) for the following monthly rate:

Monthly Data

| | Monthly Rate |
|-------------------------------------|--------------|
| - All Access Areas, First Line | Zone |
| | 1 - \$114.95 |
| | 2 - \$114.95 |
| | 3 - \$114.95 |
| - All Access Areas, Each Add'l Line | Zone |
| | 1 - \$114.95 |
| | 2 - \$114.95 |
| | 3 – \$114.95 |

Effective October 20, 2002, no new subscriptions to CoreComm Local Plus will be accepted. However, Customers subscribing to CoreComm Local Plus as of October 20, 2002 may maintain their subscriptions, subject to the following possible limitations.

- 1. The Company may determine that CoreComm Local Plus may not be moved to a new location.
- 2. The Company may determine that CoreComm Local Plus, where disconnected for non-payment, may not be re-established.

B. Additional Charges

In addition to the monthly rate provided above, the following charges apply

- 1. Monthly charges as specified in Section 4.3.4 of this Tariff
- 2. Local Calling Plus charges as specified in Section 4.4.3.2 of this Tariff
- 3. Usage charges for switched inbound 800/877/888 service at the rates provided in Section 14.2.1.7.B of this Tariff.

Further, Customers subscribing to CoreComm Local Plus may additionally subscribe to the Company's switched outbound intraLATA and interLATA long distance services at the rates provided in Section 14.2.1.9.B of this Tariff.

Customers subscribing to Optional Access Service Enhancement Features in addition to those included in CoreComm Local Plus shall be assessed the applicable rates provided in Section 4.5.3 of this Tariff.

4.4.6 <u>Miscellaneous Local Service Options</u>

A. Advantage Community Calls

1. Monthly Rate

The following incremental rates for Business service are in addition to the monthly rates specified in Section 4.3.3.1.B, Option 2 this tariff, as appropriate for the Class of Service involved. The incremental rates are for the first two (2) hours of use of Advantage Community Calls.

| Business Service | Monthly Rate |
|-------------------------|--------------|
| Message Rate* | |
| - Individual Non-Rotary | \$29.40 |
| - Individual Rotary | \$29.40 |
| - PBX Trunk | \$29.40 |

- * The local message charges applicable to outgoing calls within the basic and optional local service areas are covered in Section 4.4.3.3.B.2 of these price sheets.
- 2. The following charge applies for each additional fifteen (15) minutes of use of Advantage Community Calls, or fraction thereof:

All Business services \$3.50

- 3. Unless otherwise requested by the Customer, when Advantage Community Calls are furnished on two (2) or more services of the same class and grade on a given premises, the initial time period will be the product of two (2) hours multiplied by number of services; the additional period will be the sum of the accumulated measured times of such services in excess of that product.
- 4. Advantage Community Calls service is available only on a "grandfathered basis" to Customers who were acquired by the Company from USN Communications, Inc. and who were subscribers to the service on April 2, 2001.

B. <u>Local Calling Plan Option</u>

1. Business Services

The monthly rates for Local Calling Plan Option are the rates specified in Section 4.3.3.1.B, Option 2 of this tariff, plus the following increment, for one-way calling to the exchange areas provided above.

Non-rotary, rotary, and PBX Trunk

| | Monthly Message Rate |
|---|----------------------|
| Akron | \$7.25 |
| Dayton Metro. Area (Beavercreek, Bellbrook, | \$7.25 |
| Centerville, Fairborn, Miamisburg-West | |
| Carrollton, Vandalia) | |
| Kirtland and Mentor | \$19.05 |
| Painesville | \$24.25 |
| Xenia | \$19.05 |
| Yellow Springs - Clifton | \$19.05 |
| | |

- * The monthly message rate is in addition to the monthly rate as provided in Section 4.3.3.1.B, Option 2. The local message charges applicable to outgoing calls within the basic and optional local service areas are covered in Section 4.4.3.3.B.2 of these price sheets.
- 2. Local Calling Plan Option is available only on a "grandfathered basis" to Customers who were acquired by the Company from USN Communications, Inc. and who were subscribers to the service on April 2, 2001.

C. Advantage Call Option

1. The following incremental rates are in addition to the monthly rate as provided in Section 4.3.3.1.B, Option 2. The incremental rates are per individual exchange.

Monthly Rate

Business

\$5.75

- 2. When Advantage Call is provided on message rate service, the additional local message charges applicable to outgoing calls within both the basic and optional local service areas are covered in Section 4.4.3.3.B.2.
- 3. The following charge applies per local message for Advantage Call

<u>Charge</u> \$0.25

For message service, each local message

4. Advantage Call Option is available only on a "grandfathered basis" to Customers who were acquired by the Company from USN Communications, Inc. and who were subscribers to the service on April 2, 2001

4.4.7 CoreComm Local Plus Five

4.4.7.1 General

CoreComm Local Plus Five ("CLPF") offers Customers a basic exchange access line; unlimited local calling to parties located in the Customer's local usage service area (as established in Section 4.4.2 of this Tariff); Call Waiting, Caller ID with Name, Three Way Calling, Call Forwarding, and Repeat Dialing (as defined in Section 4.5.3 of this Tariff); and a toll-free number for the following monthly rate:

| | Monthly Rate |
|-------------------------------------|--------------|
| - All Access Areas, First Line | \$114.95 |
| - All Access Areas, Each Add'l Line | \$114.95 |

Effective October 20, 2002, no new subscriptions to CoreComm Local Plus Five will be accepted. However, Customers subscribing to CoreComm Local Plus as of October 20, 2002 may maintain their subscriptions, subject to the following possible limitations.

- 1. The Company may decide that CoreComm Local Plus Five may not be moved to a new location.
- 2. The Company may decide that CoreComm Local Plus Five, where disconnected for non-payment, may not be re-established.

4.4.7.2 Additional Charges

- A. In addition to the monthly rate provided above, the following charges apply
 - 1. Monthly charges as specified in Section 4.3.4 of this Tariff
 - 2. Local Calling Plus charges as specified in Section 4.4.3.2 of this Tariff
 - 3. Usage charges for switched inbound toll-free service at the rates provided in Section 14.2.1.7.B of this Tariff.
- B. Customers subscribing to CLPF may additionally subscribe to the Company's switched outbound intraLATA and interLATA long distance services at the rates provided in Section 14.2.1.9.B of this Tariff.
- C. Customers subscribing to Optional Access Service Enhancement Features in addition to those included in CLPF are assessed the applicable rates for the features selected as provided in Section 4.5.3 of this Tariff.

4.4.8 <u>Core100</u>

A. Monthly Rate

Core 100 is a competitive bundled service that offers a basic exchange access line, unlimited calling to parties located in the Customer's local usage service areas A & B (as established in Section 2.4.2 of this Tariff) and any of the features listed in Section B below for the following single monthly rate to Residential Customers. Core 100 also includes 100 minutes of switched outbound intraLATA and interLATA MTS long distance per month for no additional charge. Usage over 100 minutes a month will be billed at the rates listed in Section C below.

| | Monthly Rate |
|--------------------|--------------|
| - All Access Areas | Zone |
| | 1 - \$114.95 |
| | 2 - \$114.95 |
| | 3 - \$114.95 |

These monthly charges apply in addition to all other applicable charges, including those established in Section 2.3.6 of this Tariff.

B. <u>Features Available</u>*

| Automatic Callback | Call Waiting |
|--------------------|-------------------|
| Line Shield | Call Screening |
| Busy Line Transfer | Multi Ring |
| Alternate Answer | Repeat Dialing |
| Caller ID w/ Name | Speed Dialing |
| Call Forwarding | Three Way Calling |

^{*} Local features are not included in all areas; some restrictions apply

4.5 OPTIONAL EXCHANGE ACCESS SERVICE ENHANCEMENT FEATURES

4.5.1 General

Services in this section may be purchased in addition to a Company-provided Exchange Access Service. These features are available only when purchased in combination with a Company-provided Exchange Access Service.

4.5.2 <u>Identification of Out Dialed Calls (IOD) Service</u>

Central Office facilities to provide identification of station numbers of a PBX system on outgoing long distance calls will be provided at the following monthly rates. These monthly rates are in addition to the rates and charges applicable for PBX trunks:

| | Monthly Price |
|--------------------------------------|---------------|
| First 10 trunks, or fraction thereof | \$474.15 |
| Each additional trunk | \$48.45 |

4.5.3 Optional Exchange Access Service Enhancement Features -- Rate Schedule

| | Residence | <u>Business</u> |
|--|-------------------------------------|------------------|
| ALTERNATE ANSWERING | \$0.75 | \$0.75 |
| AUTOMATIC CALL BACK, per line per call | \$7.25 \$0.97 1 Feature Point | \$7.25 \$0.97 |
| BUSY LINE TRANSFER | \$0.75 | \$0.75 |
| CALL CONTROL | \$7.95 | \$7.95 |
| CALL FORWARDING - Variable | \$9.35 1 Feature Point | \$9.35 |
| CALL SCREENING | \$4.00 1 Feature Point | \$4.00 |
| CALL TRACE (per activation) | \$3.50 | \$3.50 |
| CALL WAITING | \$4.00 1 Feature Point | \$5.00 |

| | Residence | Business |
|--|----------------------------|------------------|
| CALLER ID | \$6.10 | \$6.50 |
| CALLER ID WITH NAME | \$8.00 2 Feature Points | \$8.50 |
| CALLER ID BLOCK per line per use | \$1.00 \$0.00 | \$1.00 \$0.00 |
| CUSTOMER CONTROL OPTION | \$1.00 | \$1.00 |
| EASY CALL | \$1.50 | \$1.50 |
| INTERNATIONAL BLOCKING Non-recurring, per line and trunk | \$3.50 | \$3.50 |
| MESSAGE WAITING TONE | \$0.25 | \$0.25 |
| MULTI - RING SERVICE 1st Number Feature Points | \$4.00 1 Feature Point | \$4.00 |
| 2d Number | \$2.00 | \$3.95 |

| | Residence | <u>Business</u> |
|---|-------------------------------------|--|
| NAME AND NUMBER DELIVERY Per Delivered Message | \$0.75 | \$0.75 |
| PRIVACY MONITOR, per line | \$5.00 1 Feature Point | Not Available |
| REMOTE CALL FORWARDING | N/A | Zone 1 - \$34.95Zone 2 - \$34.95 Zone 3 - \$34.95 |
| REPEAT DIALING, per line per call | \$4.00 \$1.50 1 Feature Point | \$4.00 \$1.50 |
| SPEED CALL THREE-WAY CALLING | \$4.00 1 Feature Point | \$4.00 |
| Per Line Per Call | \$4.00 \$1.50 1 Feature Point | \$4.00 \$1.50 |
| WAIT AND SEE | \$0 | \$0 |
| 900 SPECIAL ACCESS CODE BLOCKING | \$0.00 | \$0.00 |
| 976 PREFIX BLOCKING SERVICE | \$0.00 | \$0.00 |

4.5.4 Other Optional Exchange Access Service Enhancement Features

4.5.4.1 CoreComm Link

E. Rates and Charges

1. Service Elements

a. Plan 1

The following rates are applicable to all CoreComm Link Customers except those who qualify under Section 4.5.4.1.E.1.b.

| Description | Non-recurring Charge | Monthly |
|---|----------------------|--|
| Service Establishment Charge - All Access Areas, Per Order | \$50.00 | |
| Access Area B, Rate per Line Package with 1 Line Package with 2 - 4 Lines Package with 5 - 11 Lines Package with 12 - 19 Lines Package with 20 - 30 Lines | | \$9.00 \$9.00 \$9.00 \$9.00 \$9.00 |
| Access Area C, Rate per Line Package with 1 Line Package with 2 - 4 Lines Package with 5 - 11 Lines Package with 12 - 19 Lines Package with 20 - 30 Lines | | \$9.00 \$9.00 \$9.00 \$9.00 \$9.00 |
| Access Area D, Rate per Line Package with 1 Line Package with 2 - 4 Lines Package with 5 - 11 Lines Package with 12 - 19 Lines Package with 20 - 30 Lines | | \$9.00 \$9.00 \$9.00 \$9.00 \$9.00 |
| Optional Features Speed Calling – Thirty - Including initial access to list, per list Ground Start Line - Per line Caller ID with Name * | \$10.75 \$5.00 | \$6.80 \$10.00 |

^{*} Charges apply as provided in Section 4.5.3 of this tariff

b. Plan 2

The following rates are applicable only to Customers who were previously subscribed to Zone Centrex service (also referred to as Citywide or Telemanagement Centrex) and who convert to CoreComm Link service.

| Description | Non-recurring Charge | Monthly |
|--|----------------------|---------|
| Service Establishment Charge | | |
| - All Access Areas, Per Order | \$50.00 | |
| Access Area B, Rate per Line | | |
| Package with 1 Line | | \$6.00 |
| Package with 2 - 4 Lines | | \$6.00 |
| Package with 5 - 11 Lines | | \$6.00 |
| Package with 12 - 19 Lines | | \$6.00 |
| Package with 20 - 30 Lines | | \$6.00 |
| Access Area C, Rate per Line | | |
| Package with 1 Line | | \$6.00 |
| Package with 2 - 4 Lines | | \$6.00 |
| Package with 5 - 11 Lines | | \$6.00 |
| Package with 12 - 19 Lines | | \$6.00 |
| Package with 20 - 30 Lines | | \$6.00 |
| Access Area D, Rate per Line | | |
| Package with 1 Line | | \$6.00 |
| Package with 2 - 4 Lines | | \$6.00 |
| Package with 5 - 11 Lines | | \$6.00 |
| Package with 12 - 19 Lines | | \$6.00 |
| Package with 20 - 30 Lines | | \$6.00 |
| Optional Features | | |
| Speed Calling – Thirty | | |
| - Including initial access to list, per list | \$10.75 | \$6.80 |
| Ground Start Line | + = | + 0 |
| - Per line | \$5.00 | \$10.00 |
| Caller ID with Name * | · | |

^{*} Charges apply as provided in Section 4.5.3 of this tariff

2. Other Applicable Charges and Payments

Charges apply for subsequent moves and changes as specified in Section 4.1 of this tariff.

4.5.5 Advantage Custom Business Service

Advantage Custom Business Service is an optional service arrangement of Electronic Switching System (ESS) and Data Management System (DMS) central office features which enables an individual exchange access customer served by the same central office to combine up to 6 exchange access lines into a group.

The following rates and charges are for the Advantage System only, and are in addition to the applicable service and equipment charges, monthly rates and non-recurring charges for access lines and other services or equipment with which they are associated.

| | Non-recurring | Monthly |
|--|---------------|------------------|
| | <u>Charge</u> | Recurring Charge |
| Service Established Charge, per system Advantage System per line (includes Call Hold, | \$43.00 | Not Avail. |
| Call Pickup, Transfer, Conferencing and Intercom) | Not Avail. | \$6.25 |
| Optional Features | | |
| - Call Waiting, per line | Not Avail. | \$1.50 |
| - Call Forwarding, per line | Not Avail. | \$0.75 |
| - Alternate Answering, per line | Not Avail. | \$2.00 |
| - Convenience Dialing 30 (including | | |
| initial access to list) per list | Not Avail. | \$6.80 |
| - Each additional access to list | Not Avail. | \$1.30 |
| - Feature Package 1 (including Call Forwarding | | |
| and Call Waiting) per line | Not Avail. | \$2.00 |
| - Feature Package 2 (including Call Forwarding | | |
| and Alternate Answering) per line | Not Avail. | \$2.40 |
| - Deluxe Call Transfer, per line | Not Avail. | \$3.15 |
| - Distinctive Ringing, per line | Not Avail. | \$4.00 |

Advantage Custom Business Service is available only on a "grandfathered" basis to Customer who were acquired by the Company from USN Communications, Inc. and who were subscribers to the service on April 2, 2001.

4.5.6 <u>CoreComm Advantage Line Plus Service</u>

A. Description

CoreComm Advantage Line Plus Service is a service arrangement offered from a Company central office which provides an enhanced feature capability to individual business exchange access customers requiring 2 to 30 lines. CoreComm Advantage Line Plus Service provides a set of standard features on each line. Optional features are also available at additional charge. CoreComm Advantage Line Plus Service is available only on a "grandfathered" basis to Customer who were acquired by the Company from USN Communications, Inc. and who were subscribers to the service on April 2, 2001.

B. Prices

The following prices are applicable to standard installations of USN Advantage Line Plus Service and are in addition to all other charges for exchange access lines or other associated services and equipment necessary to provide telecommunications service.

1. Service Elements

| | Non-Recurring Charge | Monthly Charge |
|--|----------------------|----------------|
| | | |
| Service Establishment Charge | \$50.00 | |
| Access Area B, Per Package, Per Line | | \$9.00 |
| Access Area C, Per Package, Per Line | | \$9.00 |
| Access Area D, Per Package, Per Line | | \$9.00 |
| | | |
| Optional Features | | |
| Speed Calling – Thirty | | |
| - Including initial access to list, Per list | \$10.75 | \$6.80 |
| - Each additional access to list | \$5.00 | \$1.00 |
| Ground Start Line, Per Line | \$5.00 | \$10.00 |
| Caller ID | * | ** |
| Caller ID with Name | * | ** |

^{*} Charges apply as provided in Section 4.1.

2. Other Applicable Charges and Payments

- a. Subsequent Activity: Charges apply for subsequent moves and changes as specified for exchange access lines.
- b. Month to Month: The minimum period is one month, unless otherwise specified, or greater term commitment.

^{**} Charges apply as provided in Section 4.5.

4.5.7 CoreComm Unlimited Small Business

Monthly Rate*

CoreComm Unlimited Small Business is a business service that offers two types of bundled basic exchange access lines for customers with 1 to 5 lines. The Basic SOHO line bundle offers unlimited calling to parties located in the Customer's local usage service area (as established in Section 4.4.2 of this Tariff), unlimited Long Distance and Toll calling, and any of the features listed in Section B below for the following single monthly rate. The Additional Fax/Modem line includes unlimited calling to parties located in the Customer's local usage service area (as established in Section 4.4.2 of this Tariff), with \$0.050 cent per minute Long Distance and Toll Calling, and free hunting features.

| Line Type | Monthly Charg |
|------------------------|--------------------------------------|
| - Basic SOHO | Zone 1 - \$119.95 |
| | 2 - \$119.95 3 - \$119.95 |
| - Additional Fax/Modem | Zone 1 - \$119.95 2 - \$119.95 |
| - EUCL | 3 - \$119.95 \$46.95 |

These monthly charges apply in addition to all other applicable charges, including those established in Section 4.3.4 of this Tariff. Local Calling Plus charges also apply as provided in Section 4.4.3.2 of this Tariff.

^{*} Customers purchasing service under this plan or package may qualify for the Multi Product Credit. The Multi Product Credit provides Customers purchasing qualifying commercial telephone line package(s) at the same service address as an On Net 1.5mb or greater Circuit, including, but not limited to, the Essential products (VoIP), with an \$8 credit on the Customer's commercial telephone line package(s). In the event of On Net circuit disconnection, the Multi Product Credit shall no longer be available.

4.6.2 INTRALATA OPERATOR ASSISTANCE SURCHARGES

IntraLATA Operator Assistance Surcharges apply when a Customer subscribing to CoreComm's long distance service utilizes either an automated or live Company-provided operator for purposes of completing or billing a call. Operator Assistance Surcharges apply in addition to either local usage or long-distance usage services as identified in CoreComm's Tariff No. 3. Customers that do not subscribe to CoreComm's long distance service will be subject to the charges established by their long distance providers.

<u>IntraLATA Operator Assistance Surcharges — Rate Schedule</u>

| | Residence | <u>Business</u> |
|--------------------------|-----------|-----------------|
| STATION-TO-STATION | \$1.10 | \$1.10 |
| PERSON-TO-PERSON | \$3.00 | \$3.00 |
| BILLED TO A THIRD NUMBER | \$1.50 | \$1.50 |
| COLLECT CALLS | \$1.10 | \$1.10 |

| | Residence | <u>Business</u> |
|--|------------------|------------------|
| BUSY LINE VERIFICATION | \$1.00 | \$1.20 |
| BUSY LINE VERIFICATION AND INTERRUPT Verification Interrupt | \$1.00 \$1.30 | \$1.20 \$1.30 |
| INMATE CALLING SERVICE | | |
| Inmate Calling Service Operator Service Charges | | |
| Operator Handled Station-to-Station Collect Person-to-Person Collect | \$2.50 \$4.80 | \$2.50 \$4.80 |
| Inmate Calling Service Usage Rates | | |
| Per minute | \$0.36 | \$0.36 |
| LOCAL DIRECTORY ASSISTANCE CALL | | |
| When customer direct dials, each call When operator assisted, each call | \$1.50 \$0.45 | \$1.50 \$0.45 |
| NATIONAL DIRECTORY ASSISTANCE CALL | \$1.99 | \$1.99 |

| INFORMATION CALL COMPLETION | Residence | <u>Business</u> |
|-----------------------------|-----------|-----------------|
| Per Call Completed | \$0.25 | \$0.25 |

4.7.2 DIRECTORY SERVICES

Directory services allow Customers to customize the manner in which their Company assigned telephone numbers appear in published directory and/or are used by dialable directories and Company operators. This section applies only to services provided by the Company.

| <u>Directory Services Rate Schedule</u> | | |
|--|-----------|----------|
| | Residence | Business |
| ALPHABETICAL DIRECTORY LISTING | \$0.00 | \$0.00 |
| EXTRA LISTINGS | | |
| Per month, each listing: | \$5.00 | \$5.00 |
| NON-PUBLISHED | | |
| Per month, each listing: | \$1.10 | \$1.10 |
| EMERGENCY SERVICE Per successful attempt | \$1.80 | \$1.80 |

| | Residence | Business |
|--------------------------|-----------|-----------------|
| NON-LISTED | | |
| Per month, each listing: | \$1.10 | \$1.10 |

4.8.2 INTEGRATED SERVICES DIGITAL NETWORK (ISDN) SERVICES

General

Provides integrated voice/data communications capability for transmission and combination of circuit switched voice/data and packet switched data signals on an incoming and outgoing basis over a single ISDN line. This service requires two 64 Kbps "B" channels to transmit any combination of Circuit Switched Voice/Data or high speed Packet Switched Data and one "D" channel to carry network signaling and user originated Packet Switched data at speeds up to 9. Kbps (2B+D). It is available from specially equipped digital switching equipment located in the Company's central offices or in the offices of a Company affiliated supplier/carrier) and where facilities permit and where capacity is available within specified distances from the serving central office. A maximum of 8 devices may be connected directly to an ISDN line and only 2 of these devices are permitted to access the two "B" channels.

Integrated Services Digital Network (ISDN) Services -- Rate Schedule

| <u>Install Charge</u> | <u>Variable Term Option</u> Monthly Rate | | | |
|-----------------------|--|---------------|---------------|---------------|
| | <u>1</u> <u>36</u> <u>60</u> <u>8</u> | | | |
| IGDNI G O T | <u>Month</u> | <u>Months</u> | <u>Months</u> | <u>Months</u> |
| ISDN C.O. Term. | | NOT AVAILAI | DI E | |
| IGDN Di G G | | | | |
| ISDN Direct C.O. | | NOT AVAILAI | BLE | |
| Distance | | NOT AVAILAI | BLE | |
| Extension Charge | | | | |
| for Customers | | | | |
| beyond normal | | | | |
| transmission | | | | |
| range, per line | | | | |

Circuit Switched Voice Service

Element per "B" NOT AVAILABLE

Channel, equipped -Additional

Multiple Call NOT AVAILABLE

Appearances,

Each

-Secondary

Telephone NOT AVAILABLE

Numbers, Each

Circuit Switched
Data Service

Element per "B" NOT AVAILABLE

Channel, equipped

Alternate Circuit

Switched NOT AVAILABLE

Voice/Data

Packet Switched Data "B" Channel

Service Element NOT AVAILABLE

charge, per "B"

Channel equipped.

84

Months

GRANDFATHERED SERVICES FOR PREVIOUS CUSTOMERS OF CORECOMM NEWCO, INC.

<u>Install Charge</u>

<u>Variable Term Option</u>

<u>Monthly Rate</u>

<u>1</u>

<u>36</u>

<u>60</u>

<u>Months</u>

Months

Packet Switched

Data "D" Channel NOT AVAILABLE

Month

Service Element Charge per Data Communication Equipment

On Demand

Packet Switched NOT AVAILABLE

Data "B" Channel

Subsequent

Data and/or

Changes for Circ. Voice and/or Circuit Switched

Packet Switched Data

rearrangements to NOT AVAILABLE

add line

appearances or

move line or

feature

appearance, per

line, per occasion.

Additional Call NOT AVAILABLE

Offering, each

Intercom Calling NOT AVAILABLE

Station Controlled

Conferences -6 NOT AVAILABLE

Port

Effective June 1, 2021

| Install Charge | | Variable To | erm Option | |
|-----------------|--------------|---------------|---------------|---------------|
| | Monthly Rate | | | |
| | <u>1</u> | <u>36</u> | <u>60</u> | <u>84</u> |
| | <u>Month</u> | <u>Months</u> | <u>Months</u> | <u>Months</u> |
| Message Waiting | | NOT AVAILA | BLE | |
| Indicator, Each | | | | |

Effective June 1, 2021

4.8.3 National ISDN Direct -- Rate Schedule

General National ISDN Direct is available as a residence service. Rates charged for the loop, EUCL, usage and optional features are based upon the customer's class of service. Service ordering and line connection charges are also applicable for the establishment of National ISDN Direct.

Install Charge Monthly Rate

National ISDN NOT AVAILABLE

Central Office Termination

National ISDN NOT AVAILABLE

- Direct Line

Circuit Switched Service NOT AVAILABLE

Element per "B" Channel

Additional Call Offering NOT AVAILABLE

Additional Multiple Call NOT AVAILABLE

Appearances, Each

Intercom Calling NOT AVAILABLE

Secondary Telephone NOT AVAILABLE

Numbers, Each

Station Controlled NOT AVAILABLE

Conferences - 6 Port

Message Waiting NOT AVAILABLE

Indicator, Each

Packet Switched Data "B" NOT AVAILABLE

Channel

On Demand Packet NOT AVAILABLE

Switched Data "B"

Channel

Install Charge Monthly Rate
NOT AVAILABLE

Packet Switched Data "D" Channel

Subsequent Changes for Circ. Voice and/or Circuit Switched Data and/or Packet Switched Data rearrangements to add line appearances or move line or feature appearance, per line, per occasion

NOT AVAILABLE

Effective June 1, 2021

4.8.4.2 ISDN Prime

General

ISDN Prime is a digital business service that provides PBX equipment and host computer access to a wide variety of switched services. Each ISDN Prime will allow connection of switched services via a single central office connection. This service allows PBX equipment and host computer type devices to connect to central office services in bulk quantity, rather than on a line by line or service by service basis. An ISDN Prime provides 23B channels at 64Kbps and one D channel for control at 64Kbps--also call a channelized Ti line.

ISDN Prime -- Rate Schedule

| Install Charge | | Variable Ten Monthly | - | |
|--|-----------|-------------------------|---------------|---------------|
| | 1 Month | <u>36</u> | <u>60</u> | 84 Months |
| ISDN Prime, each | NOT AVAII | <u>Months</u> LABLE | <u>Months</u> | <u>Months</u> |
| Backup "D" Channel | NOT AVAII | LABLE | | |
| System Intercom - Communication Service, Per Trunk Group | NOT AVAII | LABLE | | |
| -Circuit Switched Voice or Data | NOT AVAII | LABLE | | |
| -Packet Switched Data | NOT AVAII | LABLE | | |
| "B" Channel Packet Switched Data, per "B" Channel | NOT AVAII | LABLE | | |
| Call by Call for FX | NOT AVAII | LABLE | | |
| Call by Call for Tie Lines | NOT AVAII | LABLE | | |

| Install Charge | | Variable Terr Monthly | | |
|---|-----------|--------------------------|---------------|--------|
| | <u>1</u> | <u>36</u> | 60 | 84 |
| | Month | <u>Months</u> | <u>Months</u> | Months |
| Network Ring Again per ISDN Prime Equipped | NOT AVAII | LABLE | | |
| Network Name Display per ISDN Prime Equipped | NOT AVAII | LABLE | | |
| Changes or Additions to One or More Channel Groups on a Single Ameritech ISDN Prime, per occasion, per channel | NOT AVAII | LABLE | | |
| group | | | | |

4.8.5 Grandfathered Integrated Services Digital Network (ISDN) Service

A. General

Provides integrated voice/data communications capability for transmission and combination of circuit switched voice / data and packet switched data signals on an incoming and outgoing basis over a single ISDN line. This service requires two 64 Kbps "B" channels to transmit any combination of Circuit Switched Voice/Data or high speed Packet Switched Data and one "D" channel to carry network signaling and user originated Packet Switched data at speeds up to 9.6 Kbps (2B+D). It is available from specially equipped digital switching equipment located in the Company's central offices (or in the offices of a Company affiliated supplier/carrier) and where facilities permit and where capacity is available within specified distances from the serving central office. A maximum of 8 devices may be connected directly to an ISDN line and only 2 of these devices are permitted to access the two "B" channels.

ISDN Services provided under Section 4.8.5 are available only on a "grandfathered" basis to Customers who were acquired by the Company from USN Communications, Inc. and who were subscribers to the service on April 2, 2001.

B. Grandfathered Integrated Services Digital Network (ISDN) Services -- Rate Schedule

| | Reside | nce | Bus | siness |
|--|--------------|---------------|---------|---------------|
| | Monthly Rate | Install | Monthly | Non-Recurring |
| | | <u>Charge</u> | Rate | <u>Charge</u> |
| ISDN C.O. Termination | Not Avail. | Not Avail. | \$7.10 | \$50.00 |
| ISDN Direct C.O. | Not Avail. | Not Avail. | \$12.10 | \$0.00 |
| ISDN National C.O. Termination | | | | |
| Access Area B | Not Avail. | Not Avail. | \$15.00 | \$50.00 |
| Access Area C | Not Avail. | Not Avail. | \$15.00 | \$50.00 |
| Access Area D | Not Avail. | Not Avail. | \$15.00 | \$50.00 |
| Distance Extension Charge for Customers beyond normal transmission | | | | |
| range, per line | Not Avail. | Not Avail. | \$26.00 | Not Avail. |
| Circuit Switched Service Element per "B" Channel, VOICE | Not Avail. | Not Avail. | \$3.66 | \$18.30 |
| Circuit Switched Service Element per "B" Channel, DATA | Not Avail. | Not. Avail. | \$9.76 | \$18.30 |

| | Residence | | Buiness | |
|---|------------------------|--------------------------|------------------------|-------------------------|
| | Monthly <u>Rate</u> | Install <u>Charge</u> | Monthly <u>Rate</u> | Non-Recurring Charge |
| Additional Call Offering | Not Avail. | Not Avail. | \$3.05 | \$6.10 |
| Additional Multiple Call Appearances, Each | Not Avail. | Not Avail. | \$2.44 | \$6.10 |
| Intercom Calling | Not Avail. | Not Avail. | \$3.05 | \$6.10 |
| Secondary Telephone Numbers, Each | Not Avail. | Not Avail. | \$2.44 | \$6.10 |
| Station Controlled Conferences - 6 Port | Not Avail. | Not Avail. | \$17.08 | \$18.30 |
| Message Waiting Indicator, Each | Not Avail. | Not Avail. | \$3.05 | \$6.10 |
| On Demand Packet Switched Data "B" Channel | Not Avail. | Not Avail. | \$24.40 | \$61.00 |
| Alternate Circuit Switched Voice/Data Per "B" Channel Equipped | Not Avail. | Not Avail. | \$10.98 | \$18.30 |
| Packet Switched Data "B" Channel Standard Capabilities and features, per "B" Channel equipped. | Not Avail. | Not Avail. | \$103.70 | \$122.00 |
| Packet Switched Data "D" Channel Standard Capabilities and features, Per data communication equipment (DCE). | Not Avail. | Not Avail. | \$4.88 | \$18.30 |
| Subsequent Changes for Circ. Voice and/or Circuit Switched Data and/or Packet Switched Data rearrangements to add line appearances or move line or feature appearance, per line per occasion. | Not Avail. | Not Avail. | No Charge | \$18.30 |

Effective June 1, 2021

69

GRANDFATHERED SERVICES FOR PREVIOUS CUSTOMERS OF CORECOMM NEWCO, INC.

4.9 <u>Trunking Services</u>

4.9.3 Direct Inward Dial (DID) Service

C. <u>Rate Schedules</u>

Rates for DID Service are as established in Section 4.9.4 for DID with Call Transfer Service

4.9.4 <u>Direct Inward Dial (DID) with Call Transfer Service</u>

C. Trunk Rate Schedule

Monthly charges per trunk are as follows:

| Monthly Charge |
|----------------|
| \$33.00 |
| \$29.00 |
| \$27.00 |
| \$25.00 |
| |

In addition to the monthly trunk charge, trunk termination charges apply for termination at the wire center.

| Term Commitment | Monthly Charge | Non-Recurring Charge |
|-------------------|--------------------|----------------------|
| Month-to-Month | \$22.50 | \$220.00 |
| 1 Year | \$21.50 | \$220.00 |
| 2 Years | \$21.50 | \$220.00 |
| 3 Years | \$21.50 | \$220.00 |
| 1 Year 2 Years | \$21.50 \$21.50 | \$220.00 \$220.00 |

Also in addition, other charges apply as provided in Section 4.3.4 of this tariff.

Effective June 1, 2021

4.9.4 <u>Direct Inward Dial (DID) with Call Transfer Service</u> (continued)

D. <u>Feature Rate Schedule</u>

| <u>Feature</u> | Monthly Charge | Non-Recurring Charge |
|---|------------------|-------------------------|
| Station Numbers (Block of 20) | | |
| Month to Month | \$3.25 | \$174.50 |
| 1 Year, 2 Year or 3 Year Term | \$3.00 | \$174.50 |
| Additional Groups of 20 Numbers (same order) Month to Month 1 Year, 2 Year or 3 Year Term | \$3.25 \$3.00 | \$54.00 \$54.00 |
| Individual Station Numbers | \$0.23 | N/A |
| DID with Call Transfer | \$21.50 | \$175.00 |
| Special Interface Service (SIS) | \$3.50 | \$60.00 |

E. Changes to DID Trunks

| <u>Change Type</u> | Non-Recurring Charge |
|--|----------------------|
| Change existing DID to DID with Call Transfer or vice versa, per trunk group | \$330.00 |
| Change in Outpulsing, Start Dial or Signal Type, per trunk group | \$41.55 |
| Change or redesign in signaling or transmission interface, per occasion | \$150.00 |

4.9.4 <u>Direct Inward Dial (DID) with Call Transfer Service</u> (continued)

F. <u>Miscellaneous Charges</u> *

1. <u>CSR/Conversion Charges</u>

Charge Per CSR \$20.00

2. Order Activity Charges

| Order Activity | Order Type | Non-Recurring Charge |
|---|---------------|----------------------|
| Service Order Charge, per location, per occasion (Applies for receiving, recording, and processing information necessary to execute a Customer's request for service.) | New | \$24.50 |
| Central Office Connection Charge, per termination | New or Change | \$14.50 |
| Line Connection Charge, per termination | New or Change | \$10.35 |

3. <u>Service Conversion Charges</u>

| Conversion Type | Non-Recurring Charges |
|---|-----------------------|
| Miscellaneous Service/Feature Custom Calling Features and Advanced Custom Calling Features (to establish one or more feature, per occasion) | \$9.00 |
| Directory Services From Listed to Private or Semi-private From Semi-private to Private | \$17.90 |
| Telephone Number Change – Individual Number | \$41.55 |
| Service restored after temporary suspension | \$75.85 |
| Service restored after temporary interception | \$87.70 |

4.9.5 Two-Way and Direct Outward Dial (DOD) Service

B. <u>Trunk Rate Schedule</u>

Monthly charges per trunk are as follows:

| Term Commitment | Monthly Charge |
|-----------------|----------------|
| Month-to-Month | \$33.00 |
| 1 Year | \$29.00 |
| 2 Years | \$27.00 |
| 3 Years | \$25.00 |

In addition to the monthly charges, other charges apply as provided in Section 4.3.4 of this tariff.

C. <u>Usage Rate Schedule</u>

1. Message Rate Pricing

| Term Commitment | Charge Per Call |
|-----------------|-----------------|
| Month-to-Month | \$0.0800 |
| 1 Year | \$0.0775 |
| 2 Years | \$0.0750 |
| 3 Years | \$0.0700 |

2. <u>Measured Rate Pricing</u>

| | Peak (8 am up to but not including 9 pm) | |
|--------------|--|------------------------|
| <u>Band</u> | 1 st Minute | Each Additional Minute |
| \mathbf{A} | \$0.030 | \$0.0075 |
| В | \$0.035 | \$0.0110 |
| C | \$0.039 | \$0.0156 |
| | | |
| | Off-Peak (9 pm up to but not including 8 am) | |
| | 1 st Minute | Each Additional Minute |
| A | \$0.0160 | \$0.0038 |
| В | \$0.0180 | \$0.0060 |
| C | \$0.0195 | \$0.0079 |

The following mileage bands apply for measured rate services:

| Band | <u>Mileage</u> |
|--------------|----------------|
| \mathbf{A} | 0 - 10 |
| В | 11 - 22 |
| C | 22 + |

4.9.5 Two-Way and Direct Outward Dial (DOD) Service (continued)

D. <u>Features Rate Schedule</u>

The following non-recurring charges are applied on a per trunk basis:

| <u>Feature</u> | Non-Recurring Charge |
|---|----------------------|
| Night Terminal Number Equipment | \$23.75 |
| Night, Sunday, and Holiday Listing | \$2.35 |
| Night Terminal Arrangement - Key Controlled | \$26.35 |
| Control Channel and Stop Hunt Key - ESS CO | N/A |
| Make Busy Arrangement | \$80.00 |

E. Operator Assistance Rate Schedule

The following operator assistance services are described in Section 4.6 of this tariff.

| Operator Service | Non Recurring Charge |
|---|----------------------|
| Station-To-Station | \$1.10 |
| Person-To-Person | \$3.00 |
| 3 rd Party Billing | \$1.50 |
| Collect Calls | \$1.10 |
| Calling Card Assistance | |
| Automated | \$0.45 |
| Non-Automated | \$1.10 |
| Busy Line Verification | \$1.20 |
| Busy Line Interrupt | \$1.60 |
| Directory Assistance – Operator Assisted | \$0.45 |
| Directory Assistance – Local Direct Dial | \$1.50 |
| Directory Assistance – National Direct Dial | \$1.99 |
| Information Call Completion | \$0.25 |
| | |

4.9.5 Two-Way and Direct Outward Dial (DOD) Service (continued)

F. <u>Miscellaneous Charges</u> *

1. <u>CSR/Conversion Charges</u>

Charge Per CSR \$20.00

2. Order Activity Charges

| Order Activity | Order Type | Non-Recurring Charge |
|---|------------------|----------------------|
| Service Order Charge, per location, per occasion (Applies for receiving, recording, and processing information necessary to execute a Customer's request for service.) | New | \$24.50 |
| Central Office Connection Charge, per termination | New or Change | \$14.50 |
| Line Connection Charge, per termination | New or Change | \$10.35 |

4.9.5 Two-Way and Direct Outward Dial (DOD) Service (continued)

3. <u>Service Conversion Charges</u>

| Conversion Type | Non-Recurring Charges |
|---|-----------------------|
| Miscellaneous Service/Feature Custom Calling Features and Advanced Custom Calling Features (to establish one or more feature, per occasion) | \$9.00 |
| Directory Services From Listed to Private or Semi-private From Semi-private to Private | \$17.90 |
| Telephone Number Change – Individual Number | \$41.55 |
| Service restored after temporary suspension | \$75.85 |
| Service restored after temporary interception | \$87.70 |
| Service restored after temporary denial but prior to completion of order to discontinue service, per | |
| service or system | \$26.55 |
| Class of Service Change Message Rate to Measured Rate and vice versa Change from Ground to Loop Start Change Type of Trunk | \$41.55 |

4.9.7 <u>Grandfathered Trunking Services</u>

Trunking Services provided under Section 4.9.7 are available only on a "grandfathered" basis to Customers who were acquired by the Company from USN Communications, Inc. and who were subscribers to the service on April 2, 2001.

4.9.7.1 Grandfathered Direct Inward Dial (DID) Service

A. General

DID is a service which permits incoming dialed calls to be dialed directly by a calling party station associated with a switching system located on the Carrier's Customer Premises. These lines support inbound calling traffic only.

B. Direct Inward Dial Service -- Rate Schedule

| | Business | |
|--|-----------|---------------|
| | Monthly | Non-Recurring |
| | Rate | <u>Charge</u> |
| DID Rates and Charges | | |
| Each group of 20 Assigned DID | | |
| numbers or fraction thereof | \$3.45 | \$174.20 |
| Each Individual DID Number | \$0.25 | No Charge |
| Each DID trunk termination in central office, per trunk | \$25.00 | \$275.00 |
| In lieu of the NRC charge paragraph. Per additional group of 20 DID numbers provided on the same occasion as an initial group of | | |
| 20 DID numbers | No Charge | \$54.20 |

4.9.7.2 Grandfathered 2-Way Direct Inward Dialing (DID) with Call Transfer

A. General

2-Way Direct Inward Dialing (DID) with Call Transfer is a service that permits incoming calls to reach Customer provided equipment, without the assistance of an attendant, and allows the transfer of those calls to another line. Touch-Tone is a standard feature of this service.

B. 2-Way Direct Inward Dialing (DID) with Call Transfer -- Rate Schedule

| | Business | |
|---|------------|---------------|
| | Monthly | Non-Recurring |
| Service Establishment/Monthly Charges | Rate | <u>Charge</u> |
| With initial DID Service, per Trunk Group | No Charge | \$175.00 |
| Subsequent to Establishment of DID Service - per Trunk Group | No Charge | \$175.00 |
| 2-Way DID Trunk with Call Transfer - per Trunk Group | \$23.00 | No Charge |
| Conversions | | |
| Of entire DID Trunk Group to 2-Way DID or entire 2-Way DID Trunk Group to DID | No Charge | \$300.00 |
| Of individual DID trunks to a new 2-Way DID Trunk Group, per Trunk Group | No Charge | \$175.00 |
| Of individual Trunks between established Trunk Groups | Not Avail. | Not Avail. |
| Two-way DID Trunk with call transfer | \$23.00 | No Charge |
| Change in Outpulsing, Start Dial, or Signal Type, per trunk group | No Charge | \$41.55 |
| Change or redesign in Signaling or transmission Interface, per Occurrence | No Charge | \$150.00 |

4.9.7.3 Grandfathered Identification of Outdialed Calls (IOD) Service:

Central office facilities to provide identification of station numbers of a DIMENSION PBX system on outgoing long distance calls will be provided at the following monthly rates. These monthly rates are in addition to the rates and charges applicable for PBX trunks:

| | Monthly Rate |
|---------------------------------------|--------------|
| First 10 trunks, or fractions thereof | \$474.15 |
| Each additional trunk | \$48.45 |

4.11.2 Digital Transport Service

General

Digital Transport Service provides a cost effective method of delivering voice grade service from the serving Central Office to a customer's premises. DTS is provided exclusively with DSI Service (1.544 Mbps) and the 24 channels may be used for the transport of certain access lines for termination on customer-provided-equipment equipment. DTS support Direct Inward Dialing (DID) Service, and PBX Trunks. Touch-Tone is a standard feature of DTS. DTS is an exchange service. Rates and charges for DS1 (1.544 Mbps) Service are specified Private Line Section of this tariff, and are in addition to the charges specified in this section. Calls will be subject to usage charges for the services provisioned on the DTS channels. Termination charges are applicable for the DS 1 Private Line Service and Central Office Multiplexing, as specified in this tariff

<u>Digital Transport Service -- Rate Schedule</u>

| | Monthly Rate | Non-Recurring Charge |
|---|--------------|----------------------|
| Integrated Service Arrangement | | |
| DS1 Facility | | |
| Fixed Charge/Month/Circuit | \$570.00 | |
| Per Mile Charge/Month | \$17.30 | |
| Digital Interface Termination | \$250.00 | |
| PBX Trunk Termination (Channel) | | |
| PBX Trunk | \$11.25 | |
| DID Trunk | \$30.15 | |
| Toll Terminal | \$4.50 | |
| Dedicated 800 | \$9.00 | |
| Installation Charge, per circuit | | \$600 |
| Subsequent Addition! Rearrangement Charge (per trunk termination) | | \$10.00 |

| | Monthly Rate | Non-Recurring Charge |
|---|--------------|----------------------|
| Non-Integrated Service Arrangement | | |
| DSI Facility | | |
| Fixed Charge/Month/Circuit | \$570.00 | |
| Per Mile Charge/Month | \$17.30 | |
| Central Office Multiplexing | \$700.00 | \$700.00 |
| Per Trunk Termination (Channel) | | |
| PBX Trunk | \$11.25 | |
| DID Trunk | \$30.15 | |
| Toll Terminal | \$4.50 | |
| Dedicated 800 | \$22.10 | |
| Installation Charge, per circuit | | \$600.00 |
| Subsequent Addition/Rearrangement Charge per trunk termination | | \$10.00 |

4.12.2 Foreign Exchange Service

General

Foreign District service is a service by which a Customer can be provided the Company's Exchange Access Service from a district other than the one in which the Customer's premise is located. When a Customer subscribes to Foreign District service he/she will be considered to reside in the district he/she has chosen as the Foreign District for purposes of rating and billing the Company's services.

<u>Foreign Exchange Service — Rate Schedule</u>

The rate for Foreign District Service is (1) the usage rate in effect in the Foreign District for the class of service furnished, (2) the access rate for the access area in which the Customer is physically located; and (3) the following mileage charges:

| | Monthly Rate | |
|--|------------------|-----------------|
| | <u>Residence</u> | <u>Business</u> |
| Area Function Charge | \$14.95 | \$14.95 |
| Interexchange Circuit per mile | \$4.15 | \$4.15 |
| Interexchange Circuit Service Terminal | \$41.80 | \$41.80 |

4.13 Miscellaneous Services

| Wiscenaneous Services | Month Residence | ly Rate Business |
|--|--------------------|---------------------|
| TOLL RESTRICTION SERVICE | | |
| (Toll Restriction Service will not allow inter-MSA 1+, 0+, 0-, 10XXX, or 700 calls to be completed.) - per line equipped - Non-recurring charge - Monthly charge (Toll billing exception which prevents third number billed and collect call is also a Customer option.) | \$12.00 \$5.95 | \$29.40 \$64.90 |
| INTERCEPT REFERRAL EXTENSION SERVICE | \$6.00 | \$6.00 |
| (Provides notification to calling parties about changes in the status of the called party's telephone line. [An Add/Change Charge applies to add or change the length of months requested.]) | | |
| TEMPORARY INTERCEPT | \$9.45 | \$15.85 |
| (Enables a Customer to have incoming calls intercepted for 1 month. [Regular Exchange Access Service billing continues and an Add/Change charge applies]). | | |
| TOLL DISCONNECTION | Non-Recurr | ring Charge |
| DePICing | \$5.00 | \$5.00 |
| (Available to toll service providers to disconnect customers' toll service for nonpayment, as detailed | | |

(Available to toll service providers to disconnect customers' toll service for nonpayment, as detailed in Carrier's Local Exchange Services Tariff, P.U.C.O. No. 3, Section 2.7.8.)

4.14.1 <u>Basic Centrex Service</u>

4.14.1.1 General

CoreComm Basic Centrex Service is a local exchange telecommunications service, provided by a telecommunications system located in a central office which controls the switching of (a) calls from the exchange network to Centrex lines and (b) calls from the Centrex lines to the exchange network, intercommunicating calls between CoreComm Centrex lines. Customers may subscribe to Basic Centrex Service on a month to month, 3 year or 5 year term.

4.14.1.2 <u>Basic Centrex Service -- Monthly Station Rate Schedule</u>

D. Line Rate Schedule

Monthly Centrex line rates depend upon the term commitment selected by the Customer.

| Term Commitment | Monthly Charge, Per Line |
|-----------------|--------------------------|
| Month to Month | Zone |
| | 1 - \$31.95 |
| | 2 - \$33.95 |
| | 3 - \$33.95 |
| 3 Years | \$31.00 |
| 5 Years | \$30.00 |

Additionally, all accounts shall be assessed the following monthly system charge:

Monthly System Charge

| Per Account | Zone 1 - \$14.90 |
|-------------|------------------|
| | Zone 2 - \$19.90 |
| | Zone 3 - \$19.90 |

In addition to the above monthly line and system charges, certain charges apply as provided in Section 4.14.1.2.E.

E. Additional Charges Applicable to Centrex Lines

In addition to the monthly charges for Centrex lines described in Section 4.14.1.2.D preceding, the following charges apply to each Centrex line, unless otherwise specified:

1. The End User Common Line Charge, which shall be charged to multiple Centrex lines at the following per line rate:

Per line: \$5.97

- 2. The 911 Telecommunications Service Surcharge, which shall be a direct pass through at cost.
- 3. Any applicable municipal, state, or federal taxes, franchise fees or other charges.
- 4. Casual traffic charges that are derived from third party calls (e.g., 10XXX, 900/976, third party calls initiated by the Customer through the Company's system) and trafficked over the Company's system.

F. Station Numbers

The following monthly and non-recurring charges apply for Centrex station numbers:

| | Monthly Charge | Non-Recurring Charge |
|--|------------------|----------------------|
| Station Numbers (block of 20) - Month to Month Term - 3 or 5 Year Term | \$3.25 \$3.00 | \$174.50 \$174.50 |
| Add'l Groups of 20 Numbers (same order) | | |
| - Month to Month Term | \$3.25 | \$54.00 |
| - 3 or 5 Year Term | \$3.00 | \$54.00 |
| Individual Station Numbers | \$0.23 | N/A |

4.14.1.3 Local Usage Service

Local usage services provide Customers subscribing to the Company's Basic Centrex Service the ability to place calls to and receive calls from parties located in the Company's local usage service area.

A. <u>Local Usage Service -- Rate Schedule</u>

1. <u>Message Rate Service</u>

Message Rate Service charges apply on a per call basis, depending upon the term commitment selected by the Customer:

| Term Commitment | Message Rate |
|-----------------|--------------|
| Month to Month | \$0.0800 |
| 3 Years | \$0.0725 |
| 5 Years | \$0.0700 |

4.14.1.4 Miscellaneous Centrex Charges

B. <u>CSR/Conversion Charges</u>

The following charge applies for the conversion of existing local lines from he incumbent local exchange carrier:

| Term Commitment | Charge Per CSR |
|-----------------|----------------|
| Month to Month | \$20.00 |
| 3 Years | \$20.00 |
| 5 Years | \$20.00 |

C. <u>Installation Charges</u>

The following charge applies per Customer line installed:

| Term Commitment | Charge Per Line |
|-----------------|-----------------|
| Month to Month | \$4.00 |
| 3 Years | \$4.00 |
| 5 Years | \$4.00 |

D. Order Activity Charges

| Order Activity | Non-Recurring Charge |
|--|----------------------|
| Service Order Charge New, per location, per occasion (Applies for receiving, recording, and processing information necessary to execute a Customer's request for service.) | \$40.00 |
| Service Order Charge Change, per location, per occasion (Applies to changes to existing Centrex lines/systems, in addition to any other applicable charges.) | \$30.00 |
| Central Office Connection Charge New or Change, per station terminated | \$1.75 |
| Line Connection Charge New or Change, per station terminated | \$3.35 |

E. <u>Service Establishment Charges</u>

The following charge applies to establish any new CoreComm Centrex system and to relocate any entire CoreComm Centrex system, unless the Customer's location is within the same Central Office:

| \$275.00 |
|----------------------|
| \$275.00 |
| \$275.00 |
| \$275.00 \$275.00 |
| |

F. Service Conversion Charges

Service Conversion Charges apply depending upon the type of conversion to CoreComm Centrex Services requested by the Customer:

| Conversion Type | Applicable Charges |
|---|---|
| Conversion configured exactly the same as the Customer's Centrex Service with its prior carrier | Service Order ChargeSystem Conversion Charge waiving the NRC |
| Conversion, plus Customer's addition (via Centrex Mate) of new or additional features | Service Order ChargeSystem Conversion Charge |
| Conversion, plus addition of new or additional features as programmed by CoreComm | Č |

System Conversion Charges

| <u>Lines</u> | Monthly Charge | Non-Recurring Charge |
|--------------|----------------|----------------------|
| 2 - 50 | \$5.00 | \$100.00 |
| 51 - 100 | \$5.00 | \$200.00 |
| 101 - 200 | \$5.00 | \$275.00 |

Conversion fees are waived when switching from a grand-fathered Centrex service.

G. Change Charges

| <u>Change Type</u> | Non-Recurring Charge |
|--|----------------------|
| Centrex Mate Password Change Charges | \$10.00 |
| Station/Line Feature Change Charges, Per Line, Per Occasion - Change NCOS/CAT/LCC on Centrex Line Level - Add or change one or more features - Add or change hunting per UCD station or multiline hunt group - Change Line Class Code (Class of Service) - Add or change pickup groups - Add or change numbers in a call pickup group - Move a station line from one group to another - Add or change a restriction level | \$42.00 |
| System Feature Change Charges - Add or change deluxe time of day routing schedules - Add additional pattern groups for time of day routing, per arrangement | \$128.00 |
| All Other System Feature Changes - Change a system feature - Add or change a Centrex access code - Add, change, or rearrange Centrex access treatment (CAT/NCOS) codes - Add or change network speed calling list or access code, per list - Expand station number range of lines - Change a recorded announcement | \$64.00 |

- Change queuing arrangements

4.14.2 Centrex Smart Pack

Effective February 20, 2003, Centrex Smart Pack service is only available to current subscribers. Alternatively, current and new customers may subscribe to Smart Pack Centrex under the rates, terms and conditions of Section 15.6 of this tariff.

B. <u>Centrex Smart Pack Miscellaneous Charges</u>

2. <u>CSR/Conversion Charges</u>

| Term Commitment | <u>Charge Per CSR</u> |
|-----------------|-----------------------|
| Month-to-Month | \$20.00 |
| 3 Years | \$10.00 |
| 5 Years | \$5.00 * |

3. <u>Installation Charges</u>

| <u>Term Commitment</u> | Charge Per Line |
|------------------------|-----------------|
| Month-to-Month | \$4.00 |
| 3 Years | \$4.00 |
| 5 Years | \$4.00 |

4. <u>Order Activity Charges</u>

| Order Activity | Non-Recurring Charge |
|--|----------------------|
| Service Order Charge New, per location, per occasion (Applies for receiving, recording, and processing information necessary to execute a Customer's request for service.) | \$40.00 |
| Service Order Charge Change, per location, per occasion (Applies to changes to existing Centrex lines/systems, in addition to any other applicable charges.) | \$30.00 |
| Central Office Connection Charge New or Change, per station terminated | \$1.75 |
| Line Connection Charge New or Change, per station terminated | \$3.35 |

* Waived

5. Service Establishment Charge

| Number of Lines | <u>Charge Per Line</u> | | |
|-----------------|------------------------|--|--|
| 2-50 | \$275.00 | | |
| 51-100 | \$275.00 | | |
| 101-200 | \$275.00 | | |
| 201-500 | \$275.00 | | |
| 501 + | \$275.00 | | |

6. <u>Service Conversion Charges</u>

| Conversion Type | Applicable Charges |
|---|--|
| Conversion configured exactly the same as the Customer's Centrex Service with its prior carrier | Service Order ChargeSystem Conversion Charge waiving the NRC |
| Conversion, plus Customer's addition (via Centrex Mate) of new or additional features | - Service Order Charge - System Conversion Charge |
| Conversion, plus addition of new or additional features as programmed by CoreComm | Service Order Charge System Conversion Charge Program Change Charge (individual case basis), per line to be programmed |

System Conversion Charges

| <u>Lines</u> | Monthly Charge | Non-Recurring Charge |
|--------------|----------------|----------------------|
| 2 - 50 | \$5.00 | \$100.00 |
| 51 - 100 | \$5.00 | \$200.00 |
| 101 - 200 | \$5.00 | \$275.00 |

7. <u>Change Charges</u>

| Change Type | Non-Recurring Charge |
|---|----------------------|
| Centrex Mate Password Change Charges | \$10.00 |
| Station/Line Feature Change Charges, Per Line, Per Occasion - Change NCOS/CAT/LCC on Centrex Line Level - Add or change one or more features - Add or change hunting per UCD station or multiline hunt group - Change Line Class Code (Class of Service) - Add or change pickup groups - Add or change numbers in a call pickup group - Move a station line from one group to another - Add or change a restriction level | \$42.00 |
| System Feature Change Charges - Add or change deluxe time of day routing schedules - Add additional pattern groups for time of day routing, per arrangement | \$128.00 |
| All Other System Feature Changes - Change a system feature - Add or change a Centrex access code - Add, change, or rearrange Centrex access treatment (CAT/NCOS) codes - Add or change network speed calling list or access code, per list - Expand station number range of lines - Change a recorded announcement | \$64.00 |

- Change queuing arrangements

C. Centrex Smart Pack Centrex Services

2. <u>Centrex Smart Pack Centrex Services -- Line and Usage Charges</u>

The following charges apply to Centrex Smart Pack Customers in lieu of those line charges provided for in Section 4.14.1.2.D of this tariff. All such business line charges indicated below are comprised of the following charges: network access line, PBX central office termination, Centrex intercommunication, and standard feature rates.

| Term Commitment | Monthly Charge, Per Line |
|-----------------|--------------------------|
| Month-to-Month | Zone |
| | 1 - \$124.95 |
| | 2 - \$124.95 |
| | 3 – \$124.95 |
| 3 Years | Zone |
| | 1 - \$124.95 |
| | 2 - \$124.95 |
| | 3 – \$124.95 |
| 5 Years | \$124.95 |

Additionally, all accounts shall be assessed the following monthly system charge:

Monthly System Charge

Per Account \$5.00

Charges provided for in Section 4.14.1.2.E. of this tariff apply to each line in addition to the monthly rates provided above.

93

GRANDFATHERED SERVICES FOR PREVIOUS CUSTOMERS OF CORECOMM NEWCO, INC.

b. <u>Station Numbers</u>

In lieu of the Station Number Charges provided for under Section 4.14.1.2.F of this tariff, the following Station Number Charges shall be assessed to Centrex Smart Pack Customers:

| | Monthly Charge | Non-Recurring Charge |
|--|------------------|----------------------|
| Station Numbers (block of 20) | | |
| Month to Month Term | \$3.25 | \$174.50 |
| - 3 or 5 Year Term | \$3.00 | \$174.50 |
| Add'l Groups of 20 Numbers (same order) - Month to Month Term - 3 or 5 Year Term | \$3.25 \$3.00 | \$54.00 \$54.00 |
| Individual Station Numbers | \$0.23 | N/A |

c. Local Usage Charges

In lieu of the Local Usage Charges provided for under Section 4.14.1.3 of this tariff, the following Local Usage Charges shall be assessed to Centrex Smart Pack Customers.

1) Message Rate Service

| Term Commitment | Per Message | | |
|-----------------|-------------|--|--|
| Month-to-Month | \$0.0800 | | |
| 3 Years | \$0.0725 | | |
| 5 Years | \$0.0700 | | |

3. Smart Pack Local Services -- Service Enhancement Features

Centrex station feature rates are detariffed and apply to all lines in the Centrex system.

D. <u>Centrex Smart Pack Long Distance Service</u>

2. <u>Switched Services Rates</u>

| Term Commitment | Initial 6 Seconds | Each Additional 6 Seconds |
|------------------------|------------------------|---------------------------|
| Month-to-Month 3 Years | \$0.01000 \$0.00890 | \$0.01000 \$0.00890 |
| 5 Years | \$0.00875 | \$0.00875 |

3. <u>Dedicated Services Rates</u>

| Initial 6 Seconds | Each Additional 6 Seconds |
|-------------------|---------------------------|
| \$0.00690 | \$0.00690 |
| \$0.00580 | \$0.00580 |
| \$0.00560 | \$0.00560 |
| | \$0.00690 \$0.00580 |

4.15.1 Message Telecommunications Service

Message Telecommunications Service (MTS) is a communications service which is available for use by customers who are not presubscribed to CoreComm long distance service, as described at Section 14.0 of this tariff, or who have not selected any carrier as their primary interexchange carrier. MTS enables a user of an exchange access line to place calls to any station on the public switched telecommunications network bearing an NPA-NXX designation associated with points outside the Customer's Local Calling Area, but within the boundaries of the LATA where the Customer (or the Customer's end user) is located. Customers may arrange for interLATA service from the interxchange carriers of their choice. Customers who choose the Company as their carrier for intraLATA calls and interLATA calls are provided such services pursuant to Section 14.0 of this tariff.

MTS Calls are billed in one (1) minute increments with an initial billing period of one (1) minute. The total number of minutes will be rounded at the end of the Billing Period to the nearest minute. Fractional cents will be rounded to the nearest cent.

4.15.2 Message Telecommunications Service -- Rate Schedule

The following charges apply to Message Toll Service:

| | Residence | <u>Business</u> |
|-----------------------|-----------|-----------------|
| Flat rate, per minute | \$.20 | \$.20 |

4.16 SERVICE TRIALS AND SPECIAL PROMOTIONS

The Company may make special promotional offerings of its service on a limited basis. This promotional offering may include waiving or reducing the applicable charges for the promoted service. The promotion may also be offered for a limited duration and limited to specific locations within the state. The waiver of any charge, other than a non-recurring charge, shall be limited to ninety (90) days on a per-customer basis.

4.17 INDIVIDUAL CASE BASIS ARRANGEMENTS

When the Company furnishes a facility or service for which a rate or charge is not specified in the Company's tariffs, charges will be determined on an Individual Case Basis (ICB). ICB rates will be offered to the customer in writing, on a nondiscriminatory basis and will be filed with the PUCO for approval.

4.18 <u>RESIDENTIAL CUSTOMER REFERRAL CREDIT</u>

Residential local exchange service customers will be entitled to a referral credit of \$8.00 on their monthly bills for each new customer referred and subscribed to Company's residential local exchange service. Unused referral credits in any given month may be transferred to succeeding months.

III. PRICES FOR LONG DISTANCE SERVICE CUSTOMERS PRESUBSCRIBED TO CORECOMM SERVICE

14.2.1.1 Standard Service

<u>Per Minute Usage Rates - Day/Evening/Night-Weekend Customers and Casual</u> Callers

Intrastate Rates

| | D | AY | EVE | NING | NIGHT/V | VEEKEND |
|-----------|---------|--------|---------|--------|---------|---------|
| Mileage | Initial | Add'l. | Initial | Add'l. | Initial | Add'l. |
| Band | Minute | Minute | Minute | Minute | Minute | Minute |
| 0 to 10 | 0.1800 | 0.1200 | 0.1275 | 0.0850 | 0.0950 | 0.0540 |
| 11 to 22 | 0.1900 | 0.1600 | 0.1400 | 0.1100 | 0.1100 | 0.0780 |
| 23 to 55 | 0.2200 | 0.2000 | 0.1440 | 0.1300 | 0.1200 | 0.1020 |
| 56 to 124 | 0.2400 | 0.2300 | 0.1570 | 0.1510 | 0.1400 | 0.1200 |
| 125+ | 0.2500 | 0.2500 | 0.1670 | 0.1650 | 0.1400 | 0.1400 |

Effective August 21, 1998, Company no longer will accept new subscriptions to the Standard Service Option to its Landline Service.

14.2.1.2 Option No. I Per Minute Usage Rates - Customers

Customers will be charged a sliding flat, per minute rate determined by the accumulated minutes of use per customer account or the length of time Customer is a customer of Carrier. The more minutes of use that are accumulated, or the more time that passes in which Customer is a customer of Carrier, the rate will drop from one tier to the next. Customers whose minimum usage is at least 250 minutes on at least one line per account may be started at the third tier of the rate schedule.

Effective August 21, 1998, Company no longer will accept new subscriptions to Option No. 1 to its Landline Service.

| Minutes of Use | Length of Time as a | Per Minute Flat Rate |
|----------------|---------------------|----------------------|
| | Customer | |
| 0 to 250 | | \$0.1500 |
| 251 to 500 | 6 Months | \$0.1450 |
| 501 to 1,000 | 1 Year | \$0.1400 |
| 1,001 to 2,500 | 1.5 Years | \$0.1350 |
| 2,501 to 3,750 | 2 Years | \$0.1325 |
| 3,751 to 5,000 | 2.5 Years | \$0.1300 |
| 5,001 to 6,250 | 3 Years | \$0.1275 |
| 6,251 to 7,500 | 3.5 Years | \$0.1250 |
| 7,501 to 8,750 | 4 Years | \$0.1225 |
| Over 8,750 | 4.5 Years | \$0.1200 |

14.2.1.3 Option No. 2 Flat Rate Plans - Residential and Business Customers

| | <u>Initial Minute</u> | Each Additional 6 Seconds | Monthly Service Charge (per line) |
|--------------------------------|-----------------------|---------------------------|--------------------------------------|
| Primary Plan Practical Plan | \$0.15 \$0.12 | \$0.015 \$0.012 | N/A \$5.00 |
| | Initial 6 Seconds | Each Additional 6 Seconds | Monthly Service Charge (per line) |
| Premium Plan | \$0.0105 | \$0.0105 | \$25.00 |

Premium Plan Discount

When the Land Line Premium Plan customer's account reaches \$250 for total Land Line Long Distance billed revenues and recurring charges (inclusive of the \$25 monthly per line service charge(s)) during any billing cycle, then all \$25 per line charges, or applicable prorated amount, associated with that account will be waived. The waiver will take the form of a discount noted as a waiver of recurring charges to be applied to the individual service identification recurring charge only up to the total recurring charges billed. In all cases the \$25 service charge(s) will appear and, when qualifying, the account will also have a corresponding waiver(s). The Premium Plan Discount is available to Premium Plan Customers effective with the billing cycle commencing after September 29, 1997.

Effective August 21, 1998, Company no longer will accept new subscriptions to Option No. 2 to its Landline Service.

14.2.1.4 800, 877 Or 888 Service

| | <u>Initial</u> <u>Minute</u> | Each Additional 6 Seconds | Monthly Service Charge (Per Number) |
|--------------------------------|---------------------------------|---------------------------|---|
| Primary Plan Practical Plan | \$0.19 \$0.17 | \$0.019 \$0.017 | N/A \$5.00 |
| | <u>Initial</u> 6 Seconds | Each Additional 6 Seconds | Monthly Service Charge (Per Number) |
| Premium Plan | \$0.015 | \$0.015 | \$25.00 |

Premium Plan Discount

The monthly service charge for the Premium Plan is subject to waiver when the total revenue for this product in the customer's account reaches \$250 (inclusive of the \$25 monthly per number service charge(s)) during any billing cycle. The waiver will take the form of a discount which will be noted as a waiver of recurring charges to be applied to the individual service identification recurring charge only up to the total recurring charges billed. In all cases the \$25 monthly per number service charge(s) will appear and, when qualifying, the account will also have a corresponding waiver(s). The Premium Plan Discount is available to Premium Plan Customers effective with the billing cycle commencing after September 29, 1997.

Effective August 21, 1998, Company no longer will accept new subscriptions to the above 800. 877 or 888 services.

Payphone Usage Charge

\$1.25 per call

14.2.1.5 700 IntraLATA Service

<u>Per Minute Usage Rates</u> - Land Line Customers

Customers will be charged the rate as set forth on pages two and three of this Price List under the heading CoreComm Land Line Service.

14.2.1.6 Presubscribed Long Distance Customer Plans

A. Residential

| | <u>Initial</u> <u>Minute</u> | Each Additional 6 | Monthly Service Charge |
|--|---------------------------------|----------------------|---------------------------|
| | | Seconds | (per number) |
| Option A | \$0.125 | \$0.0125 | N/A |
| Option B | \$0.115 | \$0.00115 | N/A |
| Option C | \$0.115 | \$0.0115 | N/A |
| Option D | \$0.11 | \$0.011 | N/A |
| Option E | \$0.105 | \$0.0105 | N/A |
| Option F | \$0.10 | \$0.010 | N/A |
| B. Residential 800. 877 and 888 Service | | | |
| Option A | \$0.15 | \$0.015 | N/A |
| Option B | \$0.13 | \$0.013 | \$4.00 |

Option C

Option D

Option E

Option F

\$4.00

\$4.00

\$4.00

\$4.00

GRANDFATHERED SERVICES FOR PREVIOUS CUSTOMERS OF CORECOMM NEWCO, INC.

Non Posidential

| C. <u>Non-Residential</u> | | | |
|---|---------------------------------|---------------------------------|-------------------------------------|
| | <u>Initial</u> <u>Minute</u> | Each Additional 6 Seconds | Monthly Service Charge (per number) |
| Option A | \$0.12 | \$0.012 | N/A |
| Option B | \$0.1025 | \$0.001025 | N/A |
| Option C | \$0.10 | \$0.010 | N/A |
| Option D | \$0.0975 | \$0.00975 | N/A |
| Option E | \$0.095 | \$0.0095 | N/A |
| Option F | \$0.0925 | \$0.00925 | N/A |
| D. Non-Residential 800, 877 or 888 Service | | | |
| | <u>Initial</u> <u>Minute</u> | Each Additional 6 Seconds | Monthly Service Charge (per number) |
| Option A | \$0.12 | \$0.012 | \$4.00 |
| Option B | \$0.1025 | \$0.001025 | \$4.00 |

Effective June 1, 2021 104

\$0.10

\$0.0975

\$0.095

\$0.0925

\$0.010

\$0.00975

\$0.0095

\$0.00925

14.2.1.7 Local Customer Plans

A. Residential

| | Initial Minute | Each Additional 6 Seconds | Monthly Service Charge (per line) |
|-----------|----------------|---------------------------|-----------------------------------|
| Option A* | \$0.12 | \$0.012 | N/A |
| Option B | \$0.11 | \$0.011 | N/A |
| Option C | \$0.11 | \$0.011 | N/A |
| Option D | \$0.105 | \$0.0105 | N/A |
| Option E | \$0.10 | \$0.010 | N/A |
| Option F | \$0.095 | \$0.0095 | N/A |

^{*} Existing customers of this optional service package as of August 21, 1998, will continue to be charged the existing rate of \$0.11 5 per minute until January 31, 1999. Effective that date, the \$0.11 5 per minute rate will sunset and all Option A customers will be charged the price list rate.

B. Residential 800, 877 and 888 Service

| | Initial Minute | Each Additional | Monthly Service |
|----------|-----------------------|-----------------|-----------------|
| | | 6 Seconds | Charge |
| | | | (per line) |
| Option A | \$0.15 | \$0.015 | N/A |
| Option B | \$0.13 | \$0.013 | \$4.00 |

C. Non-Residential

| | <u>Initial</u> <u>Minute</u> | Each Additional 6 Seconds | Monthly Service Charge (per number) |
|---|---------------------------------|---------------------------|-------------------------------------|
| Option A | \$0.105 | \$0.0105 | N/A |
| Option B | \$0.0975 | \$0.00975 | N/A |
| Option C | \$0.095 | \$0.0095 | N/A |
| Option D | \$0.0925 | \$0.00925 | N/A |
| Option E | \$0.09 | \$0.009 | N/A |
| Option F | \$0.0875 | \$0.00875 | N/A |
| D. Non-Residential 800, 877 Or 888 Service | | | |

8// Or 888 Service

| | <u>Initial</u> <u>Minute</u> | <u>Each</u> <u>Additional 6</u> <u>Seconds</u> | Monthly Service Charge (per number) |
|----------|---------------------------------|--|-------------------------------------|
| Option A | \$0.105 | \$0.0105 | \$4.00 |
| Option B | \$0.0975 | \$0.00975 | \$4.00 |
| Option C | \$0.095 | \$0.0095 | \$4.00 |
| Option D | \$0.0925 | \$0.00925 | \$4.00 |

14.2.1.8 Business Customer Term and Volume Plans

B. <u>Usage Rate Schedule</u>

Effective April 13, 2001, the Company will not accept new subscriptions to Business Customer Term and Volume Plans (a) on a month-to-month basis, or (b) from applicants whose usage would be less than 500 minutes per month. Customers subscribing to Business Customer Term and Volume Plans at such service levels prior to April 13, 2001, may maintain such subscriptions as provided herein.

1. Switched Inbound IntraLATA and InterLATA Service

| Term Commitment | | Monthly Minutes of Use | | | |
|--|---|--|--|---|--|
| _ | 0 - 499 M | inutes of Use | <u>500 - 2499 :</u> | Minutes of Use | |
| | Initial 6 Seconds | Each Add'l 6 Seconds | Initial 6 Seconds | Each Add'l 6 Seconds | |
| Month to Month | \$0.01050 | \$0.01050 | \$0.01000 | \$0.01000 | |
| 1 Year | \$0.01050 | \$0.01050 | \$0.00890 | \$0.00890 | |
| 2 Years | \$0.01050 | \$0.01050 | \$0.00875 | \$0.00875 | |
| 3 Years | \$0.01050 | \$0.01050 | \$0.00860 | \$0.00860 | |
| | | | | | |
| Term Commitment | | Monthly Min | nutes of Use | | |
| _ | 2,500 - 4,999 | Minutes of Use | <u>5,000 - 7,499</u> | Minutes of Use | |
| | Initial 6 Seconds | Each Add'l 6 Seconds | Initial 6 Seconds | Each Add'l 6 Seconds | |
| Month to Month | \$0.01000 | \$0.01000 | \$0.01000 | \$0.01000 | |
| 1 Year | \$0.00875 | \$0.00875 | \$0.00860 | \$0.00860 | |
| 2 Years | \$0.00860 | \$0.00860 | \$0.00845 | \$0.00845 | |
| 3 Years | \$0.00845 | \$0.00845 | \$0.00830 | \$0.00830 | |
| | | | | | |
| Term Commitment | | Monthly Mir | nutes of Use | | |
| | 7,500 - 9,999 | | | 99 Minutes of Use | |
| | Initial 6 Seconds | | Initial 6 Seconds | Each Add'l 6 Seconds | |
| Month to Month | \$0.01000 | \$0.01000 | \$0.01000 | \$0.01000 | |
| 1 Year | \$0.00845 | \$0.00845 | \$0.00830 | \$0.00830 | |
| 2 Years | \$0.00830 | \$0.00830 | \$0.00815 | \$0.00815 | |
| 3 Years | \$0.00815 | \$0.00815 | \$0.00800 | \$0.00800 | |
| 1 Year 2 Years 3 Years Term Commitment Month to Month 1 Year 2 Years | \$0.01000 \$0.00875 \$0.00860 \$0.00845 7,500 - 9,999 Initial 6 Seconds \$0.01000 \$0.00845 \$0.00830 | \$0.01000 \$0.00875 \$0.00860 \$0.00845 Monthly Min Minutes of Use Each Add'l 6 Seconds \$0.01000 \$0.00845 \$0.00830 | \$0.01000 \$0.00860 \$0.00845 \$0.00830 hutes of Use 10,000 - 24,99 Initial 6 Seconds \$0.01000 \$0.00830 \$0.00815 | \$0.01000 \$0.00860 \$0.00845 \$0.00830 99 Minutes of Use Each Add'l 6 Second \$0.01000 \$0.00830 \$0.00815 | |

1. <u>Switched Inbound IntraLATA and InterLATA Service</u> (continued)

| Term Commitment | Monthly Minutes of Use | | | |
|-----------------|---|----------------------|------------------------|----------------------|
| _ | 25,000 - 49,999 Minutes of Use 50,000 - 74,999 Minutes of Use | | | |
| | Initial 6 Seconds | Each Add'l 6 Seconds | Initial 6 Seconds | Each Add'l 6 Seconds |
| Month to Month | \$0.01000 | \$0.01000 | \$0.01000 | \$0.01000 |
| 1 Year | \$0.00815 | \$0.00815 | \$0.00800 | \$0.00800 |
| 2 Years | \$0.00800 | \$0.00800 | \$0.00785 | \$0.00785 |
| 3 Years | \$0.00785 | \$0.00785 | \$0.00770 | \$0.00770 |
| | | | | |
| | | | | |
| Term Commitment | | Monthly Mi | | |
| | <u>75,000 - 99,99</u> | 9 Minutes of Use | <u>100,000 - 149,9</u> | 999 Minutes of Use |
| | Initial 6 Seconds | Each Add'l 6 Seconds | Initial 6 Seconds | Each Add'l 6 Seconds |
| Month to Month | \$0.01000 | \$0.01000 | \$0.01000 | \$0.01000 |
| 1 Year | \$0.00785 | \$0.00785 | \$0.00770 | \$0.00770 |
| 2 Years | \$0.00770 | \$0.00770 | \$0.00755 | \$0.00755 |
| 3 Years | \$0.00755 | \$0.00755 | \$0.00740 | \$0.00740 |
| | | | | |
| | | | | |
| Term Commitment | | Monthly Mi | | |
| | <u> 150,000 - 199,9</u> | 99 Minutes of Use | 200,000 Minute | es of Use and above |
| | Initial 6 Seconds | Each Add'l 6 Seconds | Initial 6 Seconds | Each Add'l 6 Seconds |
| Month to Month | \$0.01000 | \$0.01000 | \$0.01000 | \$0.01000 |
| 1 Year | \$0.00755 | \$0.00755 | \$0.00740 | \$0.00740 |
| 2 Years | \$0.00740 | \$0.00740 | \$0.00725 | \$0.00725 |
| 3 Years | \$0.00725 | \$0.00725 | \$0.00705 | \$0.00705 |

2. Switched IntraLATA and InterLATA Outbound

| Term Commitment | Monthly Minutes of Use | | | | |
|-----------------|------------------------|----------------------|----------------------|----------------------------------|--|
| | <u>0 - 499 M</u> | inutes of Use | <u>500 - 2499</u> | <u>500 - 2499 Minutes of Use</u> | |
| | Initial 6 Seconds | Each Add'l 6 Seconds | Initial 6 Seconds | Each Add'l 6 Seconds | |
| Month to Month | \$0.01050 | \$0.01050 | \$0.01000 | \$0.01000 | |
| 1 Year | \$0.01050 | \$0.01050 | \$0.00890 | \$0.00890 | |
| 2 Years | \$0.01050 | \$0.01050 | \$0.00875 | \$0.00875 | |
| 3 Years | \$0.01050 | \$0.01050 | \$0.00860 | \$0.00860 | |
| | | | | | |
| Term Commitment | | Monthly Mi | nutes of Use | | |
| _ | <u> 2,500 - 4,999</u> | Minutes of Use | <u>5,000 - 7,499</u> | Minutes of Use | |
| | Initial 6 Seconds | Each Add'l 6 Seconds | Initial 6 Seconds | Each Add'l 6 Seconds | |
| Month to Month | \$0.01000 | \$0.01000 | \$0.01000 | \$0.01000 | |
| 1 Year | \$0.00875 | \$0.00875 | \$0.00860 | \$0.00860 | |
| 2 Years | \$0.00860 | \$0.00860 | \$0.00845 | \$0.00845 | |
| 3 Years | \$0.00845 | \$0.00845 | \$0.00830 | \$0.00830 | |
| | | | | | |
| Term Commitment | | Monthly Mi | nutes of Use | | |
| _ | <u> 7,500 - 9,999</u> | Minutes of Use | 10,000 - 24,99 | 99 Minutes of Use | |
| | Initial 6 Seconds | Each Add'l 6 Seconds | Initial 6 Seconds | Each Add'l 6 Seconds | |
| Month to Month | \$0.01000 | \$0.01000 | \$0.01000 | \$0.01000 | |
| 1 Year | \$0.00845 | \$0.00845 | \$0.00830 | \$0.00830 | |
| 2 Years | \$0.00830 | \$0.00830 | \$0.00815 | \$0.00815 | |
| 3 Years | \$0.00815 | \$0.00815 | \$0.00800 | \$0.00800 | |

2. <u>Switched IntraLATA and InterLATA Outbound</u> (continued)

| Term Commitment | Monthly Minutes of Use | | | | |
|-----------------|--------------------------------|----------------------|-----------------------|--------------------------------|--|
| _ | 25,000 - 49,999 Minutes of Use | | <u>50,000 - 74,99</u> | 50,000 - 74,999 Minutes of Use | |
| | Initial 6 Seconds | Each Add'l 6 Seconds | Initial 6 Seconds | Each Add'l 6 Seconds | |
| Month to Month | \$0.01000 | \$0.01000 | \$0.01000 | \$0.01000 | |
| 1 Year | \$0.00815 | \$0.00815 | \$0.00800 | \$0.00800 | |
| 2 Years | \$0.00800 | \$0.00800 | \$0.00785 | \$0.00785 | |
| 3 Years | \$0.00785 | \$0.00785 | \$0.00770 | \$0.00770 | |
| | | | | | |
| Term Commitment | | Monthly Mi | nutes of Use | | |
| | 75,000 - 99,99 | 9 Minutes of Use | 100,000 - 149,9 | 999 Minutes of Use | |
| | Initial 6 Seconds | Each Add'l 6 Seconds | Initial 6 Seconds | Each Add'l 6 Seconds | |
| Month to Month | \$0.01000 | \$0.01000 | \$0.01000 | \$0.01000 | |
| 1 Year | \$0.00785 | \$0.00785 | \$0.00770 | \$0.00770 | |
| 2 Years | \$0.00770 | \$0.00770 | \$0.00755 | \$0.00755 | |
| 3 Years | \$0.00755 | \$0.00755 | \$0.00740 | \$0.00740 | |
| | | | | | |
| Term Commitment | | Monthly Mi | nutes of Use | | |
| | 150,000 - 199,9 | 99 Minutes of Use | 200,000 Minute | es of Use and above | |
| | Initial 6 Seconds | Each Add'l 6 Seconds | Initial 6 Seconds | Each Add'l 6 Seconds | |
| Month to Month | \$0.01000 | \$0.01000 | \$0.01000 | \$0.01000 | |
| 1 Year | \$0.00755 | \$0.00755 | \$0.00740 | \$0.00740 | |
| 2 Years | \$0.00740 | \$0.00740 | \$0.00725 | \$0.00725 | |
| 3 Years | \$0.00725 | \$0.00725 | \$0.00705 | \$0.00705 | |

3. Dedicated IntraLATA and InterLATA Inbound

| Term Commitment | Monthly Minutes of Use | | | |
|-----------------|------------------------|----------------------|--------------------------------|----------------------|
| _ | <u>0 - 499 M</u> | inutes of Use | 500 - 2499 Minutes of Use | |
| | Initial 6 Seconds | Each Add'l 6 Seconds | Initial 6 Seconds | Each Add'l 6 Seconds |
| Month to Month | \$0.00700 | \$0.00700 | \$0.00690 | \$0.00690 |
| 1 Year | \$0.00700 | \$0.00700 | \$0.00790 | \$0.00790 |
| 2 Years | \$0.00700 | \$0.00700 | \$0.00775 | \$0.00775 |
| 3 Years | \$0.00700 | \$0.00700 | \$0.00760 | \$0.00760 |
| | | | | |
| Term Commitment | | Monthly Mi | nutes of Use | |
| | <u> 2,500 - 4,999</u> | Minutes of Use | <u>5,000 - 7,499</u> | Minutes of Use |
| | Initial 6 Seconds | Each Add'l 6 Seconds | Initial 6 Seconds | Each Add'l 6 Seconds |
| Month to Month | \$0.00690 | \$0.00690 | \$0.00690 | \$0.00690 |
| 1 Year | \$0.00775 | \$0.00775 | \$0.00760 | \$0.00760 |
| 2 Years | \$0.00760 | \$0.00760 | \$0.00745 | \$0.00745 |
| 3 Years | \$0.00745 | \$0.00745 | \$0.00730 | \$0.00730 |
| | | | | |
| Term Commitment | | Monthly Mi | nutes of Use | |
| | 7,500 - 9,999 | Minutes of Use | 10,000 - 24,999 Minutes of Use | |
| | Initial 6 Seconds | Each Add'l 6 Seconds | Initial 6 Seconds | Each Add'l 6 Seconds |
| Month to Month | \$0.00690 | \$0.00690 | \$0.00690 | \$0.00690 |
| 1 Year | \$0.00745 | \$0.00745 | \$0.00730 | \$0.00730 |
| 2 Years | \$0.00730 | \$0.00730 | \$0.00715 | \$0.00715 |
| 3 Years | \$0.00715 | \$0.00715 | \$0.00700 | \$0.00700 |

3. <u>Dedicated IntraLATA and InterLATA Inbound</u> (continued)

| Term Commitment | Monthly Minutes of Use | | | |
|-----------------|-------------------------|----------------------|------------------------|----------------------|
| _ | <u> 25,000 - 49,99</u> | 9 Minutes of Use | <u>50,000 - 74,99</u> | 99 Minutes of Use |
| | Initial 6 Seconds | Each Add'l 6 Seconds | Initial 6 Seconds | Each Add'l 6 Seconds |
| Month to Month | \$0.00690 | \$0.00690 | \$0.00690 | \$0.00690 |
| 1 Year | \$0.00715 | \$0.00715 | \$0.00700 | \$0.00700 |
| 2 Years | \$0.00700 | \$0.00700 | \$0.00685 | \$0.00685 |
| 3 Years | \$0.00685 | \$0.00685 | \$0.00670 | \$0.00670 |
| | | | | |
| Term Commitment | | Monthly Mi | nutes of Use | |
| | <u> 75,000 - 99,99</u> | 9 Minutes of Use | <u>100,000 - 149,9</u> | 999 Minutes of Use |
| | Initial 6 Seconds | Each Add'l 6 Seconds | Initial 6 Seconds | Each Add'l 6 Seconds |
| Month to Month | \$0.00690 | \$0.00690 | \$0.00690 | \$0.00690 |
| 1 Year | \$0.00685 | \$0.00685 | \$0.00670 | \$0.00670 |
| 2 Years | \$0.00670 | \$0.00670 | \$0.00655 | \$0.00655 |
| 3 Years | \$0.00655 | \$0.00655 | \$0.00640 | \$0.00640 |
| | | | | |
| Term Commitment | | Monthly Mi | nutes of Use | |
| | <u> 150,000 - 199,9</u> | 99 Minutes of Use | 200,000 Minute | es of Use and above |
| | Initial 6 Seconds | Each Add'l 6 Seconds | Initial 6 Seconds | Each Add'l 6 Seconds |
| Month to Month | \$0.00690 | \$0.00690 | \$0.00690 | \$0.00690 |
| 1 Year | \$0.00655 | \$0.00655 | \$0.00640 | \$0.00640 |
| 2 Years | \$0.00640 | \$0.00640 | \$0.00625 | \$0.00625 |
| 3 Years | \$0.00625 | \$0.00625 | \$0.00605 | \$0.00605 |

4. Dedicated IntraLATA and InterLATA Outbound

| Term Commitment | Monthly Minutes of Use | | | |
|-----------------|------------------------|----------------------|--------------------------------|----------------------|
| _ | <u>0 - 499 M</u> | inutes of Use | 500 - 2499 Minutes of Use | |
| | Initial 6 Seconds | Each Add'l 6 Seconds | Initial 6 Seconds | Each Add'l 6 Seconds |
| Month to Month | \$0.00700 | \$0.00700 | \$0.00690 | \$0.00690 |
| 1 Year | \$0.00700 | \$0.00700 | \$0.00790 | \$0.00790 |
| 2 Years | \$0.00700 | \$0.00700 | \$0.00775 | \$0.00775 |
| 3 Years | \$0.00700 | \$0.00700 | \$0.00760 | \$0.00760 |
| | | | | |
| Term Commitment | | Monthly Mi | nutes of Use | |
| _ | <u> 2,500 - 4,999</u> | Minutes of Use | <u>5,000 - 7,499</u> | Minutes of Use |
| | Initial 6 Seconds | Each Add'l 6 Seconds | Initial 6 Seconds | Each Add'l 6 Seconds |
| Month to Month | \$0.00690 | \$0.00690 | \$0.00690 | \$0.00690 |
| 1 Year | \$0.00775 | \$0.00775 | \$0.00760 | \$0.00760 |
| 2 Years | \$0.00760 | \$0.00760 | \$0.00745 | \$0.00745 |
| 3 Years | \$0.00745 | \$0.00745 | \$0.00730 | \$0.00730 |
| | | | | |
| Term Commitment | | Monthly Mi | nutes of Use | |
| | <u> 7,500 - 9,999</u> | Minutes of Use | 10,000 - 24,999 Minutes of Use | |
| | Initial 6 Seconds | Each Add'l 6 Seconds | Initial 6 Seconds | Each Add'l 6 Seconds |
| Month to Month | \$0.00690 | \$0.00690 | \$0.00690 | \$0.00690 |
| 1 Year | \$0.00745 | \$0.00745 | \$0.00730 | \$0.00730 |
| 2 Years | \$0.00730 | \$0.00730 | \$0.00715 | \$0.00715 |
| 3 Years | \$0.00715 | \$0.00715 | \$0.00700 | \$0.00700 |

4. <u>Dedicated IntraLATA and InterLATA Outbound</u> (continued)

| Term Commitment | | Monthly Mi | nutes of Use | |
|-----------------|---------------------------|----------------------|--------------------|---|
| _ | <u> 25,000 - 49,999</u> | Minutes of Use | 50,000 - 74,9 | 99 Minutes of Use |
| | Initial 6 Seconds | Each Add'l 6 Seconds | Initial 6 Seconds | Each Add'l 6 Seconds |
| Month to Month | \$0.00690 | \$0.00690 | \$0.00690 | \$0.00690 |
| 1 Year | \$0.00715 | \$0.00715 | \$0.00700 | \$0.00700 |
| 2 Years | \$0.00700 | \$0.00700 | \$0.00685 | \$0.00685 |
| 3 Years | \$0.00685 | \$0.00685 | \$0.00670 | \$0.00670 |
| | + ********* | 7 ********* | + ******* | 4 |
| Term Commitment | | Monthly Mi | nutes of Use | |
| Term Communicat | 75,000 - 99,999 | · · · | | 999 Minutes of Use |
| | Initial 6 Seconds | Each Add'l 6 Seconds | Initial 6 Seconds | Each Add'l 6 Seconds |
| Month to Month | \$0.00690 | \$0.00690 | \$0.00690 | \$0.00690 |
| 1 Year | \$0.00685 | \$0.00685 | \$0.00670 | \$0.00670 |
| 2 Years | \$0.00670 | \$0.00670 | \$0.00655 | \$0.00675 |
| 3 Years | \$0.00655 | \$0.00655 | \$0.00640 | \$0.00640 |
| 3 Tears | ψ0.00033 | φ0.00033 | Ψ0.00040 | \$0.00040 |
| | | | | |
| Term Commitment | | Monthly Min | nutes of Use | |
| _ | <u> 150,000 - 199,999</u> | 9 Minutes of Use | 200,000 Minute | es of Use and above |
| | Initial 6 Seconds | Each Add'l 6 Seconds | Initial 6 Seconds | Each Add'l 6 Seconds |
| Month to Month | \$0.00690 | \$0.00690 | \$0.00690 | \$0.00690 |
| 1 Year | \$0.00655 | \$0.00655 | \$0.00640 | \$0.00640 |
| 2 Years | \$0.00640 | \$0.00640 | \$0.00625 | \$0.00625 |
| 3 Years | \$0.00625 | \$0.00625 | \$0.00605 | \$0.00605 |
| 5. | Calling Card Se | rvice | | |
| · . | carring cara so | 11100 | | |
| | | Initial 30 Secon | nds Each Add | itional 6 Seconds |
| | Month to Mont | th \$0.125 | | \$0.025 |
| | 1 Year | \$0.125 | | \$0.025 |
| | 2 Years | \$0.125 \$0.125 | | \$0.025 |
| | 3 Years | \$0.125 \$0.125 | | \$0.025 |
| | 3 Tears | Ψ0.123 | | ψ0.023 |
| | | <u>P</u> : | ayphone Use Charge | |
| | Month to Mont | ·h | \$1.25 | |
| | 1 Year | .11 | \$1.25 \$1.25 | |
| | 2 Years | | \$1.25 \$1.25 | |
| | | | | |
| | 3 Years | | \$1.25 | |

C. <u>Miscellaneous Monthly Recurring and Non-Recurring Charges</u>

1. Toll Free Service

Toll Free Service Monthly Recurring Charge

- per toll free number

\$4.00

2. Project Account Codes

| Code | Non-Recurring Set-Up Charge | Monthly Recurring Charge | Change Charge, per change order |
|--------------|-----------------------------|--------------------------|------------------------------------|
| Non-Verified | \$5.00 * | \$5.00 | \$5.00 |
| Verified | \$5.00 * | \$5.00 | \$5.00 |
| Custom | \$50.00 | \$5.00 | \$5.00 |

3. <u>Directory Assistance Charges</u>

Directory Assistance Charges shall apply as provided in Section 14.2.4.1 of this tariff.

* Waived

14.2.1.9 Residential Customer Rate Plans

B. <u>Standard Rate Plan</u>

1. Option 1 – Basic

The minimum call duration is sixty (60) seconds. Usage is measured thereafter in six (6) second increments and rounded to the next higher six (6) second period.

| | Initial 60 Seconds | Each Add'l 6 Seconds |
|-----------|--------------------|----------------------|
| IntraLATA | \$0.10 | \$0.01 |
| InterLATA | \$0.10 | \$0.01 |

2. Option 2 – 30 Minute Premium

Customers subscribing to service pursuant to Section 4.4.7 may additionally subscribe to this option, which includes a combined thirty (30) minutes of intraLATA and interLATA long distance services per month. Minimum call duration is sixty (60) seconds. Usage is measured thereafter in six (6) second increments and rounded to the next higher six (6) second period. Fractional amounts are rounded to the next full cent.

| | <u>Per Month</u> |
|---------|------------------|
| Monthly | \$2.00 |

The following charges apply to usage in excess of the monthly allotment:

| | Initial 60 Seconds | Each Add'l 6 Seconds |
|-----------|--------------------|----------------------|
| IntraLATA | \$0.08 | \$0.008 |
| InterLATA | \$0.08 | \$0.008 |

14.2.1.9 Residential Customer Rate Plans (cont'd)

3. Option 3 – 75 Minute Premium

Customers subscribing to service pursuant to Section 4.4.5.1 may additionally subscribe to this option, which includes a combined seventy-five (75) minutes of intraLATA and interLATA long distance services per month. Minimum call duration is sixty (60) seconds. Usage is measured thereafter in six (6) second increments and rounded to the next higher six (6) second period. Fractional amounts are rounded to the next full cent.

Monthly Per Month \$5.00

The following charges apply to usage in excess of the monthly allotment:

| | Initial 60 Seconds | Each Add'l 6 Seconds |
|-----------|--------------------|----------------------|
| IntraLATA | \$0.07 | \$0.007 |
| InterLATA | \$0.07 | \$0.007 |

14.2.1.9 <u>Residential Customer Rate Plans</u> (cont'd.)

C. Optional Calling Plans

1. 6.9 Cents / No Fee

Charges are based on actual usage of the Company network. Chargeable time begins when a connection is established between the calling station and the called station. Chargeable time ends when either party "hangs up" thereby releasing the network connection. For billing purposes, the minimum call duration is sixty (60) seconds, and usage is measured thereafter in six (6) second increments and rounded to the next highest six (6) second period.

a. Monthly Rate

| | Per Month |
|---------|-----------|
| Monthly | \$0.00 |

b. <u>Direct Dial Rates</u> - Domestic

| | Initial 60 Seconds | Each Add'l 6 Seconds |
|----------------------|--------------------|----------------------|
| Intrastate InterLATA | \$0.0690 | \$0.0069 |
| Intrastate IntraLATA | \$0.0690 | \$0.0069 |

c. Calling Card Service Rates

The rates, terms and conditions for calling card services are contained in Section 14.2.1.9.D in this tariff.

14.2.1.9 <u>Residential Customer Rate Plans</u> (cont'd.)

C. Optional Calling Plans (cont'd.)

2. 5 Cents / \$2.95 Fee

Charges are based on actual usage of the Company network. Chargeable time begins when a connection is established between the calling station and the called station. Chargeable time ends when either party "hangs up" thereby releasing the network connection. For billing purposes, the minimum call duration is sixty (60) seconds, and usage is measured thereafter in six (6) second increments and rounded to the next highest six (6) second period.

a. Monthly Rate

| | Per Month |
|---------|-----------|
| Monthly | \$2.95 |

The customer may receive a \$1.00 discount off of the per-month rate if the customer chooses online credit card billing.

b. Direct Dial Rates - Domestic

| | Initial 60 Seconds | Each Add'l 6 Seconds |
|----------------------|--------------------|----------------------|
| Intrastate InterLATA | \$0.0500 | \$0.0050 |
| Intrastate IntraLATA | \$0.0500 | \$0.0050 |

c. <u>Calling Card Service Rates</u>

The rates, terms and conditions for calling card services are contained in Section 14.2.1.9.D in this tariff.

14.2.1.9 <u>Residential Customer Rate Plans</u> (cont'd.)

C. Optional Calling Plans (cont'd.)

3. <u>CoreComm 200 Plan</u>

Charges are based on actual usage of the Company network. Chargeable time begins when a connection is established between the calling station and the called station. Chargeable time ends when either party "hangs up" thereby releasing the network connection. For billing purposes, the minimum call duration is sixty (60) seconds, and usage is measured thereafter in six (6) second increments and rounded to the next highest six (6) second period.

a. <u>Monthly Rate</u>

Monthly Per Month \$5.95

The customer may receive a \$1.00 discount off of the per-month rate if the customer chooses online credit card billing.

b. Direct Dial Rates - Domestic

CoreComm 200 plan is comprised of a combination of intrastate usage which is regulated by the PUCO and interstate usage which is not regulated by the PUCO. Direct dialed intrastate rates are as follows:

| | Initial 60 Seconds | Each Add'l 6 |
|----------------------|--------------------|--------------|
| | | Seconds |
| Intrastate InterLATA | \$0.0550 | \$0.0055 |
| Intrastate IntraLATA | \$0.0550 | \$0.0055 |

c. Calling Card Service Rates

The rates, terms and conditions for calling card services are contained in Section 14.2.1.9.D in this tariff.

14.2.1.9 Residential Customer Rate Plans (cont'd.)

C. Optional Calling Plans (cont'd.)

4. <u>CoreComm Unlimited</u>

Charges are based on actual usage of the Company network. Chargeable time begins when a connection is established between the calling station and the called station. Chargeable time ends when either party "hangs up" thereby releasing the network connection. For billing purposes, the minimum call duration is sixty (60) seconds, and usage is measured thereafter in six (6) second increments and rounded to the next highest six (6) second period.

a. Monthly Rate

| | Per Month |
|---------|-------------------|
| Monthly | Zone 1 - \$119.95 |
| | Zone 2 – \$119.95 |
| | Zone 3 - \$119.95 |

The customer may receive a \$1.00 discount off of the per-month rate if the customer chooses online credit card billing.

b. Direct Dial Rates - Domestic

CoreComm Unlimited plan is comprised of a combination of intrastate usage which is regulated by the PUCO and interstate usage which is not regulated by the PUCO. All intrastate interLATA usage is included in the monthly rate. Direct dialed intrastate rates are as follows:

| | Initial 60 Seconds | Each Add'l 6 Seconds |
|----------------------|--------------------------------------|--------------------------------------|
| Intrastate InterLATA | Unlimited – included in monthly rate | Unlimited – included in monthly rate |
| Intrastate IntraLATA | \$0.0550 | \$0.0055 |

c. Calling Card Service Rates

The rates, terms and conditions for calling card services are contained in Section 14.2.1.9.D in this tariff.

14.2.1.9 Residential Customer Rate Plans (cont'd.)

D. <u>Calling Card Service Rates</u>

The following rates shall apply for all calling card calls:

| | <u>Initial (</u> | Initial 60 Seconds | | d'l 6 Seconds |
|----------------------|------------------|-----------------------------|---------|-----------------|
| | Peak Period | Peak Period Off-Peak Period | | Off-Peak Period |
| Intrastate InterLATA | \$0.260 | \$0.220 | \$0.026 | \$0.022 |
| Intrastate IntraLATA | \$0.260 | \$0.220 | \$0.026 | \$0.022 |

Additionally, the following per call surcharge shall apply to all calling card calls:

| | Per Call Rate |
|---|---------------|
| Per Calling Card Call | \$0.25 |
| Per Calling Card Call Originating From A Payphone | \$1.25 |

14.2.1.10 Business Customer Rate Plans

B. <u>Standard Rate Plan</u>

The minimum call duration is sixty (60) seconds. Usage is measured thereafter in sixty (60) second increments and rounded to the next higher sixty (60) second period.

| | Initial 60 Seconds | Each Add'l 60 Seconds |
|-----------|--------------------|-----------------------|
| IntraLATA | \$0.075 | \$0.075 |
| InterLATA | \$0.075 | \$0.075 |

14.2.1.11 Optional IntraLATA Message Toll Services

A. The Optional IntraLATA Message Toll Services established in this section are available only on a "grandfathered basis" to Customers who were acquired by the Company from USN Communications, Inc. and who on April 2, 2001 were subscribers to the services as assumed by USN from Ameritech and as continued by the Company since May 26, 1999. No new subscriptions will be accepted to Optional IntraLATA MTS.

B. <u>USN AdvantageLink Rates and Charges</u>

USN AdvantageLink customers will be charged a fixed monthly rate, per account. This rate is in addition to the applicable usage rate. The monthly and usage rates are as set forth in the rate schedule below. A Customer's CoreComm AdvantageLink rate is determined by selecting the USN AdvantageLink rate which corresponds to the original length of the Customer's ValueLink contract assumed by USN Communications, Inc., the Customer's former carrier.

Fixed Rate, per account: \$10.00

Original Length of ValueLink Contract Being AssumedMonth-to-Month18 Months36 Months\$0.19\$0.17\$0.15

C. CoreComm AdvantageLink Plus Rates and Charges

.1 CY7.1 Y 1

Usage Rate, per minute:

| Original Length of ValueLink | | AdvantageLink |
|---|---|---|
| Contract Assumed by USN | <u>MMUC</u> | Plus Rate |
| 18 Months | \$50.00 | \$0.16 |
| 18 Months | \$100.00 | \$0.16 |
| 18 Months | \$250.00 | \$0.16 |
| 18 Months | \$500.00 | \$0.15 |
| 18 Months | \$1,000.00 | \$0.15 |
| 18 Months | \$2,500.00 | \$0.13 |
| | | |
| Original Length of ValueLink | | AdvantageLink |
| Original Length of ValueLink <u>Contract Assumed by USN</u> | <u>MMUC</u> | AdvantageLink <u>Plus Rate</u> |
| 2 | MMUC \$50.00 | <u>e</u> |
| Contract Assumed by USN | | Plus Rate |
| Contract Assumed by USN 36 Months | \$50.00 | Plus Rate \$0.14 |
| Contract Assumed by USN 36 Months 36 Months | \$50.00 \$100.00 | Plus Rate \$0.14 \$0.14 |
| Contract Assumed by USN 36 Months 36 Months 36 Months | \$50.00 \$100.00 \$250.00 | Plus Rate \$0.14 \$0.14 \$0.14 |
| Contract Assumed by USN 36 Months 36 Months 36 Months 36 Months | \$50.00 \$100.00 \$250.00 \$500.00 | Plus Rate \$0.14 \$0.14 \$0.14 \$0.13 |

D. Enhanced USN AdvantageLink Plus Usage Rates

| Original Length of ValueLink | | | |
|------------------------------|--|---|---|
| Contract Assumed by USN | <u>MMUC</u> | Peak Rate | Off Peak Rate |
| Month-to-Month | \$ 25.00 | \$ 0.190 | \$ 0.172 |
| | \$ 50.00 | \$ 0.190 | \$ 0.172 |
| | \$ 100.00 | \$ 0.190 | \$ 0.172 |
| | \$ 250.00 | \$ 0.190 | \$ 0.172 |
| | \$ 500.00 | \$ 0.175 | \$ 0.155 |
| | \$1,000.0 | \$ 0.175 | \$ 0.155 |
| | \$2,500.0 | \$ 0.155 | \$ 0.135 |
| | <u>MAUC</u> | Peak Rate | Off Peak Rate |
| Twelve Months | \$ 300.00 | \$ 0.190 | \$ 0.172 |
| | \$ 600.00 | \$ 0.190 | \$ 0.172 |
| | \$1,200.00 | \$ 0.190 | \$ 0.172 |
| | \$3,000.00 | \$ 0.190 | \$ 0.172 |
| | \$6,000.00 | \$ 0.175 | \$ 0.155 |
| | \$ 12,000.00 | \$ 0.175 | \$ 0.155 |
| | \$ 30,000.00 | \$ 0.155 | \$ 0.135 |
| | MAUC | Peak Rate | Off Peak Rate |
| Twenty-Four Months | \$ 300.00 | \$ 0.165 | \$ 0.150 |
| | Φ <00 00 | ¢ 0.165 | |
| | \$ 600.00 | \$ 0.165 | \$ 0.150 |
| | \$ 600.00 \$1,200.00 | \$ 0.165 | \$ 0.150 \$ 0.150 |
| | | | |
| | \$1,200.00 | \$ 0.165 | \$ 0.150 |
| | \$1,200.00 \$3,000.00 | \$ 0.165 \$ 0.165 | \$ 0.150 \$ 0.150 |
| | \$1,200.00 \$3,000.00 \$6,000.00 | \$ 0.165 \$ 0.165 \$ 0.140 | \$ 0.150 \$ 0.150 \$ 0.125 |
| | \$1,200.00 \$3,000.00 \$6,000.00 \$ 12,000.00 | \$ 0.165 \$ 0.165 \$ 0.140 \$ 0.140 | \$ 0.150 \$ 0.150 \$ 0.125 \$ 0.125 |
| Thirty-Six Months | \$1,200.00 \$3,000.00 \$6,000.00 \$ 12,000.00 \$ 30,000.00 | \$ 0.165 \$ 0.165 \$ 0.140 \$ 0.140 \$ 0.120 | \$ 0.150 \$ 0.150 \$ 0.125 \$ 0.125 \$ 0.105 |
| Thirty-Six Months | \$1,200.00 \$3,000.00 \$6,000.00 \$ 12,000.00 \$ 30,000.00 | \$ 0.165 \$ 0.165 \$ 0.140 \$ 0.140 \$ 0.120 | \$ 0.150 \$ 0.150 \$ 0.125 \$ 0.125 \$ 0.105 |
| Thirty-Six Months | \$1,200.00 \$3,000.00 \$6,000.00 \$ 12,000.00 \$ 30,000.00 MAUC \$ 300.00 | \$ 0.165 \$ 0.165 \$ 0.140 \$ 0.140 \$ 0.120 Peak Rate \$ 0.155 | \$ 0.150 \$ 0.150 \$ 0.125 \$ 0.125 \$ 0.105 Off Peak Rate \$ 0.140 |
| Thirty-Six Months | \$1,200.00 \$3,000.00 \$6,000.00 \$ 12,000.00 \$ 30,000.00 <u>MAUC</u> \$ 300.00 \$ 600.00 | \$ 0.165 \$ 0.165 \$ 0.140 \$ 0.140 \$ 0.120 Peak Rate \$ 0.155 \$ 0.155 | \$ 0.150 \$ 0.150 \$ 0.125 \$ 0.125 \$ 0.105 Off Peak Rate \$ 0.140 \$ 0.140 |
| Thirty-Six Months | \$1,200.00 \$3,000.00 \$6,000.00 \$ 12,000.00 \$ 30,000.00 MAUC \$ 300.00 \$ 600.00 \$1,200.00 | \$ 0.165 \$ 0.165 \$ 0.140 \$ 0.140 \$ 0.120 Peak Rate \$ 0.155 \$ 0.155 \$ 0.155 | \$ 0.150 \$ 0.150 \$ 0.125 \$ 0.125 \$ 0.105 Off Peak Rate \$ 0.140 \$ 0.140 \$ 0.140 |
| Thirty-Six Months | \$1,200.00 \$3,000.00 \$6,000.00 \$ 12,000.00 \$ 30,000.00 MAUC \$ 300.00 \$ 600.00 \$1,200.00 \$3,000.00 | \$ 0.165 \$ 0.165 \$ 0.140 \$ 0.140 \$ 0.120 Peak Rate \$ 0.155 \$ 0.155 \$ 0.155 | \$ 0.150 \$ 0.150 \$ 0.125 \$ 0.125 \$ 0.105 Off Peak Rate \$ 0.140 \$ 0.140 \$ 0.140 \$ 0.140 |

^{*}Peak period is from 8:00 AM to 5:00 PM, Monday through Friday

^{*}Off Peak period is from 5:00 PM to 8:00 AM, Monday through Friday and all day Saturday and Sunday

E. <u>USN AdvantageLink Premier Rates</u>

PBX Systems

| Original Length of ValueLink | | | |
|------------------------------|-------------|-----------|---------------|
| Contract Assumed by USN | <u>MMUC</u> | Peak Rate | Off Peak Rate |
| | | | |
| Month to Month | \$500.00 | \$0.110 | \$0.085 |
| | \$1,000.00 | \$0.110 | \$0.085 |
| | \$1,500.00 | \$0.110 | \$0.085 |
| | \$2,000.00 | \$0.110 | \$0.085 |
| | \$2,500.00 | \$0.110 | \$0.085 |
| | | | |
| | <u>MAUC</u> | | |
| Twelve Months | \$ 6,000.00 | \$0.110 | \$0.085 |
| | \$12,000.00 | \$0.110 | \$0.085 |
| | \$18,000.00 | \$0.110 | \$0.085 |
| | \$24,000.00 | \$0.110 | \$0.085 |
| | \$30,000.00 | \$0.110 | \$0.085 |
| | | | |
| | MAUC | | |
| Twenty-Four Months | \$ 6,000.00 | \$0.095 | \$0.075 |
| | \$12,000.00 | \$0.095 | \$0.075 |
| | \$18,000.00 | \$0.095 | \$0.075 |
| | \$24,000.00 | \$0.095 | \$0.075 |
| | \$30,000.00 | \$0.095 | \$0.075 |
| | | | |
| | <u>MAUC</u> | | |
| Thirty-Six Months | \$ 6,000.00 | \$0.095 | \$0.075 |
| | \$12,000.00 | \$0.095 | \$0.075 |
| | \$18,000.00 | \$0.095 | \$0.075 |
| | \$24,000.00 | \$0.095 | \$0.075 |
| | \$30,000.00 | \$0.095 | \$0.075 |
| | | | |

E. <u>USN AdvantageLink Premier Rates</u> (continued)

Centrex Systems

| Original Length of ValueLink | | | |
|------------------------------|-------------|-----------|---------------|
| Contract Assumed by USN | <u>MMUC</u> | Peak Rate | Off Peak Rate |
| | | | |
| Month to Month | \$500.00 | \$0.105 | \$0.085 |
| | \$1,000.00 | \$0.105 | \$0.085 |
| | \$1,500.00 | \$0.105 | \$0.085 |
| | \$2,000.00 | \$0.105 | \$0.085 |
| | \$2,500.00 | \$0.105 | \$0.085 |
| | | | |
| | MAUC | | |
| Twelve Months | \$ 6,000.00 | \$0.105 | \$0.085 |
| | \$12,000.00 | \$0.105 | \$0.085 |
| | \$18,000.00 | \$0.105 | \$0.085 |
| | \$24,000.00 | \$0.105 | \$0.085 |
| | \$30,000.00 | \$0.105 | \$0.085 |
| | | | |
| | <u>MAUC</u> | | |
| Twenty-Four Months | \$ 6,000.00 | \$0.100 | \$0.080 |
| | \$12,000.00 | \$0.100 | \$0.080 |
| | \$18,000.00 | \$0.100 | \$0.080 |
| | \$24,000.00 | \$0.100 | \$0.080 |
| | \$30,000.00 | \$0.100 | \$0.080 |
| | | | |
| | <u>MAUC</u> | | |
| Thirty-Six Months | \$ 6,000.00 | \$0.100 | \$0.080 |
| | \$12,000.00 | \$0.100 | \$0.080 |
| | \$18,000.00 | \$0.100 | \$0.080 |
| | \$24,000.00 | \$0.100 | \$0.080 |
| | \$30,000.00 | \$0.100 | \$0.080 |
| | | | |

F. <u>USN AdvantageLink Select Rates</u>

| Minimum Annual Revenue Commitment | IntraLATA To | oll and 800/888 Pric | ce Per Minute |
|-----------------------------------|--------------|----------------------|---------------|
| | 1 Year | 2 Year | 3 Year |
| \$ 700.00 to \$ 5,999.00 | \$0.145 | \$0.145 | \$0.145 |
| \$ 6,000.00 + | \$0.125 | \$0.125 | \$0.125 |

14.2.1.12 <u>CoreComm Standalone Long Distance</u>

B. <u>Usage Rate Schedule</u>

1. <u>Switched IntraLATA and InterLATA Long Distance Services</u>

a. Switched Inbound IntraLATA Service

| Term Commitment | Monthly Minutes of Use | | | |
|-----------------|---|----------------------|------------------------------|----------------------|
| | 500 - 2499 N | Minutes of Use | 2,500 - 4,999 Minutes of Use | |
| | Initial 6 Seconds | Each Add'l 6 Seconds | Initial 6 Seconds | Each Add'l 6 Seconds |
| 1 Year | \$0.00790 | \$0.00790 | \$0.00775 | \$0.00775 |
| 2 Years | \$0.00775 | \$0.00775 | \$0.00760 | \$0.00760 |
| 3 Years | \$0.00760 | \$0.00760 | \$0.00745 | \$0.00745 |
| | | | | |
| | | Minutes of Use | | Minutes of Use |
| | Initial 6 Seconds | Each Add'l 6 Seconds | Initial 6 Seconds | Each Add'l 6 Seconds |
| 1 Year | \$0.00760 | \$0.00760 | \$0.00745 | \$0.00745 |
| 2 Years | \$0.00745 | \$0.00745 | \$0.00730 | \$0.00730 |
| 3 Years | \$0.00730 | \$0.00730 | \$0.00715 | \$0.00715 |
| | 10,000 - 24,999 Minutes of Use 25,000 - 49,999 Minutes of Use | | |) Minutes of Use |
| | Initial 6 Seconds | Each Add'l 6 Seconds | Initial 6 Seconds | Each Add'l 6 Seconds |
| 1 Year | \$0.00730 | \$0.00730 | \$0.00715 | \$0.00715 |
| 2 Years | \$0.00715 | \$0.00736 | \$0.00719 | \$0.00713 |
| 3 Years | \$0.00713 | \$0.00713 | \$0.00700 | \$0.00685 |
| 5 Tears | ψ0.00700 | ψ0.00700 | ψ0.00003 | ψ0.0003 |
| | <u>50,000 - 74,99</u> | 9 Minutes of Use | <u>75,000 - 99,999</u> | Minutes of Use |
| | Initial 6 Seconds | Each Add'l 6 Seconds | Initial 6 Seconds | Each Add'l 6 Seconds |
| 1 Year | \$0.00700 | \$0.00700 | \$0.00685 | \$0.00685 |
| 2 Years | \$0.00685 | \$0.00685 | \$0.00670 | \$0.00670 |
| 3 Years | \$0.00670 | \$0.00670 | \$0.00655 | \$0.00655 |
| | 100.000 140.0 | 0036 | 150,000, 100,00 | 20.15 |
| | | 99 Minutes of Use | | 99 Minutes of Use |
| 1.37 | Initial 6 Seconds | Each Add'l 6 Seconds | Initial 6 Seconds | Each Add'l 6 Seconds |
| 1 Year | \$0.00670 | \$0.00670 | \$0.00655 | \$0.00655 |
| 2 Years | \$0.00655 | \$0.00655 | \$0.00640 | \$0.00640 |
| 3 Years | \$0.00640 | \$0.00640 | \$0.00625 | \$0.00625 |
| | 200,000 Minutes | s of Use and above | | |
| | Initial 6 Seconds | Each Add'l 6 Seconds | | |
| 1 Year | \$0.00640 | \$0.00640 | | |
| 2 Years | \$0.00625 | \$0.00625 | | |
| 3 Years | \$0.00605 | \$0.00605 | | |
| | | | | |

b. <u>Switched Inbound InterLATA Service</u>

| Term Commitment | Monthly Minutes of Use | | | |
|-----------------|------------------------------|----------------------|---------------------------------------|----------------------|
| | 500 - 2499 N | Minutes of Use | 2,500 - 4,999 Minutes of Use | |
| | Initial 6 Seconds | Each Add'l 6 Seconds | Initial 6 Seconds | Each Add'l 6 Seconds |
| 1 Year | \$0.00790 | \$0.00790 | \$0.00775 | \$0.00775 |
| 2 Years | \$0.00775 | \$0.00775 | \$0.00760 | \$0.00760 |
| 3 Years | \$0.00760 | \$0.00760 | \$0.00745 | \$0.00745 |
| | 5,000 5,400 | 257 | 7.500 0.000 | N.C. C.T. |
| | | Minutes of Use | · · · · · · · · · · · · · · · · · · · | Minutes of Use |
| 1 37 | Initial 6 Seconds | Each Add'l 6 Seconds | Initial 6 Seconds | Each Add'l 6 Seconds |
| 1 Year | \$0.00760 | \$0.00760 | \$0.00745 | \$0.00745 |
| 2 Years | \$0.00745 | \$0.00745 | \$0.00730 | \$0.00730 |
| 3 Years | \$0.00730 | \$0.00730 | \$0.00715 | \$0.00715 |
| | 10.000 - 24.99 | 9 Minutes of Use | 25,000 - 49,999 | Minutes of Use |
| | Initial 6 Seconds | Each Add'l 6 Seconds | Initial 6 Seconds | Each Add'l 6 Seconds |
| 1 Year | \$0.00730 | \$0.00730 | \$0.00715 | \$0.00715 |
| 2 Years | \$0.00715 | \$0.00715 | \$0.00700 | \$0.00700 |
| 3 Years | \$0.00700 | \$0.00700 | \$0.00685 | \$0.00685 |
| | 7 0 000 7 4 00 | 2.5 | | |
| | | 9 Minutes of Use | | Minutes of Use |
| 4.77 | Initial 6 Seconds | Each Add'l 6 Seconds | Initial 6 Seconds | Each Add'l 6 Seconds |
| 1 Year | \$0.00700 | \$0.00700 | \$0.00685 | \$0.00685 |
| 2 Years | \$0.00685 | \$0.00685 | \$0.00670 | \$0.00670 |
| 3 Years | \$0.00670 | \$0.00670 | \$0.00655 | \$0.00655 |
| | 100.000 - 149.9 | 99 Minutes of Use | 150.000 - 199.99 | 99 Minutes of Use |
| | Initial 6 Seconds | Each Add'l 6 Seconds | Initial 6 Seconds | Each Add'l 6 Seconds |
| 1 Year | \$0.00670 | \$0.00670 | \$0.00655 | \$0.00655 |
| 2 Years | \$0.00655 | \$0.00655 | \$0.00640 | \$0.00640 |
| 3 Years | \$0.00640 | \$0.00640 | \$0.00625 | \$0.00625 |
| | | | | |
| | | s of Use and above | | |
| | Initial 6 Seconds | Each Add'l 6 Seconds | | |
| 1 Year | \$0.00640 | \$0.00640 | | |
| 2 Years | \$0.00625 | \$0.00625 | | |
| 3 Years | \$0.00605 | \$0.00605 | | |

c. <u>Switched Outbound IntraLATA Service</u>

| Term Commitment | Monthly Minutes of Use | | | |
|-----------------|---|----------------------|--------------------------------------|----------------------|
| | 500 - 2499 N | Minutes of Use | 2,500 - 4,999 | Minutes of Use |
| | Initial 6 Seconds | Each Add'l 6 Seconds | Initial 6 Seconds | Each Add'l 6 Seconds |
| 1 Year | \$0.00790 | \$0.00790 | \$0.00775 | \$0.00775 |
| 2 Years | \$0.00775 | \$0.00775 | \$0.00760 | \$0.00760 |
| 3 Years | \$0.00760 | \$0.00760 | \$0.00745 | \$0.00745 |
| | | | | |
| | <u>5,000 - 7,499</u> | Minutes of Use | <u>7,500 - 9,999</u> | Minutes of Use |
| | Initial 6 Seconds | Each Add'l 6 Seconds | Initial 6 Seconds | Each Add'l 6 Seconds |
| 1 Year | \$0.00760 | \$0.00760 | \$0.00745 | \$0.00745 |
| 2 Years | \$0.00745 | \$0.00745 | \$0.00730 | \$0.00730 |
| 3 Years | \$0.00730 | \$0.00730 | \$0.00715 | \$0.00715 |
| | | | | |
| | | 9 Minutes of Use | | Minutes of Use |
| | Initial 6 Seconds | Each Add'l 6 Seconds | <u>Initial 6 Seconds</u> | Each Add'l 6 Seconds |
| 1 Year | \$0.00730 | \$0.00730 | \$0.00715 | \$0.00715 |
| 2 Years | \$0.00715 | \$0.00715 | \$0.00700 | \$0.00700 |
| 3 Years | \$0.00700 | \$0.00700 | \$0.00685 | \$0.00685 |
| | 50,000, 74,000 | 9 Minutes of Use | 75,000,00,000 | Minutes of Use |
| | <u>50,000 - 74,999</u> Initial 6 Seconds | Each Add'l 6 Seconds | 75,000 - 99,999 Initial 6 Seconds | Each Add'l 6 Seconds |
| 1 Year | \$0.00700 | \$0.00700 | \$0.00685 | \$0.00685 |
| 2 Years | \$0.00700 | \$0.00700 | \$0.00670 | \$0.00670 |
| 3 Years | \$0.00670 | \$0.00670 | \$0.00675 | \$0.00655 |
| 3 Tears | \$0.00070 | \$0.00070 | \$0.00033 | \$0.00033 |
| | 100,000 - 149,99 | 99 Minutes of Use | 150,000 - 199,99 | 99 Minutes of Use |
| | Initial 6 Seconds | Each Add'l 6 Seconds | Initial 6 Seconds | Each Add'l 6 Seconds |
| 1 Year | \$0.00670 | \$0.00670 | \$0.00655 | \$0.00655 |
| 2 Years | \$0.00655 | \$0.00655 | \$0.00640 | \$0.00640 |
| 3 Years | \$0.00640 | \$0.00640 | \$0.00625 | \$0.00625 |
| | | | | |
| | | s of Use and above | | |
| | Initial 6 Seconds | Each Add'l 6 Seconds | | |
| 1 Year | \$0.00640 | \$0.00640 | | |
| 2 Years | \$0.00625 | \$0.00625 | | |
| 3 Years | \$0.00605 | \$0.00605 | | |
| | | | | |

d. <u>Switched Outbound InterLATA Service</u>

| Term Commitment | Monthly Minutes of Use | | | |
|-----------------|---|----------------------|------------------------------|----------------------|
| | 500 - 2499 N | Minutes of Use | 2,500 - 4,999 Minutes of Use | |
| | Initial 6 Seconds | Each Add'l 6 Seconds | Initial 6 Seconds | Each Add'l 6 Seconds |
| 1 Year | \$0.00790 | \$0.00790 | \$0.00775 | \$0.00775 |
| 2 Years | \$0.00775 | \$0.00775 | \$0.00760 | \$0.00760 |
| 3 Years | \$0.00760 | \$0.00760 | \$0.00745 | \$0.00745 |
| | | | | |
| | | Minutes of Use | | Minutes of Use |
| | Initial 6 Seconds | Each Add'l 6 Seconds | Initial 6 Seconds | Each Add'l 6 Seconds |
| 1 Year | \$0.00760 | \$0.00760 | \$0.00745 | \$0.00745 |
| 2 Years | \$0.00745 | \$0.00745 | \$0.00730 | \$0.00730 |
| 3 Years | \$0.00730 | \$0.00730 | \$0.00715 | \$0.00715 |
| | 10,000 - 24,999 Minutes of Use 25,000 - 49,999 Minutes of Use | | | Minutes of Use |
| | Initial 6 Seconds | Each Add'l 6 Seconds | Initial 6 Seconds | Each Add'l 6 Seconds |
| 1 Year | \$0.00730 | \$0.00730 | \$0.00715 | \$0.00715 |
| 2 Years | \$0.00715 | \$0.00715 | \$0.00700 | \$0.00700 |
| 3 Years | \$0.00700 | \$0.00700 | \$0.00685 | \$0.00685 |
| | - 0.000 - 4.000 | 2.5 | | |
| | | 9 Minutes of Use | | Minutes of Use |
| | Initial 6 Seconds | Each Add'l 6 Seconds | Initial 6 Seconds | Each Add'l 6 Seconds |
| 1 Year | \$0.00700 | \$0.00700 | \$0.00685 | \$0.00685 |
| 2 Years | \$0.00685 | \$0.00685 | \$0.00670 | \$0.00670 |
| 3 Years | \$0.00670 | \$0.00670 | \$0.00655 | \$0.00655 |
| | 100,000 - 149,99 | 99 Minutes of Use | 150.000 - 199.99 | 99 Minutes of Use |
| | Initial 6 Seconds | Each Add'l 6 Seconds | Initial 6 Seconds | Each Add'l 6 Seconds |
| 1 Year | \$0.00670 | \$0.00670 | \$0.00655 | \$0.00655 |
| 2 Years | \$0.00655 | \$0.00655 | \$0.00640 | \$0.00640 |
| 3 Years | \$0.00640 | \$0.00640 | \$0.00625 | \$0.00625 |
| | 200 000 15 | CTT 1 1 | | |
| | | s of Use and above | | |
| 1.37 | Initial 6 Seconds | Each Add'l 6 Seconds | | |
| 1 Year | \$0.00640 | \$0.00640 | | |
| 2 Years | \$0.00625 | \$0.00625 | | |
| 3 Years | \$0.00605 | \$0.00605 | | |

2. <u>Dedicated IntraLATA and InterLATA Long Distance Services</u>

a. <u>Dedicated Inbound IntraLATA Service</u>

| Term Commitment | Monthly Minutes of Use | | | |
|-----------------|---|----------------------|--------------------------------------|----------------------|
| | <u> 500 - 2499 N</u> | Minutes of Use | 2,500 - 4,999 Minutes of Use | |
| | Initial 6 Seconds | Each Add'l 6 Seconds | Initial 6 Seconds | Each Add'l 6 Seconds |
| 1 Year | \$0.00690 | \$0.00690 | \$0.00675 | \$0.00675 |
| 2 Years | \$0.00675 | \$0.00675 | \$0.00660 | \$0.00660 |
| 3 Years | \$0.00660 | \$0.00660 | \$0.00645 | \$0.00645 |
| | | | | |
| | | Minutes of Use | | Minutes of Use |
| | Initial 6 Seconds | Each Add'l 6 Seconds | Initial 6 Seconds | Each Add'l 6 Seconds |
| 1 Year | \$0.00660 | \$0.00660 | \$0.00645 | \$0.00645 |
| 2 Years | \$0.00645 | \$0.00645 | \$0.00630 | \$0.00630 |
| 3 Years | \$0.00630 | \$0.00630 | \$0.00615 | \$0.00615 |
| | 10,000 - 24,999 Minutes of Use 25,000 - 49,999 Minutes of Use | | | Minutes of Hea |
| | Initial 6 Seconds | Each Add'l 6 Seconds | 25,000 - 49,999 Initial 6 Seconds | Each Add'l 6 Seconds |
| 1 Year | \$0.00630 | \$0.00630 | \$0.00615 | \$0.00615 |
| 2 Years | \$0.00615 | \$0.00615 | \$0.00613 | \$0.00600 |
| 3 Years | \$0.00600 | \$0.00613 | \$0.00585 | \$0.00585 |
| 3 Tears | \$0.0000 | \$0.00000 | \$0.00363 | \$0.00383 |
| | <u>50,000 - 74,99</u> | 9 Minutes of Use | <u>75,000 - 99,999</u> | Minutes of Use |
| | Initial 6 Seconds | Each Add'l 6 Seconds | Initial 6 Seconds | Each Add'l 6 Seconds |
| 1 Year | \$0.00600 | \$0.00600 | \$0.00585 | \$0.00585 |
| 2 Years | \$0.00585 | \$0.00585 | \$0.00570 | \$0.00570 |
| 3 Years | \$0.00570 | \$0.00570 | \$0.00555 | \$0.00555 |
| | 100 000 140 0 | 00.15 | 150,000, 100,00 | 20.3.5 |
| | | 99 Minutes of Use | | 99 Minutes of Use |
| 1.37 | Initial 6 Seconds | Each Add'l 6 Seconds | Initial 6 Seconds | Each Add'l 6 Seconds |
| 1 Year | \$0.00570 | \$0.00570 | \$0.00555 | \$0.00555 |
| 2 Years | \$0.00555 | \$0.00555 | \$0.00540 | \$0.00540 |
| 3 Years | \$0.00540 | \$0.00540 | \$0.00525 | \$0.00525 |
| | 200,000 Minutes | s of Use and above | | |
| | Initial 6 Seconds | Each Add'l 6 Seconds | | |
| 1 Year | \$0.00540 | \$0.00540 | | |
| 2 Years | \$0.00525 | \$0.00525 | | |
| 3 Years | \$0.00505 | \$0.00505 | | |

b. <u>Dedicated Inbound InterLATA Service</u>

| Term Commitment | Monthly Minutes of Use | | | |
|--------------------|------------------------|---|---------------------------------------|---|
| | 500 - 2499 N | Minutes of Use | 2,500 - 4,999 | Minutes of Use |
| | Initial 6 Seconds | Each Add'l 6 Seconds | Initial 6 Seconds | Each Add'l 6 Seconds |
| 1 Year | \$0.00690 | \$0.00690 | \$0.00675 | \$0.00675 |
| 2 Years | \$0.00675 | \$0.00675 | \$0.00660 | \$0.00660 |
| 3 Years | \$0.00660 | \$0.00660 | \$0.00645 | \$0.00645 |
| | | | | |
| | <u>5,000 - 7,499</u> | Minutes of Use | <u> 7,500 - 9,999 :</u> | Minutes of Use |
| | Initial 6 Seconds | Each Add'l 6 Seconds | Initial 6 Seconds | Each Add'l 6 Seconds |
| 1 Year | \$0.00660 | \$0.00660 | \$0.00645 | \$0.00645 |
| 2 Years | \$0.00645 | \$0.00645 | \$0.00630 | \$0.00630 |
| 3 Years | \$0.00630 | \$0.00630 | \$0.00615 | \$0.00615 |
| | | | | |
| | <u>10,000 - 24,999</u> | 9 Minutes of Use | <u> 25,000 - 49,999</u> | Minutes of Use |
| | Initial 6 Seconds | Each Add'l 6 Seconds | Initial 6 Seconds | Each Add'l 6 Seconds |
| 1 Year | \$0.00630 | \$0.00630 | \$0.00615 | \$0.00615 |
| 2 Years | \$0.00615 | \$0.00615 | \$0.00600 | \$0.00600 |
| 3 Years | \$0.00600 | \$0.00600 | \$0.00585 | \$0.00585 |
| | | | | |
| | | 9 Minutes of Use | | Minutes of Use |
| | Initial 6 Seconds | Each Add'l 6 Seconds | Initial 6 Seconds | Each Add'l 6 Seconds |
| 1 Year | \$0.00600 | \$0.00600 | \$0.00585 | \$0.00585 |
| 2 Years | \$0.00585 | \$0.00585 | \$0.00570 | \$0.00570 |
| 3 Years | \$0.00570 | \$0.00570 | \$0.00555 | \$0.00555 |
| | 100 000 140 0 | | 150,000, 100,00 | OOM: 4 CH |
| | Initial 6 Seconds | 99 Minutes of Use Each Add'l 6 Seconds | 150,000 - 199,99 Initial 6 Seconds | 99 Minutes of Use Each Add'l 6 Seconds |
| 1 Year | \$0.00570 | \$0.00570 | \$0.00555 | \$0.00555 |
| 2 Years | \$0.00570 \$0.00555 | \$0.00570 \$0.00555 | \$0.00535 \$0.00540 | \$0.00535 \$0.00540 |
| 2 Years 3 Years | \$0.00535 \$0.00540 | \$0.00535 \$0.00540 | \$0.00540 \$0.00525 | \$0.00540 \$0.00525 |
| 5 Tears | \$0.00340 | \$0.00340 | \$0.00323 | \$0.00323 |
| | 200.000 Minutes | s of Use and above | | |
| | Initial 6 Seconds | Each Add'l 6 Seconds | | |
| 1 Year | \$0.00540 | \$0.00540 | | |
| 2 Years | \$0.00525 | \$0.00525 | | |
| 3 Years | \$0.00505 | \$0.00505 | | |

c. <u>Dedicated Outbound IntraLATA Service</u>

| Term Commitment | Monthly Minutes of Use | | | |
|-------------------|--------------------------------|-----------------------------------|------------------------|----------------------|
| | 500 - 2499 N | Minutes of Use | 2,500 - 4,999 | Minutes of Use |
| | Initial 6 Seconds | Each Add'l 6 Seconds | Initial 6 Seconds | Each Add'l 6 Seconds |
| 1 Year | \$0.00690 | \$0.00690 | \$0.00675 | \$0.00675 |
| 2 Years | \$0.00675 | \$0.00675 | \$0.00660 | \$0.00660 |
| 3 Years | \$0.00660 | \$0.00660 | \$0.00645 | \$0.00645 |
| | | | | |
| | <u> 5,000 - 7,499</u> | Minutes of Use | <u>7,500 - 9,999</u> | Minutes of Use |
| | Initial 6 Seconds | Each Add'l 6 Seconds | Initial 6 Seconds | Each Add'l 6 Seconds |
| 1 Year | \$0.00660 | \$0.00660 | \$0.00645 | \$0.00645 |
| 2 Years | \$0.00645 | \$0.00645 | \$0.00630 | \$0.00630 |
| 3 Years | \$0.00630 | \$0.00630 | \$0.00615 | \$0.00615 |
| | | | | |
| | <u> 10,000 - 24,999</u> | 9 Minutes of Use | <u>25,000 - 49,999</u> | Minutes of Use |
| | Initial 6 Seconds | Each Add'l 6 Seconds | Initial 6 Seconds | Each Add'l 6 Seconds |
| 1 Year | \$0.00630 | \$0.00630 | \$0.00615 | \$0.00615 |
| 2 Years | \$0.00615 | \$0.00615 | \$0.00600 | \$0.00600 |
| 3 Years | \$0.00600 | \$0.00600 | \$0.00585 | \$0.00585 |
| | | | | |
| | | 9 Minutes of Use | | 9 Minutes of Use |
| | Initial 6 Seconds | Each Add'l 6 Seconds | Initial 6 Seconds | Each Add'l 6 Seconds |
| 1 Year | \$0.00600 | \$0.00600 | \$0.00585 | \$0.00585 |
| 2 Years | \$0.00585 | \$0.00585 | \$0.00570 | \$0.00570 |
| 3 Years | \$0.00570 | \$0.00570 | \$0.00555 | \$0.00555 |
| | 400 000 440 0 | | 4.50.000 400.00 | 22.5 |
| | | 99 Minutes of Use | | 99 Minutes of Use |
| | Initial 6 Seconds | Each Add'l 6 Seconds | Initial 6 Seconds | Each Add'l 6 Seconds |
| 1 Year | \$0.00570 | \$0.00570 | \$0.00555 | \$0.00555 |
| 2 Years | \$0.00555 | \$0.00555 | \$0.00540 | \$0.00540 |
| 3 Years | \$0.00540 | \$0.00540 | \$0.00525 | \$0.00525 |
| | 200 000 Minutes | of Han and above | | |
| | | S of Use and above | | |
| 1 Year | Initial 6 Seconds \$0.00540 | Each Add'l 6 Seconds \$0.00540 | | |
| 1 Year 2 Years | | | | |
| | \$0.00525 | \$0.00525 | | |
| 3 Years | \$0.00505 | \$0.00505 | | |

d. <u>Dedicated Outbound InterLATA Service</u>

| Substitute | | |
|--|------------------------------|--|
| 1 Year \$0.00690 \$0.00690 \$0.00675 \$0.00675 2 Years \$0.00675 \$0.00675 \$0.00660 \$0.00660 3 Years \$0.00660 \$0.00660 \$0.00645 \$0.00645 | 2,500 - 4,999 Minutes of Use | |
| 2 Years \$0.00675 \$0.00675 \$0.00660 \$0.00660 3 Years \$0.00660 \$0.00660 \$0.00645 \$0.00645 | econds | |
| 3 Years \$0.00660 \$0.00660 \$0.00645 \$0.00645 | | |
| | | |
| 5,000 - 7,499 Minutes of Use 7,500 - 9,999 Minutes of Use | | |
| 5.000 - 7.499 Minutes of Use 7.500 - 9.999 Minutes of Use | | |
| | | |
| Initial 6 Seconds Each Add'l 6 Seconds Initial 6 Seconds Each Add'l 6 Seconds | | |
| 1 Year \$0.00660 \$0.00665 \$0.00645 \$0.00645 | | |
| 2 Years \$0.00645 \$0.00645 \$0.00630 \$0.00630 | | |
| 3 Years \$0.00630 \$0.00630 \$0.00615 \$0.00615 | | |
| 10,000 - 24,999 Minutes of Use 25,000 - 49,999 Minutes of Use | | |
| Initial 6 Seconds Each Add'l 6 Seconds Initial 6 Seconds Each Add'l 6 Seconds | econds | |
| 1 Year \$0.00630 \$0.00630 \$0.00615 \$0.00615 | | |
| 2 Years \$0.00615 \$0.00615 \$0.00600 \$0.00600 | | |
| 3 Years \$0.00600 \$0.00600 \$0.00585 \$0.00585 | | |
| | | |
| <u>50,000 - 74,999 Minutes of Use</u> <u>75,000 - 99,999 Minutes of Use</u> | | |
| <u>Initial 6 Seconds</u> <u>Each Add'l 6 Seconds</u> <u>Initial 6 Seconds</u> <u>Each Add'l 6 Seconds</u> | | |
| 1 Year \$0.00600 \$0.00600 \$0.00585 \$0.00585 | | |
| 2 Years \$0.00585 \$0.00585 \$0.00570 \$0.00570 | | |
| 3 Years \$0.00570 \$0.00570 \$0.00555 \$0.00555 | | |
| 100,000 - 149,999 Minutes of Use 150,000 - 199,999 Minutes of Use | | |
| Initial 6 Seconds Each Add'l 6 Seconds Initial 6 Seconds Each Add'l 6 Seconds | aconde | |
| 1 Year \$0.00570 \$0.00570 \$0.00555 \$0.00555 | | |
| 2 Years \$0.00570 \$0.00555 \$0.00555 \$0.00540 \$0.00540 | | |
| 3 Years \$0.00535 \$0.00540 \$0.00545 \$0.00545 \$0.00525 | | |
| 5 Teals \$0.00340 \$0.00340 \$0.00323 \$0.00323 | | |
| 200,000 Minutes of Use and above | | |
| <u>Initial 6 Seconds</u> <u>Each Add'l 6 Seconds</u> | | |
| 1 Year \$0.00540 \$0.00540 | | |
| 2 Years \$0.00525 \$0.00525 | | |
| 3 Years \$0.00505 \$0.00505 | | |

C. Miscellaneous Monthly Recurring and Non-Recurring Charges *

1. Toll Free Service

Customers subscribing to the Company's toll-free inbound long distance services shall be assessed the following monthly recurring charge per toll-free number subscribed:

| | <u>Minimum</u> | <u>Maximum</u> |
|--|----------------|----------------|
| Toll Free Service Monthly Recurring Charge | | |
| per toll free number | \$0.01 | \$20.00 |

2. Project Account Codes

Project Account Codes ("PAC") permit Customers to assign telephone usage to a particular coded project of the Customer's choosing. Where a Customer elects to use non-verified codes, any code using the proper number of digits (2, 3, or 4) allows completion of the call. Where a Customer elects to use verified codes, call completion occurs only after a specific 2, 3, or 4-digit code (selected by the Customer from an existing index) is keyed in. Where a Customer elects to use custom codes, call completion occurs only after a specific verified code (between 2 to 8 digits) established by the Customer is keyed in.

| Code | Non-Recurring Set-Up Charge | Monthly Recurring Charge | Change Charge, per change order |
|--------------|--------------------------------|-----------------------------|---------------------------------|
| Non-Verified | N/A | \$5.00 | \$5.00 |
| Verified | N/A | \$5.00 | \$5.00 |
| Custom | \$50.00 | \$5.00 | \$5.00 |

14.1.2.13 CoreComm Business Switched Long Distance

A. General

The Company offers MTS intraLATA, interLATA intrastate and interLATA interstate 1+ outbound long distance telecommunications service utilizing switched or dedicated access arrangements between Customer premises and the Company's facilities for call origination. Call termination is completed through a combination of Company facilities and LEC switched access arrangements. This plan is for business customers only. See Section 4.3.4 of this tariff for surcharges and tax rate information.

B. Timing of Calls

Usage charges are based on the actual usage of the Company network. Chargeable time begins when a connection is established between the Calling Station and the Called Station. Chargeable time ends when either party "hangs up" thereby releasing the network connection. Unless otherwise specified, for billing purposes, the minimum call duration is thirty (30) seconds. In addition, unless otherwise specified, usage is measured thereafter in six (6) second increments and rounded to the next higher six (6) second period. Fractional amounts are rounded up to the next highest whole cent. International call usage is measured in six (6) second increments and has a minimum per call duration of sixty (60) seconds.

C. Usage Rates

| IntraLATA / In | nterLATA / 8XX |
|--------------------|--|
| Initial 30 Seconds | Each Add'l 6 Seconds |
| \$0.03450 | \$0.00690 |
| \$0.03125 | \$0.00625 |
| \$0.02950 | \$0.00590 |
| | Initial 30 Seconds \$0.03450 \$0.03125 |

D. Per 8XX Number Charge

For each 8XX number that the customer subscribes to, the following charge will apply in addition to the per-minute rates listed in Section C above.

Monthly charge \$4.00

E. <u>Project Account Codes</u>

Customers choosing the Company's Project Account Codes ("PAC") service shall be assessed the following charges. PACs permit Customers to assign telephone usage to a particular coded project of the Customer's choosing.

| Description | Non-Recurring Set- up Charge | Monthly Recurring Charge |
|-------------|---------------------------------|--|
| Charge | \$15.00 | \$5.00 |

14.1.2.14 CoreComm Business Dedicated Long Distance

A. General Terms

CoreComm Dedicated Long Distance service is available where facilities permit to business customers who make 1-year, 2-year, or 3-year term commitment subscription to the Company's inbound and outbound long distance services. Services will be provided via T-1 arrangement.

B. <u>Billing Terms and Timing of Usage</u>

Rates are based on the Customer's commitment to a certain term and minimum monthly usage volume. Charges are based on actual usage of the Company's network. Chargeable time begins when a connection is established between the calling station and the called station. Chargeable time ends when either party "hangs up" at either station, thereby releasing the network connection. Unless otherwise specified, for billing purposes, the minimum call duration for is thirty (30) seconds. In addition, unless otherwise specified, usage is measured thereafter in six (6) second increments and rounded to the next higher six (6) second period. Fractional amounts are rounded up to the next highest whole cent.

C. <u>Installation Charges</u>

The following installation charges apply when a customer subscribes to CoreComm Dedicated Long Distance service.

| <u>Description</u> | Non-recurring Charge |
|----------------------------------|----------------------|
| Account code charge installation | \$15.00 |
| Installation Fee | \$899.00 |

D. <u>Monthly Charges</u>

The following monthly charges apply when a customer subscribes to CoreComm Dedicated Long Distance service.

| <u>Description</u> | Monthly Recurring Charge |
|------------------------------|--------------------------|
| T1 | \$150.00 |
| 8XX Number Charge Per Number | \$4.00 |

E. <u>Minute of Use Charges for Interstate, Intrastate and 8XX Long Distance Services</u>

| Term Commitment | | Monthly Min | utes of Use | |
|-----------------|---------------------------------|-----------------------|--------------------|----------------------|
| | <u>0 – 10,000 N</u> | <u>Vinutes of Use</u> | 10,000 - 20,000 | O Minutes of Use |
| | Initial 30 Seconds | Each Add'l 6 Seconds | Initial 30 Seconds | Each Add'l 6 Seconds |
| 1 Year | \$0.04000 | \$0.00800 | \$0.02500 | \$0.00500 |
| 2 Years | \$0.04000 | \$0.00800 | \$0.02375 | \$0.00475 |
| 3 Years | \$0.04000 | \$0.00800 | \$0.02250 | \$0.00450 |
| | | | | |
| | 20,000 - 35,000 | 0 Minutes of Use | 35,000 - 50,000 | O Minutes of Use |
| | Initial 30 Seconds | Each Add'l 6 Seconds | Initial 30 Seconds | Each Add'l 6 Seconds |
| 1 Year | \$0.02250 | \$0.00450 | \$0.02000 | \$0.00400 |
| 2 Years | \$0.02000 | \$0.00400 | \$0.01750 | \$0.00350 |
| 3 Years | \$0.01750 | \$0.00350 | \$0.01625 | \$0.00325 |
| | | | | |
| | 50,000 – 100,000 Minutes of Use | | 100,000 Minutes | of Use and Above |
| | Initial 30 Seconds | Each Add'l 6 Seconds | Initial 30 Seconds | Each Add'l 6 Seconds |
| 1 Year | \$0.01750 | \$0.00350 | \$0.01500 | \$0.00300 |
| 2 Years | \$0.01625 | \$0.00325 | \$0.01375 | \$0.00275 |
| 3 Years | \$0.01500 | \$0.00300 | \$0.01250 | \$0.00250 |

14.2.2 PRIVATE LINE Inter-POP Service

<u>LEASED LINE Service</u> (90 day minimum)

| MILEAGE/CKT | RATE/MONTH/CIRCUIT |
|-------------|------------------------|
| 1-50 | \$50.00 + \$2.00/mile |
| 51-100 | \$100.00 + \$1.00/mile |
| 101 + | \$150.00 + \$0.50/mile |

HIGH CAPACITY Service

DS-1 Service

| | <u>IntraLATA</u> | |
|-----------------------|-----------------------------|-----------------------|
| Contract Term | Fixed Charge/Month/ Circuit | Per Mile Charge/Month |
| Monthly (90 Day Min.) | \$570.00 | \$17.30 |
| 1-Year | \$540.00 | \$16.40 |
| 2-Year | \$510.00 | \$15.50 |
| 5-Year | \$455.00 | \$13.75 |
| | <u>InterLATA</u> | |
| Contract Term | Fixed Charge/Month/ Circuit | Per Mile Charge/Month |
| Monthly (90 Day Min.) | \$1,545.00 | \$14.55 |
| 1-Year | \$1,470.00 | \$13.85 |
| 2-Year | \$1,400.00 | \$13.20 |
| 5-Year | \$1,200.00 | \$11.25 |
| | | |

DS-3 Service

| Contract Term | Fixed Charge/Month/ Circuit | Per Mile Charge/Month |
|-----------------------|-----------------------------|-----------------------|
| Monthly (90 Day Min.) | \$5,700.00 | \$138.40 |
| 1-Year | \$5,400.00 | \$131.20 |
| 2-Year | \$5,100.00 | \$124.00 |
| 5-Year | \$4,550.00 | \$110.00 |
| | <u>InterLATA</u> | |
| Contract Term | Fixed Charge/Month/ Circuit | Per Mile Charge/Month |
| Monthly (90 Day Min.) | \$12,000.00 | \$110.00 |
| 1-Year | \$11,000.00 | \$105.00 |
| 2-Year | \$10,000.00 | \$100.00 |
| 5-Year | \$9,000.00 | \$95.00 |

IntraLATA

<u>Installation Charge (For DS-1 and DS-3 Service)</u>

Installation charge will be equal to one month's term per circuit, \$600.00 circuit minimum (non-recurring charge)

DACS Charge per T-1 (For DS-1)

\$250.00 (non-recurring charge)

\$250.00 (monthly recurring charge)

Expedite Charge

\$250.00 (non-recurring charge for DS-1)

\$500.00 (non-recurring charge for DS-3)

Cancellation Charge

\$250.00 (non-recurring charge for DS-l)

\$500.00 (non-recurring charge for DS-3)

Change Charges

Administrative Changes

\$50.00 (non-recurring charge for DS-1 and DS-3 Service)

Physical Changes

\$100.00 (non-recurring charge for DS-l)

\$250.00 (non-recurring charge for DS-3)

Billing Changes

\$50.00 (non-recurring charge for DS-1 and DS-3 Service)

Cross Connect Charges

\$200.00 (non-recurring charge for DS-I)

\$300.00 (non-recurring charge for DS-3)

Multiplex Fee

\$700.00 (non-recurring charge for DS-3)

\$700.00 (monthly recurring charge for DS-3)

DSU/CSU Charge

\$900.00 (non-recurring charge for DS-l)

\$ 50.00 (monthly recurring charge for DS-l)

Unprotected DS- I and DS-3 Service

The charges for unprotected DS-1 and DS-3 circuits may be 35%

of the standard charges for DS-1 and DS-3 Service as set forth in this Price List.

14.2.3 Dedicated Access Service

Access facilities from Customer's premises to Carrier's POP may be required at one or both ends of Carrier's PRIVATE LINE Service. Access arrangements provided by Carrier can be either by access lines leased from LEC, access lines leased from Carrier's own Dedicated Access Service Tariff, or other access arrangement.

T-1 Access Service

For a twenty-four (24) channel equivalent dedicated facility between Customer's premises and Carrier's POP:

| | Fixed Charge/ | Per Mile | |
|-----------------------|---------------|--------------|----------------------------|
| Contract Term | Month/Circuit | Charge/Month | Installation Charge |
| Monthly (90 Day Min.) | \$400.00 | \$20.00 | \$1,200.00 |
| 1-Year | \$340.00 | \$18.00 | \$1,200.00 |
| 2-Year | \$320.00 | \$16.00 | \$1,200.00 |
| 5-Year | \$300.00 | \$13.00 | \$1,200.00 |

| Co-located T-l Access (per T-l) | Installation Charge |
|-------------------------------------|---------------------|
| | \$300.00 |
| Single Channel Access (per circuit) | |
| If obtained from LEC: | \$300.00 |
| If co-located at POP: | \$100.00 |

Any LEC charges or surcharges are additional.

Dedicated Access Service

| Rate Per Minute. per call | \$0.12 |
|---------------------------|--------|
| Discount Schedule | |
| \$0 - \$7,999.99 | 0% |
| \$8,000.00 - Over | 15% |

Dedicated Access 800 Service

| Rate Per Minute, per call | \$0.15 |
|---------------------------------------|--------|
| Discount Schedule \$0 - \$2,999.00 | 0% |
| \$3,000.00 - Over | 15% |

14.2.4 <u>Miscellaneous Services</u>

14.2.4.1 <u>Directory Assistance</u>

Local Direct Dial \$1.50

National Direct Dial \$1.99

Customers will be limited to two (2) requests per call. A credit allowance for Directory Assistance will be provided upon request if Customer experiences poor transmission quality, is cut off, receives an incorrect telephone number, or misdials the intended Directory Assistance number.

14.2.4.2 Operator Services

| <u>Service</u> | Call Placement Charge |
|---|------------------------------------|
| | (Added to the first minute of each |
| | <u>call)</u> |
| Customer dialed/LCCS Calling Card/station-to-station | \$1.70 |
| Operator-handled/station-to-station (collect, bill to third number) | \$2.50 |
| Person-to-Person Calls | \$4.80 |

Carrier accepts only cards which it can identify as valid. Usage and Call Placement Charges for LCCS Calling Card calls appear on the calling card bill for both CoreComm and non-CoreComm Customers.

14.2.4.3 <u>Inmate Calling Service</u>

| Inmate Calling Service Operator Service Charges Operator Handled Station-to-Station Collect Person-to-Person Collect | \$2.50 \$4.80 |
|--|------------------|
| Inmate Calling Service Usage Rates per minute | \$0.36 |

14.2.4.4 <u>CoreComm Calling Card Service</u> - Customers

Peak Period \$0.26
Off-Peak Period \$0.22
Enhanced Services \$0.39

<u>Surcharge</u> \$0.25 per call, regardless of the length of the call.

Payphone Usage Charge \$1.25 per call

Discount

A 10% discount shall be applied to each Customer's total calling card usage during the billing period if the customer also subscribes to Carrier's cellular long distance service.

Effective September 1, 1997, Carrier no longer will apply this discount to new customers. However, this discount will continue to be made available to existing customers of CoreComm Calling Card Service.

14.2.4.5 Restoration of Service Charge

Non-Recurring Charge

\$15.00

14.2.4.6 Project Account ("PAC") Codes

| | Non-Recurring Charge | Monthly Charge |
|---|----------------------|----------------|
| Charge Per PAC Code | | \$5.00 |
| Feature Activation Charge | \$15.00 | |
| Customer-Specific Verified PAC Index Activation | \$100.00 | |

Effective June 1, 2021

Addendum to Price List

Promotional Offerings

Beginning on November 3, 1997, Carrier will offer the following promotional program for select new customers:

From time to time, select new customers to CoreComm Calling Card Service may be entitled to a calling card with 15 free minutes included on it which the customer will have up to 90 days to use. Unused minutes cannot be transferred to succeeding months. This offer is good on direct dialed domestic long distance service only.

Beginning on January 30, 1998, Carrier will offer the following promotional program for Land Line Long Distance Service customers:

New Customers to CoreComm Service between January 30, 1998 and April 30, 1998 are entitled to up to \$5.00 off long distance usage for each of the first 10 months of service. Because CoreComm has several billing cycles, a customer's first bill may include only a partial month's charges. As a result, for purposes of this promotion, the first month's credit may be prorated unless the billing cycle date is the date of activation. The second month will constitute a full thirty day period and so on. If the first month was prorated, the customer will receive the remainder of their first month's credit in month 11. Unused credits in any given month cannot be transferred to succeeding months. This offer is good on direct dialed domestic long distance service only.

Effective June 1, 2021

15. Smart Pack

15.2 <u>Smart Pack Miscellaneous Charges</u>

15.2.2 <u>CSR/Conversion Charges</u>

| Term Commitment | <u>Charge Per CSR</u> |
|-------------------------------|-------------------------------|
| Month-to-Month 1 Year 2 Years | \$20.00 \$15.00 \$10.00 |
| 3 Years | \$5.00 * |

15.2.3 <u>Installation Charges</u>

| Term Commitment | Charge Per Installation | | |
|-----------------|-------------------------|--|--|
| Month-to-Month | \$33.50 | | |
| 1 Year | \$33.50 | | |
| 2 Years | \$33.50 | | |
| 3 Years | \$33.50 | | |

15.2.4 Service Order Charge

| Term Commitment | Charge Per Order |
|-----------------------|--------------------|
| Month-to-Month 1 Year | \$20.00 \$20.00 |
| 2 Years 3 Years | \$20.00 \$20.00 |

15.2.5 Service Change Charge

| | Charge Per Change |
|---------------------|-------------------|
| Message to Measured | \$41.55 |
| Measured to Message | \$41.55 |

* Waived

15.3 Smart Pack Local Services

15.3.2 <u>Smart Pack Local Services -- Line and Usage Charges</u>

Term Commitment

A. Access Line Charges*

The following line charges apply to Smart Pack Customers in lieu of those access line charges provided for in Section 4.3.3 of this tariff. All such business line charges indicated below include both the charge for the access line, line hunting service and any three (3) features in Section 15.3.3 of this tariff. Additional features are available at the rates contained within Section 15.3.3.

NOTE: Effective February 20, 2003, the Month-to-Month term option will be offered to current customers only. New customers can choose between 1, 2 and 3-year term plans.

Monthly Charge

| • • |
|----------------------|
| Zone 1 - \$124.95 |
| 2 - \$124.95 |
| 3 - \$124.95 |
| Zone |
| 1 - \$124.95 |
| 2 - \$124.95 |
| 3 - \$124.95 |
| Zone |
| 1 - \$124.95 |
| 2 - \$124.95 |
| 3 - \$124.95 |
| \$124.95 |
| |

Business line charges do not include the End User Common Line¹, 911, Local Number Portability, and other surcharges provided for in Section 4.3.4 of this tariff.

B. <u>Local Usage Charges</u>

| 1. <u>Message Rate Service</u> Term Commitment | Rate Per Call |
|---|--|
| Month-to-Month 1 Year 2 Years 3 Years | \$0.0800 \$0.0600 \$0.0575 \$0.0550 |

| | Measured Rate Service Term Commitment | With Long Distance | | With Local Service Only | |
|--|---------------------------------------|-----------------------|---------------------|-------------------------|---------------------|
| | | Initial 30 Seconds | Add'l. 6 Seconds | Initial 30 Seconds | Add'l. 6 Seconds |
| | 1 Year | \$0.00875 | \$0.00175 | \$0.01500 | \$0.00300 |
| | 2 Years | \$0.00750 | \$0.00150 | \$0.01375 | \$0.00275 |
| | 3 Years | \$0.00625 | \$0.00125 | \$0.01250 | \$0.00250 |

¹ The End User Common Line Charge can be located in CoreComm's FCC tariff.

Effective June 1, 2021

^{*} Customers purchasing service under this plan or package may qualify for the Multi Product Credit. The Multi Product Credit provides Customers purchasing qualifying commercial telephone line package(s) at the same service address as an On Net 1.5mb or greater Circuit, including, but not limited to, the Essential products (VoIP), with an \$8 credit on the Customer's commercial telephone line package(s). In the event of On Net circuit disconnection, the Multi Product Credit shall no longer be available.

C. Unlimited local usage line

1st line Zone 1 - \$124.95 Zone 2 - \$124.00 Zone 3 - \$124.00

Additional line Zone 1 - \$124.95

Zone 2 - \$124.95

Zone 3 - \$124.95

15.3.3 Smart Pack Local Services -- Service Enhancement Features

A. <u>Standard Features</u>

All lines are pre-equipped with the following features, as well as line hunting service, at no additional charge. Charges apply on a per line, per month basis unless otherwise indicated:

| | Rate |
|---------------------|--------|
| Automatic Callback | \$7.25 |
| Call Trace, per use | \$3.50 |
| Repeat Dialing | \$4.00 |
| Three-Way Calling | \$4.00 |

B. Optional Features

Customers may additionally subscribe to the following features. Charges apply on a per line, per month basis unless otherwise indicated:

| | Rate |
|-------------------------|---------|
| 900 Call Blocking | \$0.00 |
| 976 Call Blocking | \$0.00 |
| Alternate Answering | \$0.75 |
| Busy Line Transfer | \$0.75 |
| Call Forwarding | \$4.00 |
| Remote Call Forwarding | \$34.95 |
| Call Waiting | \$5.00 |
| Caller ID | \$6.50 |
| Caller ID With Name | \$7.25 |
| Multi-Ring Service | |
| 1 st Number | \$4.00 |
| 2 nd Number | \$3.95 |
| Speed Calling - 8 or 30 | \$4.00 |

15.4 <u>Smart Pack Long Distance Service</u>

Unless otherwise specified, for billing purposes, the minimum call duration is thirty (30) seconds. In addition, unless otherwise specified, usage is measured thereafter in six (6) second increments and rounded to the next higher six (6) second period. Fractional amounts are rounded up to the next highest whole cent. International call usage is measured in six (6) second increments and has a minimum per call duration of sixty (60) seconds.

NOTE: Effective February 20, 2003, the Month-to-Month term option will be offered to current customers only. New customers can choose between 1, 2 and 3-year term plans. Additionally, Dedicated Services Rates will be offered to current customers only as of February 20, 2003. New customers can subscribe to Switched Service Rates below.

15.4.2 <u>Switched Services Rates</u>

| Term Commitment | Initial 30 Seconds | Each Add'l 6 Seconds |
|-----------------|--------------------|----------------------|
| Month-to-Month | \$0.01000 | \$0.01000 |
| 1 Year | \$0.03000 | \$0.00600 |
| 2 Years | \$0.02875 | \$0.00575 |
| 3 Years | \$0.02750 | \$0.00550 |

15.4.3 Dedicated Services Rates

| Term Commitment | Initial 6 Seconds | Each Add'l 6 Seconds |
|-----------------|-------------------|----------------------|
| Month-to-Month | \$0.00690 | \$0.00690 |
| 1 Year | \$0.00600 | \$0.00600 |
| 2 Years | \$0.00580 | \$0.00580 |
| 3 Years | \$0.00560 | \$0.00560 |

15.5 <u>Smart Pack Association Member Discount</u>

15.5.1 General

Smart Pack Association Member Discount: 5%

15.6 Smart Pack Centrex

15.6.1 Line Rates

Smart Pack Centrex provides subscribers access to the public switched network with specified Centrex features. The following rates do not include federal access, 911 and local number portability surcharges in Section 4.3.4 of this tariff. The remaining charges provided for in Section 4.3.4 of this tariff apply to each line in addition to the monthly rates.

| Term Commitment | Monthly Line Charge, Per Line | |
|-----------------|-------------------------------|--|
| 1 Year | \$124.95 | |
| 2 Years | \$124.95 | |
| 3 Years | \$124.95 | |

15.6.2. Local Usage Charges

The following local usage charges apply to Smart Pack Centrex customers. Customers may request billing on a per call basis (message) or a per minute basis (measured) as provided in Sections 15.6.2.A and 15.6.2.B following, respectively.

A. Message Rate Service

The following per call rates apply to those Customers subscribing to the Smart Pack Message Rate Service:

| Term Commitment | Rate Per Call |
|-----------------|---------------|
| 1 Year | \$0.0600 |
| 2 Years | \$0.0575 |
| 3 Years | \$0.0550 |

B. Measured Rate Service

The following per minute rates apply to those Customers subscribing to the Smart Pack Measured Rate Service. If the customer chooses to subscribe to the Company's local service only, then the per-minute rates under "With Local Service Only" below apply. Unless otherwise specified, for billing purposes, the minimum call duration is thirty (30) seconds. In addition, unless otherwise specified, usage is measured thereafter in six (6) second increments and rounded to the next higher six (6) second period. Fractional amounts are rounded up to the next highest whole cent. International call usage is measured in six (6) second increments and has a minimum per call duration of sixty (60) seconds.

15.6 Smart Pack Centrex (continued)

15.6.2. Local Usage Charges (continued)

B. <u>Measured Rate Service (continued)</u>

| Term Commitment | With Long | Distance | With Local S | ervice Only |
|-----------------|------------|-----------|--------------|-------------|
| | Initial 30 | Add'l. 6 | Initial 30 | Add'l. 6 |
| | Seconds | Seconds | Seconds | Seconds |
| 1 Year | \$0.00875 | \$0.00175 | \$0.01500 | \$0.00300 |
| 2 Years | \$0.00750 | \$0.00150 | \$0.01375 | \$0.00275 |
| 3 Years | \$0.00625 | \$0.00125 | \$0.01250 | \$0.00250 |

C. Smart Pack Centrex Switched Long Distance Service

Smart Pack Centrex customers may subscribe to the Company's switched long distance service under the rates terms and conditions outlined in Section 15.4 of this tariff.

Smart Pack Trunks

15.7.1 General

General Smart Pack Trunk service bundles the Company's trunking service with local and long distance for business customers who want connectivity to the public switched network through PBX or DID trunks. The Company provides this connectivity for one, two or three-year terms with specified features such as blocks of 20 station numbers for DID service and Call Forward Trunk Group.

15.7.2 Trunk Charges

The End User Common Line¹, 911, Local Number Portability and remaining surcharges contained in Section 4.3.4 of this tariff are in addition to the trunk rate.

| Term Commitment | Monthly PBX Trunk Charge | Monthly DID Trunk Rate |
|-------------------|--------------------------|--|
| 1 Year 2 Years | \$124.95 \$124.95 | \$124.95 \$124.95 |
| 3 Years | \$124.95 | Zone 1 - \$124.95 2 - \$124.95 3 - \$124.95 |

15.7.3 Features

The following features can be added to Smart Pack Trunk service.

| Term Commitment | Station Numbers (block of 20) | Call Forward Trunk Group |
|-----------------|-------------------------------|--------------------------|
| 1 Year | \$3.00 | \$0.00 |
| 2 Years | \$3.00 | \$0.00 |
| 3 Years | \$3.00 | \$0.00 |

15.7.4 <u>Local Usage Charges</u>

The following local usage charges apply to Smart Pack Trunk customers. Customers may request billing on a per call basis (message) or a per minute basis (measured) as provided in Sections 15.7.4.A and 15.7.4.B following, respectively.

¹ The End User Common Line Charge can be located in CoreComm's FCC tariff.

15.7 Smart Pack Trunk Service (continued)

15.7.4 <u>Local Usage Charges (continued)</u>

A. Message Rate Service

The following per call rates apply to those Customers subscribing to the Smart Pack Message Rate Service:

| Term Commitment | Rate Per Call |
|-----------------|---------------|
| 1 Year | \$0.0600 |
| 2 Years | \$0.0575 |
| 3 Years | \$0.0550 |

B. Measured Rate Service

The following per minute rates apply to those Customers subscribing to the Smart Pack Measured Rate Service. If the customer chooses to subscribe to the Company's local service only, then the per-minute rates under "With Local Service Only" below apply. Unless otherwise specified, for billing purposes, the minimum call duration is thirty (30) seconds. In addition, unless otherwise specified, usage is measured thereafter in six (6) second increments and rounded to the next higher six (6) second period. Fractional amounts are rounded up to the next highest whole cent. International call usage is measured in six (6) second increments and has a minimum per call duration of sixty (60) seconds.

| Term Commitment | With Long | Distance | With Local S | ervice Only |
|-----------------|------------|-----------|--------------|-------------|
| | Initial 30 | Add'l. 6 | Initial 30 | Add'l. 6 |
| | Seconds | Seconds | Seconds | Seconds |
| 1 Year | \$0.00875 | \$0.00175 | \$0.01500 | \$0.00300 |
| 2 Years | \$0.00750 | \$0.00150 | \$0.01375 | \$0.00275 |
| 3 Years | \$0.00625 | \$0.00125 | \$0.01250 | \$0.00250 |

B. Smart Pack Trunk Switched Long Distance Service

Smart Pack Trunk customers may subscribe to the Company's switched long distance service under the rates terms and conditions outlined in Section 15.4 of this tariff.

16. CorePack

16.2 CorePack Business Services

16.2.2 CorePack Basic

C. Rates*

| Term Agreement | Monthly Rate |
|----------------------------|--------------|
| - Month to Month, per line | \$46.95 |
| - 1 Year Term, per line | Zone |
| , 1 | 1 - \$55.95 |
| | 2 - \$67.95 |
| | 3 - \$67.95 |
| - 2 Year Term, per line | Zone |
| , 1 | 1 - \$53.95 |
| | 2 - \$55.95 |
| | 3 - \$55.95 |
| - 3 Year Term, per line | Zone |
| | 1 - \$51.95 |
| | 2 - \$53.95 |
| | 3 - \$53.95 |

Charges provided for in Section 4.3.4 of this Tariff apply to each line in addition to the monthly rates provided above. Service Ordering Charges also apply as provided in Section 4.1 of this tariff. Operator assistance surcharges apply as provided in Section 4.6.2 of this tariff.

| <u>Unlimited local</u> | Monthly Rate |
|------------------------|--------------|
| 1 st line | Zone |
| | 1 - \$61.00 |
| | 2 - \$63.00 |
| | 3 - \$63.00 |
| Additional line | Zone |
| | 1 - \$54.00 |
| | 2 - \$56.00 |
| | 3 - \$56.00 |

^{*} Customers purchasing service under this plan or package may qualify for the Multi Product Credit. The Multi Product Credit provides Customers purchasing qualifying commercial telephone line package(s) at the same service address as an On Net 1.5mb or greater Circuit, including, but not limited to, the Essential products (VoIP), with an \$8 credit on the Customer's commercial telephone line package(s). In the event of On Net circuit disconnection, the Multi Product Credit shall no longer be available.

16.2.3 CorePack Plus

C. Rates

1. <u>Monthly Charges*</u>

| Term Agreement | Monthly Rate |
|----------------------------|--------------|
| - Month to Month, per line | \$74.95 |
| - 1 Year Term, per line | \$70.95 |
| - 2 Year Term, per line | \$67.95 |
| - 3 Year Term, per line | Zone |
| - | 1 - \$66.95 |
| | 2 - \$68.95 |
| | 3 – \$68 95 |

Charges provided for in Section 4.3.4 of this Tariff apply to each line in addition to the monthly rates provided above. Service Ordering Charges also apply as provided in Section 4.1 of this tariff. Operator assistance surcharges apply as provided in Section 4.6.2 of this tariff. Miscellaneous service charges apply as provided in Section 14.2.4 of this tariff.

* Customers purchasing service under this plan or package may qualify for the Multi Product Credit. The Multi Product Credit provides Customers purchasing qualifying commercial telephone line package(s) at the same service address as an On Net 1.5mb or greater Circuit, including, but not limited to, the Essential products (VoIP), with an \$8 credit on the Customer's commercial telephone line package(s). In the event of On Net circuit disconnection, the Multi Product Credit shall no longer be available.

2. <u>Usage Charges</u>

a. <u>Switched Outbound IntraLATA and InterLATA</u>

For switched outbound long distance usage in excess of the monthly CP Plus allotment on a per line basis, the following charges apply for usage:

| Term Agreement | Initial 6 Seconds | Each Add'l 6 Seconds |
|------------------|-------------------|----------------------|
| - Month to Month | \$0.0100 | \$0.0100 |
| - 1 Year Term | \$0.0090 | \$0.0090 |
| - 2 Year Term | \$0.0085 | \$0.0085 |
| - 3 Year Term | \$0.0080 | \$0.0080 |

For billing purposes, the minimum call duration for all interLATA and intraLATA switched outbound long distance usage is six (6) seconds. Usage is measured thereafter in six (6) second increments and rounded to the next higher six (6) second period.

Customers may carry over unused minutes of the monthly 240 minute per line long distance usage allotment to other CP Plus lines. Unused minutes may not, however, be carried over to subsequent months, nor is the Customer entitled to a credit for such unused minutes.

Effective June 1, 2021

b. Switched Inbound IntraLATA and InterLATA

Charges apply for all switched inbound intraLATA and interLATA usage as follows:

| Term Agreement | Initial 6 Seconds | Each Add'l 6 Seconds |
|------------------|-------------------|----------------------|
| - Month to Month | \$0.0100 | \$0.0100 |
| - 1 Year Term | \$0.0090 | \$0.0090 |
| - 2 Year Term | \$0.0085 | \$0.0085 |
| - 3 Year Term | \$0.0080 | \$0.0080 |

For billing purposes, the minimum call duration for all intraLATA and interLATA switched inbound usage is six (6) seconds. Usage is measured thereafter in six (6) second increments and rounded to the next higher six (6) second period.

The following monthly charge applies in addition to the above usage charges for each switched inbound toll free number:

| Term Agreement | Monthly Charge, per number | |
|------------------|----------------------------|--|
| - Month-to-Month | \$4.00 | |
| - 1 Year Term | \$4.00 | |
| - 2 Year Term | \$4.00 | |
| - 3 Year Term | \$4.00 | |

Additionally, the following per call surcharge shall apply to all switched inbound calls that originate from a payphone.

| Term Agreement | Payphone Use Charge | |
|------------------|---------------------|--|
| - Month-to-Month | \$1.25 | |
| - 1 Year Term | \$1.25 | |
| - 2 Year Term | \$1.25 | |
| - 3 Year Term | \$1.25 | |

c. <u>Dedicated Outbound IntraLATA and InterLATA</u>

Charges apply for all dedicated outbound intraLATA and interLATA usage as follows:

| Term Agreement | <u>Initial 6 Seconds</u> | Each Add'l 6 Seconds |
|------------------|--------------------------|----------------------|
| - Month to Month | \$0.0069 | \$0.0069 |
| - 1 Year Term | \$0.0060 | \$0.0060 |
| - 2 Year Term | \$0.0058 | \$0.0058 |
| - 3 Year Term | \$0.0056 | \$0.0056 |

For billing purposes, the minimum call duration for all intraLATA and interLATA dedicated outbound usage is six (6) seconds. Usage is measured thereafter in six (6) second increments and rounded to the next higher six (6) second period.

d. Dedicated Inbound IntraLATA and InterLATA

Charges apply for all dedicated inbound intraLATA and interLATA usage as follows:

| Term Agreement | <u>Initial 6 Seconds</u> | Each Add'l 6 Seconds |
|------------------|--------------------------|----------------------|
| - Month to Month | \$0.0069 | \$0.0069 |
| - 1 Year Term | \$0.0060 | \$0.0060 |
| - 2 Year Term | \$0.0058 | \$0.0058 |
| - 3 Year Term | \$0.0056 | \$0.0056 |

For billing purposes, the minimum call duration for all intraLATA and interLATA dedicated inbound usage is six (6) seconds. Usage is measured thereafter in six (6) second increments and rounded to the next higher six (6) second period.

The following monthly charge applies in addition to the above usage charges for each switched inbound toll free number:

| Term Agreement | Monthly Charge, per number |
|------------------|----------------------------|
| - Month-to-Month | \$4.00 |
| - 1 Year Term | \$4.00 |
| - 2 Year Term | \$4.00 |
| - 3 Year Term | \$4.00 |

Additionally, the following per call surcharge shall apply to all switched inbound calls that originate from a payphone.

| <u> 1erm Agreement</u> | Payphone Use Charge |
|------------------------|---------------------|
| - Month-to-Month | \$1.25 |
| - 1 Year Term | \$1.25 |
| - 2 Year Term | \$1.25 |
| - 3 Year Term | \$1.25 |

e. <u>Calling Card Service</u>

Charges apply for all intrastate calling card calls placed by CP-Plus Customers as follows:

| Term Agreement | Initial 30 Seconds | Each Add'l 6 Seconds |
|------------------|--------------------|----------------------|
| - Month-to-Month | \$0.1250 | \$0.0250 |
| - 1 Year Term | \$0.1250 | \$0.0250 |
| - 2 Year Term | \$0.1250 | \$0.0250 |
| - 3 Year Term | \$0.1250 | \$0.0250 |

For billing purposes, the minimum call duration for all intrastate calling card calls is thirty (30) seconds. Usage is measured thereafter in six (6) second increments and rounded to the next higher six (6) second period.

Additionally, the following per call surcharge shall apply to all intrastate calling card calls that originate from a payphone.

| Term Agreement | Payphone Use Charge | |
|------------------|---------------------|--|
| - Month-to-Month | \$1.25 | |
| - 1 Year Term | \$1.25 | |
| - 2 Year Term | \$1.25 | |
| - 3 Year Term | \$1.25 | |

\$0.000425

\$0.0046

GRANDFATHERED SERVICES FOR PREVIOUS CUSTOMERS OF CORECOMM NEWCO, INC.

| 17. | Switched Access Service | |
|--------|--|----------------------|
| 17.5 | Rate Regulations | |
| 17.5.2 | Cancellation of Access Service Order | |
| | B. Cancellation of Access Service Order | |
| | Cancellation Fee | \$25.00 |
| 17.5.3 | InterLATA Presubscription Charges | Nonrecurring Charges |
| | A. Per business or residence line, trunk, or port | |
| | - Initial or Additional line, trunk or port | \$5.00 |
| | B. Unauthorized InterLATA Presubscription Change Charge, per business or residence line or trunk | \$18.00 |
| 17.5.4 | IntraLATA Presubscription Change Charges | |
| | A. Per business or residence line, trunk, or port | |
| | Initial line, trunk or portAdditional line, trunk or port | \$5.00 \$1.50 |
| | B. Unauthorized IntraLATA Presubscription Change Charge, per business or residence line or trunk | \$18.00 |
| 17.6 | Rates and Charges | |
| 17.6.1 | Recurring Charges | |
| | A. Local Switching | |
| | Per Access Minute | \$0.01284 |
| | B. Switched Transport | |
| | Transport facilities, Per Access Minute, Per Mile | \$0.0030 |

C.

Transport termination, Per Access Minute

Toll Free Data Base Access Service (i.e. 800/888)

Per Query

- 18 <u>Carrier Common Line Access Service</u>
- 18.4 Rates and Charges
 - A. Ohio Bell Territory -- per access minute:

- Originating \$0.01284 - Terminating \$0.01284

19 <u>Public Telephone Service</u>

19.2 <u>Independent Payphone Provider Services</u>

19.2.1 <u>Customer-Owned, Coin Operated Telephones</u> (COCOT)

E. <u>Prices</u>

1. <u>Service Elements</u>

a. Network Access Line Rates

| | _ | Monthly Price | | |
|--|-------------|---------------|--------------------|--------------|
| | | Network | Central Office | Total |
| <u>Description</u> | Access Area | Access Line | <u>Termination</u> | Monthly Rate |
| Telephone Exchange Service Network Access Rates | | | | |
| Network Access, per Line | A | - | - | - |
| Network Access, per Line | В | \$19.15 | \$2.75 | \$21.90 |
| Network Access, per Line | C | \$21.55 | \$2.75 | \$24.30 |
| Network Access, per Line | D | \$24.55 | \$2.75 | \$27.30 |

1. <u>Service Elements</u> (cont.)

b. <u>Message Rate Service</u>

| <u>Description</u> | <u>Price</u> |
|--|--------------|
| Usage Package, per line, per month | \$7.40 |
| Add'l local calls, above the 73 call usage package, per call | \$0.10 |

c. <u>Service Establishment and Service Charges</u>

<u>Description</u> <u>Non-Recurring Charge</u>

Miscellaneous Service or Features, per individual COCOT line, per occasion

\$11.00

| <u>Description</u> | Non-Recurring Charge |
|--|----------------------|
| Service Ordering Charge, per location, per occasion | \$30.60 |
| Central Office Connection Charge, per termination | \$15.60 |
| Line Connection Charge, per termination | \$29.25 |
| Directory Listing Change Charge, per line, per change Telephone Number Change Charge | \$20.00 |
| - Initial Number Changed | \$30.00 |
| - Each Subsequent Change | \$50.00 |
| Assignment or Transfer of Service, per transfer Network Wiring Charge, billable premises work | \$20.00 |
| - First 15 minutes or fraction thereof | \$30.00 |
| - Each additional 15 minutes or fraction thereof | \$12.00 |

d. <u>Operator-Handled Services</u>

| <u>Description</u> | Non-Recurring Charge |
|--|----------------------|
| Automated Calling Card Station-to-Station Call, per call | \$0.50 |
| Customer-Dialed - Operator-Assisted - Calling Card | |
| Station-to-Station Call, per call | \$1.25 |
| Operator-Handled Station-to-Station, per call | \$1.10 |
| Operator-Handled Person-to-Person, per call | \$3.00 |
| Operator-Handled Third Number Billed, per call | \$1.50 |

1. <u>Service Elements</u> (cont.)

e. Message Toll Service

| MTS Rates | Day Rates Monday thru Fr but not including | riday, 8:00 AM to g 5:00 PM | Evening Rates Monday thru Fi but not including | riday, 5:00 PM to g 11:00 PM | not including 8:00 | ay, 11:00 PM to but |
|---------------|--|--------------------------------|--|---------------------------------|-----------------------|---------------------|
| Mileage | Initial Minute | Each Add'l Min. | Initial Minute | Each Add'l Min. | <u>Initial Minute</u> | Each Add'l Min. |
| Schedule A | | | | | | |
| 1-10 | \$0.45 | \$0.45 | \$0.45 | \$0.45 | \$0.45 | \$0.45 |
| 11-22 | \$0.45 | \$0.45 | \$0.45 | \$0.45 | \$0.45 | \$0.45 |
| 23-55 | \$0.45 | \$0.45 | \$0.45 | \$0.45 | \$0.45 | \$0.45 |
| 56-124 | \$0.45 | \$0.45 | \$0.45 | \$0.45 | \$0.45 | \$0.45 |
| 125-End | \$0.45 | \$0.45 | \$0.45 | \$0.45 | \$0.45 | \$0.45 |
| Schedule B | | | | | | |
| 1-10 | \$0.45 | \$0.45 | \$0.45 | \$0.45 | \$0.45 | \$0.45 |
| 11-22 | \$0.45 | \$0.45 | \$0.45 | \$0.45 | \$0.45 | \$0.45 |
| 23-55 | \$0.45 | \$0.45 | \$0.45 | \$0.45 | \$0.45 | \$0.45 |
| 56-124 | \$0.45 | \$0.45 | \$0.45 | \$0.45 | \$0.45 | \$0.45 |
| 125-End | \$0.45 | \$0.45 | \$0.45 | \$0.45 | \$0.45 | \$0.45 |
| Operator Assi | sted Calls | | | | | |
| 1-10 | \$0.45 | \$0.45 | \$0.45 | \$0.45 | \$0.45 | \$0.45 |
| 11-22 | \$0.45 | \$0.45 | \$0.45 | \$0.45 | \$0.45 | \$0.45 |
| 23-55 | \$0.45 | \$0.45 | \$0.45 | \$0.45 | \$0.45 | \$0.45 |
| 56-124 | \$0.45 | \$0.45 | \$0.45 | \$0.45 | \$0.45 | \$0.45 |
| 125-End | \$0.45 | \$0.45 | \$0.45 | \$0.45 | \$0.45 | \$0.45 |

f. Answer Supervision - Line Side

| <u>Description</u> | Non Recurring Charge ¹ | Monthly Price |
|--|-----------------------------------|---------------|
| Answer Supervision equipped with line side interface, per line | \$3.00 | \$1.95 |

2. Other Applicable Charges and Payments

In addition to Telephone Exchange Service Network Access Rates end user common line charges apply at the Multi-Line Business Customer - Individual Line or Trunk rate as provided in Section 4.3.4 of this tariff.

Effective June 1, 2021

¹ Charge does not apply when a Line Connection Charge is otherwise applicable

20. CoreConnect

20.3.1 CoreConnect Preferred ("CCP")

A. Monthly Rates*

1. Standard Rate Plan

| Term Commitment | Monthly Rate |
|----------------------|------------------|
| One Year, per line | Zone 1 - \$35.95 |
| | Zone 2 - \$37.95 |
| | Zone 3 - \$37.95 |
| Two Year, per line | \$21.95 |
| Three Year, per line | Zone 1 - \$33.95 |
| | Zone 2 - \$35.95 |
| | Zone 3 - \$40.95 |

2. Unlimited Usage Rate Plan

| Term Commitment | Monthly Rate |
|----------------------|--------------|
| One Year, per line | \$34.95 |
| Two Year, per line | \$33.95 |
| Three Year, per line | \$32.95 |

^{*} Customers purchasing service under this plan or package may qualify for the Multi Product Credit. The Multi Product Credit provides Customers purchasing qualifying commercial telephone line package(s) at the same service address as an On Net 1.5mb or greater Circuit, including, but not limited to, the Essential products (VoIP), with an \$8 credit on the Customer's commercial telephone line package(s). In the event of On Net circuit disconnection, the Multi Product Credit shall no longer be available.

B. <u>Local Usage Rates</u>

1. Per Call Local Usage Plan

| Term Commitment | Per Call Rate |
|-----------------|---------------|
| One Year | \$0.0600 |
| Two Year | \$0.0575 |
| Three Year | \$0.0550 |

2. <u>Unlimited Local Usage Plan</u>

| Term Commitment | Per Call Rate |
|-----------------|---------------|
| One Year | \$0.0600 |
| Two Year | \$0.0575 |
| Three Year | \$0.0550 |

C. <u>Dedicated Long Distance Usage Rates</u>

1. Outbound IntraLATA and InterLATA Rate Schedule

<u>IntraLATA</u>

| Term Agreement | Initial 6 Seconds | Each Add'l 6 Seconds |
|----------------|-------------------|----------------------|
| - 1 Year Term | \$0.0060 | \$0.0060 |
| - 2 Year Term | \$0.0575 | \$0.0575 |
| - 3 Year Term | \$0.0550 | \$0.0550 |

InterLATA

| Term Agreement | Initial 6 Seconds | Each Add'l 6 Seconds |
|----------------|-------------------|----------------------|
| - 1 Year Term | \$0.0060 | \$0.0060 |
| - 2 Year Term | \$0.0575 | \$0.0575 |
| - 3 Year Term | \$0.0550 | \$0.0550 |

2. <u>Inbound IntraLATA and InterLATA Rate Schedule</u>

IntraLATA

| Term Agreement | <u>Initial 6 Seconds</u> | Each Add'l 6 Seconds |
|----------------|--------------------------|----------------------|
| - 1 Year Term | \$0.0060 | \$0.0060 |
| - 2 Year Term | \$0.0575 | \$0.0575 |
| - 3 Year Term | \$0.0550 | \$0.0550 |

InterLATA

| <u>Term Agreement</u> | Initial 6 Seconds | Each Add'l 6 Seconds |
|-----------------------|-------------------|----------------------|
| - 1 Year Term | \$0.0060 | \$0.0060 |
| - 2 Year Term | \$0.0575 | \$0.0575 |
| - 3 Year Term | \$0.0550 | \$0.0550 |

The following monthly charge applies in addition to the above usage charges for each dedicated inbound toll free number:

Monthly

| - 1 Year Term | \$4.00 |
|---------------|--------|
| - 2 Year Term | \$4.00 |
| - 3 Year Term | \$4.00 |

Additionally, the following per call surcharge shall apply to all dedicated inbound calls that originate from a payphone.

Payphone Use Charge

| - | 1 Year Term | \$1.25 |
|---|-------------|--------|
| - | 2 Year Term | \$1.25 |
| - | 3 Year Term | \$1.25 |

D. Optional Features

3. <u>Rate Schedule</u>

| Automatic Callback | |
|------------------------------------|--------|
| - Monthly, Per Line | \$7.25 |
| - Per Use | \$0.97 |
| Call Forwarding - Variable | \$9.35 |
| Call Forwarding with Remote Access | \$4.00 |
| Call Screening | \$4.00 |
| Call Trace, Per Use | \$3.50 |
| Call Transfer | \$4.00 |
| Call Waiting | \$5.00 |
| Caller ID with Name | |
| - Monthly, Per Line | \$9.00 |
| - Per Use | \$0.75 |
| Easy Call | \$1.50 |
| Multi-Ring Service | |
| - First line | \$4.00 |
| - Second line | \$3.95 |
| Repeat Dialing | |
| - Monthly, Per Line | \$4.00 |
| - Per Use | \$1.50 |
| Speed Call | |
| - 8 Numbers | \$4.00 |
| - 30 Numbers | \$4.00 |
| Temporary Intercept | |
| Three Way Calling | |
| - Monthly, Per Line | \$4.00 |
| - Per Use | \$1.50 |
| Wait and See | \$0.00 |
| 900 / 976 Blocking | \$0.00 |

E. <u>Calling Card Service</u>

Charges apply for all intrastate calling card calls placed by CCP Customers as follows:

| | Initial 30 Seconds | Each Add'l 6 Seconds |
|---------------|--------------------|----------------------|
| - 1 Year Term | \$0.125 | \$0.025 |
| - 2 Year Term | \$0.125 | \$0.025 |
| - 3 Year Term | \$0.125 | \$0.025 |

Additionally, the following per call surcharge shall apply to all intrastate calling card calls that originate from a payphone.

Payphone Use Charge

| - 1 Year Term | \$1.25 |
|---------------|--------|
| - 2 Year Term | \$1.25 |
| - 3 Year Term | \$1.25 |

F. <u>Miscellaneous Charges</u>

1. Project Account Codes ("PAC")

| | Non-Recurring Set-Up Charge | Monthly Recurring Charge | Change Charge/Order |
|--------------|--------------------------------|-----------------------------|---------------------|
| Non-Verified | N/A | \$5.00 | \$5.00 |
| Verified | N/A | \$5.00 | \$5.00 |
| Custom | \$50.00 | \$5.00 | \$5.00 |

20.3.2 CoreConnect Preferred Plus ("CCPP")

A. Monthly Rates*

1. Standard Rate Plan

| Term Commitment | Monthly Rate |
|----------------------|--------------|
| One Year, per line | \$30.95 |
| Two Year, per line | \$29.95 |
| Three Year, per line | \$28.95 |

2. Unlimited Usage Rate Plan

| <u>Term Commitment</u> | Monthly Rate |
|------------------------|--------------|
| One Year, per line | \$34.95 |
| Two Year, per line | \$33.95 |
| Three Year, per line | \$32.95 |

^{*} Customers purchasing service under this plan or package may qualify for the Multi Product Credit. The Multi Product Credit provides Customers purchasing qualifying commercial telephone line package(s) at the same service address as an On Net 1.5mb or greater Circuit, including, but not limited to, the Essential products (VoIP), with an \$8 credit on the Customer's commercial telephone line package(s). In the event of On Net circuit disconnection, the Multi Product Credit shall no longer be available.

B. <u>Local Usage Rates</u>

1. Per Call Local Usage Plan

| Term Commitment | Per Call Rate |
|-----------------|---------------|
| One Year | \$0.0600 |
| Two Year | \$0.0575 |
| Three Year | \$0.0550 |

2. <u>Unlimited Local Usage Plan</u>

| Term Commitment | Per Call Rate |
|-----------------|---------------|
| One Year | \$0.0600 |
| Two Year | \$0.0575 |
| Three Year | \$0.0550 |

C. <u>Dedicated Long Distance Usage Rates</u>

1. Outbound IntraLATA and InterLATA Rate Schedule

IntraLATA

Charges apply for all dedicated outbound intraLATA usage as follows:

| Term Agreement | Initial 6 Seconds | Each Add'l 6 Seconds |
|----------------|-------------------|----------------------|
| - 1 Year Term | \$0.0060 | \$0.0060 |
| - 2 Year Term | \$0.0575 | \$0.0575 |
| - 3 Year Term | \$0.0550 | \$0.0550 |

InterLATA

Charges apply for all dedicated outbound interLATA usage as follows:

| Term Agreement | <u>Initial 6 Seconds</u> | Each Add'l 6 Seconds |
|----------------|--------------------------|----------------------|
| - 1 Year Term | \$0.0060 | \$0.0060 |
| - 2 Year Term | \$0.0575 | \$0.0575 |
| - 3 Year Term | \$0.0550 | \$0.0550 |

2. <u>Inbound IntraLATA and InterLATA Rate Schedule</u>

IntraLATA

Charges apply for all dedicated inbound intraLATA usage as follows:

| Term Agreement | Initial 6 Seconds | Each Add'l 6 Seconds |
|----------------|-------------------|----------------------|
| - 1 Year Term | \$0.0060 | \$0.0060 |
| - 2 Year Term | \$0.0575 | \$0.0575 |
| - 3 Year Term | \$0.0550 | \$0.0550 |

InterLATA

Charges apply for all dedicated inbound interLATA usage as follows:

| Term Agreement | Initial 6 Seconds | Each Add'l 6 Seconds |
|----------------|-------------------|----------------------|
| - 1 Year Term | \$0.0060 | \$0.0060 |
| - 2 Year Term | \$0.0575 | \$0.0575 |
| - 3 Year Term | \$0.0550 | \$0.0550 |

The following monthly charge applies in addition to the above usage charges for each dedicated inbound toll free number:

| | Monthly |
|---------------|---------|
| - 1 Year Term | \$4.00 |
| - 2 Year Term | \$4.00 |
| - 3 Year Term | \$4.00 |

Additionally, the following per call surcharge shall apply to all dedicated inbound calls that originate from a payphone.

| | Payphone Use Charge |
|---|----------------------------|
| 1 Year Term2 Year Term3 Year Term | \$1.25 \$1.25 \$1.25 |
| | |

D. <u>Optional Features</u>

3. <u>Rate Schedule</u>

| Automatic Callback | |
|------------------------------------|--------|
| - Monthly, Per Line | \$7.25 |
| - Per Use | \$0.97 |
| Call Forwarding - Variable | \$9.35 |
| Call Forwarding with Remote Access | \$4.00 |
| Call Screening | \$4.00 |
| Call Trace, Per Use | \$3.50 |
| Call Transfer | \$4.00 |
| Call Waiting | \$5.00 |
| Caller ID with Name | |
| - Monthly, Per Line | \$9.00 |
| - Per Use | \$0.75 |
| Easy Call | \$1.50 |
| Multi-Ring Service | |
| - First line | \$4.00 |
| - Second line | \$3.95 |
| Repeat Dialing | |
| - Monthly, Per Line | \$4.00 |
| - Per Use | \$1.50 |
| Speed Call | |
| - 8 Numbers | \$4.00 |
| - 30 Numbers | \$4.00 |
| Temporary Intercept | |
| Three Way Calling | |
| - Monthly, Per Line | \$4.00 |
| - Per Use | \$1.50 |
| Wait and See | \$0.00 |
| 900 / 976 Blocking | \$0.00 |
| | |

E. Calling Card Service

Charges apply for all intrastate calling card calls placed by CCPP Customers as follows:

| | Initial 30 Seconds | Each Add'l 6 Seconds |
|---------------|--------------------|----------------------|
| - 1 Year Term | \$0.125 | \$0.025 |
| - 2 Year Term | \$0.125 | \$0.025 |
| - 3 Year Term | \$0.125 | \$0.025 |

Additionally, the following per call surcharge shall apply to all intrastate calling card calls that originate from a payphone.

Payphone Use Charge

| - | 1 Year Term | \$1.25 |
|---|-------------|--------|
| - | 2 Year Term | \$1.25 |
| - | 3 Year Term | \$1.25 |

F. <u>Miscellaneous Charges</u>

1. Project Account Codes ("PAC")

| Code | Non-Recurring Set-Up Charge | Monthly Recurring Charge | Change Charge/Order |
|--------------|--------------------------------|--------------------------|---------------------|
| Non-Verified | N/A | \$5.00 | \$5.00 |
| Verified | N/A | \$5.00 | \$5.00 |
| Custom | \$50.00 | \$5.00 | \$5.00 |

20.3.3 CoreConnect Trunks

Customers subscribing to CoreConnect Trunk service receive connectivity to the public switched network through PBX and DID trunks bundled with local and long distance service with a choice of one, two or three year terms. The trunk rate includes Caller ID, Caller ID with Name, and Call Forward Trunk Group. End User Common Line¹, 911 and Local Number Portability and remaining charges specified in Section 4.3.4 of this tariff are in addition to the monthly trunk rate.

20.3.3.1 Trunk Charges

| Term Commitment | Monthly PBX Trunk Charge | Monthly DID Trunk Rate | Non-Recurring Charge (Installation) |
|-----------------|-----------------------------|---------------------------|-------------------------------------|
| 1 Year | \$19.00 | \$27.00 | \$899.00 |
| 2 Years | \$18.00 | \$26.00 | \$899.00 |
| 3 Years | \$17.00 | \$25.00 | \$899.00 |

20.3.3.2 Features

The following features can be added for DID Trunk service.

| Term Commitment | Station Numbers (block of 20) |
|-----------------|-------------------------------|
| 1 Year | \$3.00 |
| 2 Years | \$3.00 |
| 3 Years | \$3.00 |

20.3.3.3 Local Usage

CoreConnect Trunk customers may choose between per-call and unlimited local usage rate plans according to the terms and conditions of local usage under CoreConnect Preferred and CoreConnect Preferred Plus, Sections 20.3.1.B and 20.3.2.B, respectively.

20.3.3.4 Long Distance Service

CoreConnect Trunk customers may choose long distance service according to the terms and conditions of long distance under CoreConnect Preferred and CoreConnect Preferred Plus, Sections 20.3.1.C and 20.3.2.C, respectively.

¹ The End User Common Line Charge can be located in CoreComm's FCC tariff.

20.3.4 CoreConnect PRI

CoreConnect PRI offers Customers the option of designing service packages to fit their specific communications needs. CoreConnect PRI services will be provided to the Customer by the Company via an ISDN access line. Certain monthly rates plus usage charges will apply as provided herein depending upon the term commitment made by the Customer and the services to which the Customer subscribes. The Company's central office trunk port will be connected to the Customer's location via a DS-1 (1.544 Mbps) facility. CoreConnect PRI's standard configuration will be 23 B channels (64 Kbps) for user information, plus 1 D channel (64 Kbps) for signaling and control functions which will remain idle when not in use. The Company will require a 23-B and 1-D channel configuration where 47 or more channels are controlled by a single primary D channel. CoreConnect PRI is available only where facilities permit to Business Customers who subscribe for a term of one, two, three, or five years.

20.3.4.1 <u>Line Charges</u>

| Term Commitment | Monthly PRI Charge | Back Up D-Channel | Non-recurring |
|-----------------|--------------------|-------------------|----------------|
| | | | Charge |
| | | | (Installation) |
| 1 Year | \$525.00 | \$150.00 | \$899.00 |
| 2 Years | \$499.00 | \$150.00 | \$899.00 |
| 3 Years | \$488.00 | \$150.00 | \$899.00 |
| 5 Years | \$473.00 | \$150.00 | \$899.00 |

20.3.4.2 <u>Feature Charges</u>

CoreConnect PRI customers can add the following features to their service.

| Term Commitment | DID Station Numbers | Call By Call Feature | Caller ID |
|-----------------|----------------------------|----------------------|-----------|
| | (block of 20) | | |
| 1Year | \$5.60 | \$75.00 | \$75.00 |
| 2 Years | \$5.60 | \$75.00 | \$75.00 |
| 3 Years | \$5.60 | \$75.00 | \$75.00 |
| 5 Years | \$5.60 | \$75.00 | \$75.00 |

20.3.4 CoreConnect PRI (continued)

20.3.4.3 <u>Local Usage Charges</u>

Local usage is provided as a measured service on a per-minute basis. Unless otherwise specified, for billing purposes, the minimum call duration is thirty (30) seconds for domestic calls and sixty (60) seconds for international calls. In addition, unless otherwise specified, usage is measured thereafter in six (6) second increments and rounded to the next higher six (6) second period. Fractional amounts are rounded up to the next highest whole cent.

| Term Commitment | Per-Minute Charge | |
|-----------------|--------------------|----------------------|
| | Initial 30 Seconds | Each Add'l 6 Seconds |
| 1 Year | \$0.005 | \$0.001 |
| 2 Years | \$0.005 | \$0.001 |
| 3 Years | \$0.005 | \$0.001 |
| 5 Years | \$0.005 | \$0.001 |

20.3.4.4 Long Distance Service

Customers subscribing to CoreConnect PRI service can receive intraLATA, interLATA and 8XX service at the following per minute rates. Unless otherwise specified, for billing purposes, the minimum call duration is thirty (30) seconds for domestic calls and sixty (60) seconds for international calls. In addition, unless otherwise specified, usage is measured thereafter in six (6) second increments and rounded to the next higher six (6) second period. Fractional amounts are rounded up to the next highest whole cent.

| | IntraLATA / InterLATA / 8XX | |
|-----------------|-----------------------------|----------------------|
| Term Commitment | Initial 30 Seconds | Each Add'l 6 Seconds |
| 1 Year | \$0.02500 | \$0.00500 |
| 2 Years | \$0.02500 | \$0.00500 |
| 3 Years | \$0.02500 | \$0.00500 |
| 5 Years | \$0.02500 | \$0.00500 |

PRICE LIST

| 21. | Smart Link |
|-----|------------|
| | |

21.2 <u>Smart Link Miscellaneous Charges</u>

21.2.2 <u>CSR/Conversion Charges</u>

<u>Charge Per CSR</u> \$20.00

21.2.3 <u>Installation Charges</u>

<u>Charge Per Installation</u> \$33.50

21.2.4 <u>Service Order Charge</u>

<u>Charge Per Order</u> \$21.85

21.2.5 <u>Service Change Charge</u>

Charge Per Change

Message to Measured \$41.55 Measured to Message \$41.55

21.3 Smart Link Local Services

21.3.2 Smart Link Local Services -- Line and Usage Charges

A. Access Line Charges

The following line charges apply to Smart Link Customers in lieu of those access line charges provided for in Section 4.3.3 of this tariff. All such business line charges indicated below include both the charge for the access line, line hunting service and any three (3) features in Section 21.3.3 of this tariff. Additional features are available at the rates contained within Section 21.3.3.

| Term Commitment | Monthly Charge | Monthly Charge |
|-----------------|--------------------|------------------|
| | With Long Distance | No Long Distance |
| Month-to-Month | Zone | Zone |
| | 1 - \$124.95 | 1 - \$124.95 |
| | 2 - \$124.95 | 2 - \$124.95 |
| | 3 – \$12.95 | 3 - \$124.95 |
| 1 Year | Zone | \$124.95 |
| | 1 - \$124.95 | |
| | 2 - \$124.95 | |
| | 3 – \$124.95 | |
| 2 Years* | Zone | \$124.95 |
| | 1 - \$124.95 | |
| | 2 - \$124.95 | |
| | 3 - \$124.95 | |
| 3 Years* | Zone | \$124.95 |
| | 1 - \$124.95 | , |
| | 2 - \$124.95 | |
| | 3 - \$124.95 | |

Business line charges do not include the End User Common Line, 911 and Local Number Portability surcharges in Section 4.3.4 of this tariff. The charges provided for in Section 4.3.4 of this tariff apply to each line in addition to the monthly rates provided above.

^{*} Customers purchasing service under this plan or package may qualify for the Multi Product Credit. The Multi Product Credit provides Customers purchasing qualifying commercial telephone line package(s) at the same service address as an On Net 1.5mb or greater Circuit, including, but not limited to, the Essential products (VoIP), with an \$8 credit on the Customer's commercial telephone line package(s). In the event of On Net circuit disconnection, the Multi Product Credit shall no longer be available.

21.3.2 Smart Link Local Services -- Line and Usage Charges

B. <u>Local Usage Charges</u>

1. <u>Message Rate Service</u>

| Term Commitment | Rate Per Call |
|-----------------|---------------|
| Month-to-Month | \$0.0800 |
| 1 Year | \$0.0750 |
| 2 Years | \$0.0725 |
| 3 Years | \$0.0700 |

2. Measured Rate Service

The following per minute rates apply to those Customers subscribing to the Smart Link Measured Rate Service. If the customer chooses to subscribe to the Company's local service only, then the per-minute rates under "With Local Service Only" below apply. Unless otherwise specified, for billing purposes, the minimum call duration is thirty (30) seconds. In addition, unless otherwise specified, usage is measured thereafter in six (6) second increments and rounded to the next higher six (6) second period. Fractional amounts are rounded up to the next highest whole cent.

| Term Commitment | With Long | Distance | With Local S | ervice Only |
|-----------------|------------|-----------|--------------|-------------|
| | Initial 30 | Add'l. 6 | Initial 30 | Add'l. 6 |
| | Seconds | Seconds | Seconds | Seconds |
| 1 Year | \$0.00875 | \$0.00175 | \$0.01500 | \$0.00300 |
| 2 Years | \$0.00750 | \$0.00150 | \$0.01375 | \$0.00275 |
| 3 Years | \$0.00625 | \$0.00125 | \$0.01250 | \$0.00250 |

21.3.3 Smart Link Local Services -- Service Enhancement Features

A. Optional Features

Customers may additionally subscribe to the following features. Charges apply on a per line, per month basis unless otherwise indicated:

| | Rate |
|------------------------|------------------|
| 900 Call Blocking | \$0.00 |
| 976 Call Blocking | \$0.00 |
| Alternate Answering | \$0.75 |
| Automatic Callback | \$7.25 |
| per use | \$0.97 |
| Busy Line Transfer | \$0.75 |
| Call Forwarding | \$6.50 |
| Caller ID | \$7.00 |
| Caller ID With Name | \$10.50 |
| Call Trace, per use | \$3.50 |
| Call Waiting | \$5.50 |
| Line Hunting Service | \$0.00 |
| Multi-Ring Service | |
| 1 st Number | \$5.50 |
| 2 nd Number | \$5.00 |
| Remote Call Forwarding | Zone 1 - \$34.95 |
| | Zone 2 - \$34.95 |
| | Zone 3 - \$34.95 |
| Repeat Dialing | \$5.50 |
| Speed Calling - 8 | \$6.00 |
| Speed Calling - 30 | \$6.10 |
| Three-Way Calling | \$6.50 |
| per use | \$1.50 |

21.4 Smart Link Long Distance Service

Unless otherwise specified, for billing purposes, fractional amounts are rounded up to the next highest whole cent.

21.4.2 Switched Services Rates

| Term Commitment | Per Minute |
|-----------------|------------|
| Month-to-Month | \$0.0600 |
| 1 Year | \$0.0600 |
| 2 Years | \$0.0575 |
| 3 Years | \$0.0550 |

21.6 Smart Link Centrex

21.6.1 <u>Line Rates*</u>

Smart Link Centrex provides subscribers access to the public switched network with specified Centrex features.

| Term Commitment | Monthly Line Charge, Per Line | |
|-----------------|-------------------------------|--|
| 1 Year | \$124.95 | |
| 2 Years | \$124.95 | |
| 3 Years | \$124.95 | |

Smart Link Centrex charges do not include the End User Common Line, 911 and Local Number Portability surcharges in Section 4.3.4 of this tariff. The charges provided for in Section 4.3.4 of this tariff apply to each line in addition to the monthly rates provided above.

^{*} Customers purchasing service under this plan or package may qualify for the Multi Product Credit. The Multi Product Credit provides Customers purchasing qualifying commercial telephone line package(s) at the same service address as an On Net 1.5mb or greater Circuit, including, but not limited to, the Essential products (VoIP), with an \$8 credit on the Customer's commercial telephone line package(s). In the event of On Net circuit disconnection, the Multi Product Credit shall no longer be available.

21.6.2. Local Usage Charges

The following local usage charges apply to Smart Link Centrex customers. Customers may request billing on a per call basis (message) or a per minute basis (measured) as provided in Sections 21.6.2.A and 21.6.2.B following, respectively.

A. Message Rate Service

The following per call rates apply to those Customers subscribing to the Smart Link Message Rate Service:

| Rate Per Call |
|---------------|
| \$0.0750 |
| \$0.0725 |
| \$0.0700 |
| |

B. Measured Rate Service

The following per minute rates apply to those Customers subscribing to the Smart Link Measured Rate Service. If the customer chooses to subscribe to the Company's local service only, then the per-minute rates under "With Local Service Only" below apply. Unless otherwise specified, for billing purposes, the minimum call duration is thirty (30) seconds. In addition, unless otherwise specified, usage is measured thereafter in six (6) second increments and rounded to the next higher six (6) second period. Fractional amounts are rounded up to the next highest whole cent.

| Term Commitment | With Long | Distance | With Local S | ervice Only |
|-----------------|------------|-----------|--------------|-------------|
| | Initial 30 | Add'l. 6 | Initial 30 | Add'l. 6 |
| | Seconds | Seconds | Seconds | Seconds |
| 1 Year | \$0.00875 | \$0.00175 | \$0.01500 | \$0.00300 |
| 2 Years | \$0.00750 | \$0.00150 | \$0.01375 | \$0.00275 |
| 3 Years | \$0.00625 | \$0.00125 | \$0.01250 | \$0.00250 |

C. <u>Smart Link Centrex Switched Long Distance Service</u>

Smart Link Centrex customers may subscribe to the Company's switched long distance service under the rates terms and conditions outlined in Section 21.4 of this tariff.

21.6.3 <u>Smart Link Centrex Local Services -- Service Enhancement Features</u>

Customers may additionally subscribe to the following features. Charges apply on a per line, per month basis unless otherwise indicated:

| | Rate |
|--------------------------------------|--------|
| Automatic Call back - Per line | \$7.25 |
| Call Diverting | \$0.00 |
| Call Forward - Selective | \$0.00 |
| Call Forward Busy | \$0.00 |
| Call Forward No Answer | \$0.00 |
| Call Forward -Variable | \$9.35 |
| Call Hold | \$0.00 |
| Consultation Hold | \$0.00 |
| Call Park | \$0.00 |
| Call Pick Up | \$0.00 |
| Call Screening | \$0.00 |
| Call Trace | \$0.00 |
| Call Transfer-Deluxe | \$0.00 |
| Call Waiting | \$0.00 |
| Caller ID | \$0.00 |
| Caller ID w/name | \$7.25 |
| Conference Calling | \$0.00 |
| Distinctive Ring | \$0.00 |
| Intercom Calling (3-5 Digit Dialing) | \$0.00 |
| Message Waiting Indicator (Audible) | \$0.00 |
| Multi Ring- First Line | \$9.35 |
| Multi Ring- Addtl Line | \$7.49 |
| Music on Hold | \$0.00 |
| Repeat Dialing | \$0.00 |
| Return Call | \$0.00 |
| Speed Calling Short | \$0.00 |
| Speed Calling (up to 30) | \$0.00 |

15.8 Smart Link Trunks

15.8.1 General

Smart Link Trunk service bundles the Company's trunking service with local and long distance for business customers who want connectivity to the public switched network through PBX or DID trunks. The Company provides this connectivity for one, two or three-year terms with specified features such as blocks of 20 station numbers for DID service and Call Forward Trunk Group.

15.8.2 Trunk Charges

| Term Commitment | Monthly PBX Trunk Charge | Monthly DID Trunk Rate |
|-----------------|--------------------------|------------------------|
| 1 Year | \$19.00 | \$27.00 |
| 2 Years | \$18.00 | \$26.00 |
| 3 Years | \$17.00 | \$25.00 |

Trunk charges do not include the End User Common Line, 911 and Local Number Portability surcharges in Section 4.3.4 of this tariff. The charges provided for in Section 4.3.4 of this tariff apply to each line in addition to the monthly rates provided above.

15.8.3 Features

The following features can be added to Smart Link Trunk service.

| | <u>Rate</u> |
|-------------------------------|-------------|
| Station Numbers (block of 20) | \$3.00 |
| Call Forward Trunk Group | \$0.00 |
| Call Forwarding Variable | \$9.35 |
| Caller ID | \$7.00 |
| Caller ID with Name | \$10.50 |

21.7 <u>Smart Link Trunk Service (continued)</u>

21.7.4 Local Usage Charges

The following local usage charges apply to Smart Link Trunk customers. Customers may request billing on a per call basis (message) or a per minute basis (measured) as provided in Sections 21.7.4.A and 21.7.4.B following, respectively.

A. Message Rate Service

The following per call rates apply to those Customers subscribing to the Smart Link Message Rate Service:

| Term Commitment | Rate Per Call |
|-----------------|---------------|
| 1 Year | \$0.0750 |
| 2 Years | \$0.0725 |
| 3 Years | \$0.0700 |

B. Measured Rate Service

The following per minute rates apply to those Customers subscribing to the Smart Link Measured Rate Service. If the customer chooses to subscribe to the Company's local service only, then the per-minute rates under "With Local Service Only" below apply. Unless otherwise specified, for billing purposes, the minimum call duration is thirty (30) seconds. In addition, unless otherwise specified, usage is measured thereafter in six (6) second increments and rounded to the next higher six (6) second period. Fractional amounts are rounded up to the next highest whole cent.

| Term Commitment | With Long Distance | | With Local Service Only | |
|-----------------|--------------------|-----------|-------------------------|-----------|
| | Initial 30 | Add'l. 6 | Initial 30 | Add'l. 6 |
| | Seconds | Seconds | Seconds | Seconds |
| 1 Year | \$0.00875 | \$0.00175 | \$0.01500 | \$0.00300 |
| 2 Years | \$0.00750 | \$0.00150 | \$0.01375 | \$0.00275 |
| 3 Years | \$0.00625 | \$0.00125 | \$0.01250 | \$0.00250 |

C. Smart Link Trunk Switched Long Distance Service

Smart Link Trunk customers may subscribe to the Company's switched long distance service under the rates terms and conditions outlined in Section 21.4 of this tariff.

TARIFFING NOTE: Promotional pages are actually addenda to the underlying price list page on which the standard charge is provided for

PROMOTIONAL OFFERING

Current residential Internet-only customers of the Company who add local exchange service and switched long distance service to their account by December 31, 2002 will receive a free Caller ID phone.¹ The customer must be a new residential local exchange customer who has existing local exchange service with The Ohio Bell Telephone Company (SBC Ameritech Ohio) and who subscribes to the Company's *Unlimited* residential bundled service plan.² The free phone will be provided within 30 days of activation of local service. There is a limit of one Caller ID phone per household. This offer is limited to areas where bundled residential service is available.

Additional Federal Access charges, Number Portability and 911 Service Fees will apply, plus any applicable state and local taxes. Customer may not retain current phone number when number portability is not available. Optional extended local calling not included with unlimited local calls. Customer must mention promotion to be eligible. Offer not valid with any other promotional offers, offer expires December 31, 2002. Customer must be 18 years of age or older.

-

¹ The PUCO does not regulate the Company's internet services.

²See Section 4.4.3.8.3 in this price sheet document.

PROMOTIONAL OFFERING

This Winback promotional offer is for Ohio residential local exchange service customers that have left CoreComm Newco, Inc. ("the Company") for another local exchange service carrier. Customers who come back to the Company and sign up for its Local Plus 5 or 2-Line Premium Connect bundled residential local exchange services, ³ will receive two (2) free months of local exchange service and 250 minutes of free domestic long distance service.

The two (2) free months of local exchange service will be given as credits in the 3rd and 9th month of service, as long as the customer's account is in good standing. The Customer must sign up for the Company's Local Plus 5 or 2-Line Premium Connect service. The Customer must respond to a marketing promotional contact (direct mail or outbound telemarketing call) to be eligible for the offer. The 250 minutes of free long distance is for direct dial domestic calls only. Unused minutes will carry over to the next month until all minutes are used.

The Customer must not have had service discontinued for non-payment or have any past due bills owed to the Company. If the Customer discontinues the package, they will forfeit any future credits.

This offer expires July 31, 2003.

-

See Sections 4.4.3.8.2 and 4.4.3.8.4, respectively, in this price sheet document.

PROMOTIONAL OFFERING

IntraLATA and InterLATA Presubscription Charges will be waived for Customers who pre-subscribe to the Company's intraLATA or intraLATA service during the period of March 9, 2001 through May 31, 2001.

PROMOTIONAL OFFERING

Line Establishment Charges for the conversion of existing local lines from an incumbent local exchange carrier will be waived for Residential Line Customers who subscribe such lines to the Company's local exchange service during the period of March 9, 2001 through May 31, 2001.

PROMOTIONAL OFFERING

This promotional offer is for new Ohio residential local exchange service customers that subscribe to the Company's following bundled residential local exchange service plans: Local Plus 5, Unlimited or 2 Line Premium Connect located at Sections 4.4.3.8.2, 4.4.3.8.3 and 4.4.3.8.4, respectively, within this tariff.

This offer subject to service availability and available only for customers who are age 18 or over when ordering Local Plus 5, Unlimited or 2-Line Premium service between May 1, 2003 and July 31, 2003 and who maintain service for at least 90 days following such order. For customers who order Local Plus 5 service, a credit of \$55.90 will be applied to Customer's next bill after three months of service in good standing. For customers who order the Unlimited or 2-Line Premium Connect package, a \$60 credit will be applied to the Customer's next bill after three months of service in good standing. Offer is not valid with any other promotional offers. Pricing, terms and product subject to change without notice. This offer expires July 31, 2003.

PROMOTIONAL OFFERING

This promotional offer increases the number of features included in the monthly access line rate from two (2) to five (5) for Customers subscribing to two residential local exchange service plans: (1) Unlimited and (2) 2-Line Premium Connect located in Sections 4.4.3.8.3 and 4.4.3.8.4, respectively, within this tariff. There is no change in price to either service plan.

PROMOTIONAL OFFERING

In lieu of the standard price list rate, a Telephone Number Change charge of \$30.00 shall be assessed to Customers who request such a change in service during the period of July 18, 2000 through October 14, 2000 ("Eligible Customers"). Eligible Customers shall be limited to one such change in telephone number per Customer at the \$30.00 promotional rate. Standard price list rates shall thereafter apply for all for telephone number changes on the Eligible Customer's account.

eCHAT SERVICE PROMOTIONAL OFFERING

Service Elements

A. Monthly Rate

The eChat Service Promotion offers an access line, unlimited local calling, Call Waiting, Caller ID with Name, Wait and See, and a personal 800 number for a single monthly rate to Residential Customers who subscribe to the Company's eChat Service during the Promotional Period ("Eligible Customers"). Eligible Customers may select an eChat second option which offers only an access line and unlimited local calling for a flat monthly rate.

| | Monthly Rate | | |
|--|-------------------------|-----------------------|--|
| | | Each Add'l eChat | |
| | First eChat Access Line | Access Line, Per Line | |
| - Option 1 (includes unlimited local calling, Call | | Zone 1 - \$218.70 | |
| Waiting, Caller ID with Name, Wait and See, | | Zone 2 - \$23.70 | |
| and personal 800 number) | \$28.70 | Zone 2 - \$28.70 | |
| - Option 2 (includes only unlimited local calling) | \$21.95 | \$16.70 | |

B. 800 Switched Inbound Usage

For billing purposes, the minimum call duration for switched inbound usage on the Customer's personal 800 number shall be sixty (60) seconds. Usage is measured thereafter in six (6) second increments and rounded to the next higher six (6) period.

| | Initial 60 Seconds | Each Add'l 6 Seconds |
|------------------------------|--------------------|----------------------|
| - Switched Inbound 800 Usage | \$0.150 | \$0.015 |

Additionally, a \$1.25 per call surcharge shall apply to all calls to the Customer's personal 800 number that originate from a payphone.

Promotional Period

The promotional period will be from February 15, 2000 through May 15, 2000.

Conditions and Limitations

- A. Charges provided for in Section 4.3.4 of this Tariff apply to each access line in addition to the monthly rate provided above.
- B. Service Ordering Charges apply as provided in Section 4.1 of this Tariff.
- C. Customers may subscribe to additional optional exchange access service enhancement features at the rates provided in Section 4.5.3 of this Tariff. Rates for pay per use/per activation optional exchange access service enhancement features apply at all times as provided in Section 4.5.3 of this Tariff.
- D. Rates for Local Calling Plus apply as provided in Section 4.4.3.2 of this Tariff.
- E. Operator assistance surcharges apply as provided in Section 4.6.2 of this Tariff.

eCHAT PLUS SERVICE PROMOTIONAL OFFERING

Service Elements

A. Monthly Rate

The eChat Plus Service Promotion offers an access line, unlimited local calling, Call Waiting, Caller ID with Name, Wait and See and a personal 800 number for a single rate to Residential Customers who subscribe to the Company's eChat Plus Service during the Promotional Period ("Eligible Customers"). Eligible Customers may select an eChat Plus second option which offers only an access line and unlimited local calling for a flat monthly rate.

| | Monthly Rate | |
|--|------------------|-----------------------|
| | First eChat | Each Add'l eChat Plus |
| | Plus Access Line | Access Line, Per Line |
| - Option 1 (includes unlimited local calling, Call | | Zone 1 - \$18.70 |
| Waiting, Caller ID with Name, Wait and See, | | Zone 2 - \$20.70 |
| and personal 800 number) | \$25.70 | Zone 2 - \$25.70 |
| - Option 2 (includes only unlimited local calling) | \$18.95 | \$13.70 |

B. 800 Switched Inbound Usage

For billing purposes, the minimum call duration for switched inbound usage on the Eligible Customer's personal 800 number shall be sixty (60) seconds. Usage is measured thereafter in six (6) second increments and rounded to the next higher six (6) period.

| | Initial 60 Seconds | Each Add'l 6 Seconds | |
|------------------------------|--------------------|----------------------|--|
| - Switched Inbound 800 Usage | \$0.150 | \$0.015 | |

Additionally, a \$1.25 per call surcharge shall apply to all calls to the Eligible Customer's personal 800 number that originate from a payphone.

C. Switched Outbound Long Distance Usage

Eligible Customers are entitled to the following rates for all switched outbound intraLATA and interLATA long distance usage:

| | Initial 60 Seconds | Each Add'l 6 Seconds | |
|---------------------------|--------------------|----------------------|--|
| - IntraLATA Long Distance | \$0.100 | \$0.010 | |
| - InterLATA Long Distance | \$0.100 | \$0.010 | |

For billing purposes, the minimum call duration for switched outbound intraLATA and interLATA long distance usage shall be sixty (60) seconds. Usage is measured thereafter in six (6) second increments and rounded to the next higher six (6) period.

eCHAT PLUS PROMOTIONAL OFFERING (continued)

D. <u>Calling Card Service</u>

Eligible Customers are entitled to the following rates for all intraLATA and interLATA calling card calls:

| | Initial 60 Seconds | | Each Add'l 6 Seconds | |
|---------------------------|--------------------|----------|----------------------|----------|
| | <u>Peak</u> | Off-Peak | <u>Peak</u> | Off-Peak |
| - IntraLATA Long Distance | \$0.260 | \$0.220 | \$0.026 | \$0.022 |
| - InterLATA Long Distance | \$0.260 | \$0.220 | \$0.026 | \$0.022 |

Additionally, a \$0.25 per call surcharge shall apply to all calls and a \$1.25 per call surcharge shall additionally apply to all calling card calls that originate from a payphone.

For billing purposes, the minimum call duration for switched outbound intraLATA and interLATA long distance usage shall be sixty (60) seconds. Usage is measured thereafter in six (6) second increments and rounded to the next higher six (6) period.

Peak period is defined as the period from 8:00 am up to but not including 5:00 pm local time at the originating city Monday through Friday.

Off peak period is defined as the period from 5:00 pm up to but not including 8:00 am local time at the originating city Monday through Friday, and any time on Saturday, Sunday, and Companydesignated holidays.

Promotional Period

The promotional period will be from February 15, 2000 through May 15, 2000.

Conditions and Limitations

- A. Charges provided for in Section 4.3.4 of this Tariff apply to each access line in addition to the monthly rate provided above.
- B. Service Ordering Charges apply as provided in Section 4.1 of this Tariff.
- C. Customers may subscribe to additional optional exchange access service enhancement features at the rates provided in Section 4.5.3 of this Tariff. Rates for pay per use/per activation optional exchange access service enhancement features apply at all times as provided in Section 4.5.3 of this Tariff.
- D. Rates for Local Calling Plus apply as provided in Section 4.4.3.2 of this Tariff.
- E. Operator assistance surcharges apply as provided in Section 4.6.2 of this Tariff.
- F. Miscellaneous Service charges apply as provided in Section 14.2.4 of this Tariff.

PROMOTIONAL OFFERING

Residential customers who subscribe to the Company's local exchange services pursuant to Section 4.4.3.7 or 4.4.5.1 during the period of March 8, 2001 through May 31, 2001 ("Eligible Customers") will receive a credit of one (1) month's access line charges. The credit shall be applied during the Eligible Customer's second month's billing cycle.

PROMOTIONAL OFFERING

This promotional offer is for new Ohio residential local exchange service customers that subscribe to the Company's residential Unlimited plan located at Section 4.4.3.8.3 within this tariff pursuant to this promotional offering.

This offer is subject to service availability and available only for customers who are age 18 or over when ordering Unlimited between May 10, 2003 and June 15 and who maintain service for at least 90 days following such order. For customers who order the Unlimited service plan, a credit of \$60.00 will be applied to Customer's next bill after three months of service in good standing. Offer is not valid with any other promotional offers. This offer expires June 15, 2003.

PROMOTIONAL OFFERING

Refer a Friend Promotion

General

The Refer a Friend Promotion is available to Customers subscribing to the Company's services pursuant to Section 4.4.5.1 during the promotional period ("Eligible Customers"). The Company will issue a service credit to an Eligible Customer who refers a new Customer to subscribe to the Company's service pursuant to Section 4.4.5.1 during the promotional period (a "New Subscriber"). The service credit issued to the Eligible Customer will be equal in value to the Eligible Customer's monthly rate of service for the Eligible Customer's first line only. Subject to the limitations following, a credit will accrue to the Eligible Customer's account only after the completion of the New Subscriber's second billing cycle.

Promotional Period

The promotional period will be from July 17, 2001 to September 14, 2001.

Limitations

At the time of subscription, the New Subscriber must identify the Eligible Customer to the Company by name and telephone number. The Company will not permit retroactive identification of an Eligible Customer following such subscription.

The New Subscriber must maintain service with the Company for two consecutive billing cycles. If prior to the completion of the second such cycle, the New Subscriber cancels service or the Company discontinues the same for violation of this Tariff, the Eligible Customer will not be entitled to the accrual of any credit for that referral.

The following are not deemed to be New Subscribers: (a) existing Customers who add to or change services or (b) former Customers who cancel the Company's service during the promotional period and subsequently re-subscribe for services during the same period.

Where multiple credits accrue to an Eligible Customer's account during a single billing cycle, the Company will apply one such accrued credit to the account, and defer the remainder of the credits. Such deferred credits will be applied to the Eligible Customer's account at a rate of one credit per billing cycle in each successive month until such deferred credits are exhausted. Eligible Customers may not otherwise defer the application of credits to any subsequent month. Regardless of the foregoing, no Eligible Customer may receive a credit pursuant to this promotional offering for more than three consecutive billing cycles.

PROMOTIONAL OFFERING

This promotional offer is for residential local exchange service customers who have left CoreComm Newco, Inc., (the "Company") to take service with another local exchange carrier and upon returning to the Company subscribe to one of the Company's bundled offerings including; Local Plus 5, 2 Line premium Connect and Unlimited as described in Section 4.4 of this tariff.

This offer is subject to service availability and available only for customers who are age 18 or over when ordering one of the Local Plus 5, 2 Line Premium Connect or Unlimited plans pursuant to this promotional offering between May 13, 2003 and June 15, 2003. These customers will receive a monthly credit on their invoices from the Company of \$14.00 for the first (6) six months of service after returning to the Company. Offer is not valid with any other promotional offers and is contingent upon customers maintaining accounts that are in good standing. This offer will expire June 16, 2003.

PROMOTIONAL OFFERING

Employee Long Distance Promotion

General

The Employee Long Distance Promotion is available to all employees of the Company who subscribe to the Company's Residential intraLATA and/or interLATA long distance services during the promotional period ("Eligible Customers"). Subject to the limitations following, Eligible Customers shall receive the following flat per minute rate for all switched outbound calls in lieu of the intraLATA and interLATA long distance service rates otherwise applicable under Section 14.1 of this price list:

Employee Long Distance Promotion Rate: \$0.08 per minute

The minimum call duration for such calls shall be sixty (60) seconds, and usage shall be measured thereafter in sixty (60) second increments, rounding up to the next full sixty (60) second increment.

Promotional Period

The promotional period will be from December 2, 1999 through February 28, 2000.

Limitations

The Employee Long Distance Promotion Rate is not applicable to intraLATA and interLATA dedicated inbound or outbound service, intratLATA and interLATA switched inbound service, Calling Card Calls, Toll Free Calls, or any usage on lines pre-subscribed to the Company's business long distance services.

PROMOTIONAL OFFERING

Residential customers who subscribe to (a) the Company's local exchange services pursuant to Section 4.4.3.7 or 4.4.5.1, and (b) the Company's intraLATA and interLATA services during the period of March 8, 2001 through May 31, 2001 ("Eligible Customers") will receive 60 free minutes of switched outbound intraLATA and interLATA message toll service usage (the "Allotment") for each of their first three months of service. The Allotment will appear as a credit on each Eligible Customer's invoice issued in the month after each relevant month of service.

Customers subscribing to multiple lines from the Company will receive only one monthly Allotment applicable to aggregated usage on all pre-subscribed lines. Eligible Customers may not carry forward any unused portion of an Allotment in any month for application against a subsequent month's usage, nor are they entitled to a credit or refund for the same.

PROMOTIONAL OFFERING

Smart Pack Association Member Promotion

General

The Smart Pack Association Member Promotion (the "Association Promotion") is available to Eligible Business Customers subscribing to the Company's Smart Pack Local and Long Distance Services under Sections 15.3 and 15.4, respectively, of this tariff. For the purposes of this promotional offering, an Eligible Business Customer is a business that (a) subscribes to the Company's Smart Pack Local and Long Distance Services pursuant to this promotional offering; (b) is a member of an association or a chamber of commerce that is also a Customer of the Company's Smart Pack Local and Long Distance Services; and (c) submits proof of such membership in the form of a membership agreement or a copy of a cancelled check for membership dues. Subject to the limitations following, Eligible Business Customers shall receive the following discount on all Smart Pack access line, local usage, standard feature, optional feature, and long distance charges applicable under Sections 15.3 and 15.4 of this tariff:

Smart Pack Association Member Discount: 5%

Promotional Period

The promotional period for the CRP will be from December 2, 1999 through February 28, 2000.

Limitations

The Smart Pack Association Member Discount shall not apply to the following:

- 1. Smart Pack Miscellaneous Charges established in Section 15.2 of this tariff
- 2. Centrex Smart Pack Service Charges established in Section 4.14.2 of this tariff
- 3. Any charges identified in Section 4.3.4 of this tariff
- 4. Any charges for optional exchange access service enhancement features subscribed to by a Customer in addition to those features available as Smart Pack standard and optional features.

PROMOTIONAL OFFERING

This promotional offer is subject to service availability and will be made available only for residential customers who are age 18 or over and who mention promotional code XXXXXXXX.

For an additional \$27.95/month, residential customers who purchase CoreComm's Unlimited Plan as contained within Section 4.4.3.8.3 of this tariff will able to add unlimited calling during all time of day rate periods for direct dialed 1+ interLATA interstate, interLATA intrastate, intraLATA intrastate, and intraLATA intrastate domestic calls.

Minutes used for Operator Assisted Calling, Calling Card, Toll Free Service, Directory Assistance, and International calling are excluded. CoreComm reserves the right to assess an \$0.08/minute charge for non local minutes of use in excess of 2,000 minutes per month in cases where Customer's use of this service is for non-residential purposes including, but not limited to, any commercial activities, profit-making or non-profit, home office, business, sales, telecommuting, telemarketing, fax broadcast, or for any other activity which is inconsistent with normal residential usage patterns.

PROMOTIONAL OFFERING

This promotional offer is subject to service availability and will be made available only for residential customers who are age 18 or over.

During the period April 15, 2003 through June 15, 2003, residential customers who purchase CoreComm's Unlimited Plan as contained within Section 4.4.3.8.3 of this tariff AND the unlimited calling promotion described at page 73 of the Addendum to Price List of this tariff will receive a one time credit equal to \$34.00.

The Company reserves the right to reverse any such credit if either the Unlimited Plan or the unlimited calling promotion are cancelled within the first two months of service.

This promotional offer is subject to service availability and will be made available only for residential customers who are age 18 or over.

During the period December 15, 2003 through January 31, 2004, new customers who purchase CoreComm's Freedom Plan will receive a one time credit equal to \$25.00 which will be applied to the Customer's first bill. Offers are not valid with any other promotional offers. Pricing, terms and product subject to change without notice. This offer expires January 31, 2004.

<u>Go Paperless Promotion:</u> Beginning February 1, 2010, First Communications is running a 6 month promotion available to all customers who signup for an online billing account. By signing up for online billing, customers automatically receive email notification for new invoices. They can also view past invoices, make payments and review their call detail.

The "Go Paperless Promotion" provides a \$5.00 credit to residential accounts and a \$10.00 credit to commercial accounts to all eligible customers. The credit recurs monthly as long as the customer maintains their online account.